Wages and their impact on individuals, households and firms

Anna Thoresson



The Institute for Evaluation of Labour Market and Education Policy (IFAU) is a research institute under the Swedish Ministry of Employment, situated in Uppsala. IFAU's objective is to promote, support and carry out scientific evaluations. The assignment includes: the effects of labour market and educational policies, studies of the functioning of the labour market and the labour market effects of social insurance policies. IFAU shall also disseminate its results so that they become accessible to different interested parties in Sweden and abroad.

IFAU also provides funding for research projects within its areas of interest. The deadline for applications is October 1 each year. Since the researchers at IFAU are mainly economists, researchers from other disciplines are encouraged to apply for funding.

IFAU is run by a Director-General. The institute has a scientific council, consisting of a chairman, the Director-General and five other members. Among other things, the scientific council proposes a decision for the allocation of research grants. A reference group including representatives for employer organizations and trade unions, as well as the ministries and authorities concerned is also connected to the institute.

Postal address: P O Box 513, 751 20 Uppsala Visiting address: Kyrkogårdsgatan 6, Uppsala Phone: +46 18 471 70 70 Fax: +46 18 471 70 71 ifau@ifau.uu.se www.ifau.se

For dissertations, quality is ensured in the traditional way, through the academic review procedure at Uppsala /Stockholm University. This dissertation has not been reviewed by IFAU.

Dissertation presented at Uppsala University to be publicly examined in Hörsal 2, Ekonomikum, Kyrkogårdsgatan 10, Uppsala, Friday, 29 January 2021 at 09:15 for the degree of Doctor of Philosophy. The examination will be conducted in English. Faculty examiner: Associate Professor Ana Rute Cardoso (Institute for Economic Analysis (IAE-CSIC)).

ISSN 1651-4149

Economic Studies 191

Anna Thoresson Wages and Their Impact on Individuals, Households and Firms Department of Economics, Uppsala University

Visiting address:	Kyrkogårdsgatan 10, Uppsala, Sweden
Postal address:	Box 513, SE-751 20 Uppsala, Sweden
Telephone:	+46 18 471 00 00
Telefax:	+46 18 471 14 78
Internet:	http://www.nek.uu.se/

ECONOMICS AT UPPSALA UNIVERSITY

The Department of Economics at Uppsala University has a long history. The first chair in Economics in the Nordic countries was instituted at Uppsala University in 1741.

The main focus of research at the department has varied over the years but has typically been oriented towards policy-relevant applied economics, including both theoretical and empirical studies. The currently most active areas of research can be grouped into six categories:

- * Labour economics
- * Public economics
- * Macroeconomics
- * Microeconometrics
- * Environmental economics
- * Housing and urban economics

Additional information about research in progress and published reports is given in our project catalogue. The catalogue can be ordered directly from the Department of Economics.

Anna Thoresson

Wages and Their Impact on Individuals, Households and Firms



UPPSALA UNIVERSITET Dissertation presented at Uppsala University to be publicly examined in Hörsal 2, Ekonomikum, Kyrkogårdsgatan 10, Uppsala, Friday, 29 January 2021 at 09:15 for the degree of Doctor of Philosophy. The examination will be conducted in English. Faculty examiner: Associate Professor Ana Rute Cardoso (Institute for Economic Analysis (IAE-CSIC)).

Abstract

Thoresson, A. 2020. Wages and Their Impact on Individuals, Households and Firms. *Economic studies* 191. 220 pp. Uppsala: Department of Economics, Uppsala University. ISBN 978-91-506-2854-8.

Essay I: This paper studies how wages respond to employer concentration. It exploits a reform that deregulated the Swedish pharmacy market, which until 2009 was a monopoly. The reform involved a substantial increase in the number of employers on the pharmacy labor market. However, the change in employer concentration was not geographically uniform: certain areas experienced large changes while others were largely unaffected. Exploiting this geographical variation, elasticities of wages with respect to labor market concentration in concentration controlled by the policymaker to remedy the concern that actual labor market concentration are found to be most prevalent for stayers, rather than new hires, as well as those with more industry experience and longer tenure. Overall, the paper adds to a growing literature that finds that market concentration matters for workers' wages, in a context where labor is highly industry-specific.

Essay II (with Olof Åslund, Cristina Bratu and Stefano Lombardi): This paper studies the role of firm productivity in explaining earnings differences between immigrants and natives in Sweden. We first show that firms with higher value added per worker pay higher earnings and document that immigrant workers are under-represented in high-productive firms relative to natives. Next, we estimate substantial positive earnings returns to working in more productive firms, with significantly larger returns for immigrants from non-Western countries. We also find that immigrants are less likely to move up the firm productivity distribution. Sorting into less productive firms thus decreases earnings in poor-performing immigrant groups that would gain the most from firm productivity. The results are consistent with firms having differential wage-setting power over immigrants and natives.

Essay III (with Erik Grönqvist and Lena Hensvik): We study the effects of introducing a performance-based promotion program for teachers in Sweden. The program intended to make the teaching profession more attractive by raising wages for skilled teachers and taking advantage of teachers' professional competence. Our results show that: (i) high-wage teachers are more likely to be promoted; (ii) the stipulated wage increase has full pass-through onto wages for promoted teachers; (iii) schools with promotions have lower teacher separations and an improved pool of teachers; (iv) the promotion program improved student performance. These results suggest that performance-based promotions could be an important tool for raising school quality.

Essay IV (with Erik Grönqvist and Lena Hensvik): This paper studies the impact of a substantial change in household relative wage on the reallocation of childcare time across parents. Our empirical strategy takes advantage of a promotion program for teachers, which led to a sudden and persistent 20 percent wage increase for the promoted spouse and a 32 percent decrease in the couple's gender wage gap (reflecting the higher promotion rate of the female spouses). Our findings suggest that female promotions lead to a more even division of childcare and that the change in bargaining power of the promoted spouse is a contributing mechanism behind the effect. Overall, these findings suggest that improved career opportunities for women can improve gender equality in the household.

Keywords: Wages, wage inequality, firms, competition, immigrant-native earnings gaps, career opportunities, teacher labor markets, household behavior, gender gaps

Anna Thoresson, Department of Economics, Box 513, Uppsala University, SE-75120 Uppsala, Sweden.

© Anna Thoresson 2020

ISSN 0283-7668 ISBN 978-91-506-2854-8 urn:nbn:se:uu:diva-425941 (http://urn.kb.se/resolve?urn=urn:nbn:se:uu:diva-425941)

Till min mormor, Erna

Acknowledgments

As this roller-coaster comes to an end, there are many people to whom I owe thanks. First of all, I'd like to thank Lena, my main supervisor. Thank you for always believing that things are possible, for not letting me get stuck in details, and for the support and guidance, both professionally and personally. You have been a great mentor and I have learned a great deal from you. I also owe a lot of thanks to Oskar, my second supervisor. I am amazed at the speed at which you can deliver feedback that is right on point. Thank you for helping me focus my papers and improve my writing.

This thesis has benefited greatly from joint work with my co-authors. Erik, thank you for your dry humor and sometimes maddening curiosity. It's made the papers better and meetings more enjoyable. Olof, thank you for sharing your wisdom on the immigration literature and policy, and for your positivity toward our work. Cristina and Stefano, thank you for being both wonderful researchers and friends. Our dinners are now on Zoom since you moved to Helsinki, so I sadly don't get to taste your Italian/Romanian cooking any longer, but it's comforting to know you're only a short journey away. I look forward to working more with all of you in the future.

During my studies I've had the benefit of splitting my time between the economics department and IFAU. Both are nourishing environments, and I have learned a lot from friends and colleagues in both places. A particular thank you to Erik Ö, Georg, Nils and P-A for taking the time to listen to and comment on my work. Thanks also to the Uppsala Labor Group and Uppsala Center for Labor Studies for giving me the opportunity to present my work and listen to others' research. Alex and Luca, it was great to TA for you on the econometrics courses. Patrik, you have been a huge help with all my data questions. I am thankful to the Jan Wallander and Tom Hedelius Foundation, and Claudia Goldin, for getting the opportunity to spend part of my PhD at Harvard University. In addition, this thesis has improved greatly from helpful feedback received at my licentiate seminar and final seminar. A big thank you to Rune Vejlin for your thorough review of my work.

Pursuing a PhD is at times a lonely endeavor. Most days are spent by yourself, in front of your computer. With that in mind, I'd like to express my deepest thanks to my cohort: André, Daniel, Davide, Jonathan, Julian, Lillit, Sebastian and Sofia. You have made these years less lonely, and certainly a whole lot more enjoyable. Thank you for the support, whether it's comments on a paper, help with a problem set or encouraging words before a presentation. Thank you also for all the times we have not spent working. The unforgettable nights in Broddbo, Valborg and Christmas parties at Jonathan and Sofie's place, lobster rolls in Boston, and game nights at Daniel and Sofia's. The friendships we have made are indeed what I cherish the most. A special thank you to Lillit, Sofia and Sofie for the dinners and the chats. Lillit, it's been wonderful sharing an office with you. Whenever I've had doubts, you've always been there to give me an encouraging word.

In addition to my cohort, the PhD program is full of great PhD students. These last four years would have been much harder without the support from these fellow students. A particular thank you to Muskelgruppen for all the laughs, both inside and outside the gym, and Mohammad, for always smiling. The PhD Association deserves a special shout out for looking out for the PhD students, and Lucas for organizing the brown bag seminar series.

I am also deeply indebted to my family. Mamma och pappa, thank you for opening the door to so many opportunities. You have given me a great foundation to stand on, and have always encouraged me to do what I want to do. Helena, thank you for being exactly what a big sister should be. I know I can always turn to you to share the ups and downs. Thank you also for bringing Mattias, Rebecca, Alexander and Nora into our lives. Mormor, thank you for all the stories. I have dedicated this thesis to you. I know that when you were growing up, it was not as easy for women to educate themselves as much as they wanted to.

A big thank you also to my Icelandic family. Linda, Stefán and Davíð, you always make me feel so welcome. Hlynur and Ösp, it's been wonderful living near you, both in London and in Sweden. Eik and Móa, you are bundles of joy. I will miss you dearly when you move.

I also have the fortune of being surrounded by amazing friends. Anna, Baiqu and Laura, we've grown up and grown increasingly close since we lived together at Warwick. I admire you and am deeply grateful to have you in my life. Julia, we spent many years being inseparable. While we're currently separated in different cities, I am so glad to have you as a dear friend. Sarah, you are one of my oldest friends. Sometimes months or even years go by before we speak, but when we do we always pick up right where we left off. Elsa, Elsa and Sofie, without our brunches, dinners and walks I don't know how I would have managed these past years. Thank you for all the great talks, the support and the happy distraction from the PhD.

There is one person that is deserving of the biggest and final thank you, and that person, of course, is Arnaldur. It's hard to summarize in few sentences how much you mean to me (and to the production of this thesis!). These past eight years have flown by, and I can't wait to see what our lives will look like in another eight. Thank you for everything.

Uppsala, November 2020

Contents

In	troduc	tion	13
1	Emp	loyer Concentration and Wages for Specialized Workers	21
	1.1	Introduction	22
	1.2	Theoretical framework	26
		1.2.1 Sources of labor market power	26
		1.2.2 A Cournot oligopsony model	26
	1.3	Definitions, data and sample	29
		1.3.1 Definitions	29
		1.3.2 Data	30
		1.3.3 Main sample	32
	1.4	Institutional setting and descriptive patterns	32
		1.4.1 Introducing competition on the pharmacy market	32
		1.4.2 Employees	37
		1.4.3 Wages	38
	1.5	Empirical strategy	40
	1.6	Results	42
		1.6.1 Wage effects	42
		1.6.2 Composition effects	45
		1.6.3 Robustness checks	46
	1.7	Heterogeneity in estimated effects	48
		1.7.1 New hires and stayers	48
		1.7.2 Individual characteristics	52
	1.8	Conclusion	54
	Appe	endix A: Additional description and results	60
	Appe	endix B: Robustness checks	68
	rr ·		
2	Firm	s, Productivity, and the Immigrant-Native Earnings Gap	71
	2.1	Introduction	72
	2.2	Data and main sample	76
		2.2.1 Summary statistics	77
	2.3	Immigrant-native earnings ratio	77
	2.4	Firm productivity ranking	80
		2.4.1 Firm productivity and earnings	80
		2.4.2 Robustness of the firm ranking	82
	2.5	Sorting and returns to working in productive firms	85
		2.5.1 Distribution of workers in the firm productivity	
		distribution	

		2.5.2	Earnings returns to working in more productive firms	87
		2.5.3	Upward mobility in the productivity distribution	93
		2.5.4	Decomposition of the immigrant-native earnings gap	95
	2.6	Conclus	sion	99
	Appe	ndix A: A	Additional description and results	103
	Appe	ndix B: A	Additional robustness checks	108
3	Teach	er Caree	r Opportunities and School Quality	111
-	3.1	Introdu	ction	112
	3.2	Instituti	ional setting and data	116
	3.2	3 2 1	The career teacher reform	116
		322	Data	120
	3.3	Allocat	ion of career teacher positions across schools and	120
		teachers	s	123
		3.3.1	Allocation across schools	123
		3.3.2	Allocation across teachers	124
	3.4	Pass-th	rough of stipulated wage increase on promoted	
		teachers	s' wages	129
	3.5	Impact	of promotions on teacher separations, composition and	-
		student	performance	132
		3.5.1	Empirical strategy and identification	133
		3.5.2	Results on teacher turnover	134
		3.5.3	Heterogeneity analysis	137
		3.5.4	Teacher composition	142
		3.5.5	Student performance	143
	3.6	Conclus	sion	147
	Appe	ndix A: S	Supplementary description	153
	Anne	ndix B· A	Additional results	159
	Appe	ndix D: F	Robustness checks	176
4	Spous	sal Earnii	ngs and Household Dynamics: Evidence From a	
	Prom	otion Ref	form	181
	4.1	Introdu	ction	182
	4.2	Instituti	ional setting	186
		4.2.1	The promotion reform	186
		4.2.2	Temporary parental leave (TPL) system	187
	4.3	Empiric	cal strategy and data	188
		4.3.1	Data	189
	4.4	Descrip	otive statistics	192
		4.4.1	Summary statistics: teachers and couples before	
			promotion	192
		4.4.2	Raw within-couple differences by year	194
		4.4.3	Promotion impact on household relative wage	194
	4.5	Promot	ion impact on the division of childcare	198

	4.5.1	Heterogeneity analysis	
4.6	Conclu	ision	
Appe	endix A:	Additional tables and figures	

Introduction

Wages are an important determinant of individuals' well-being. They are an essential component of household income, and are a prerequisite for obtaining a decent standard of living. Still, wages differ considerably between groups: CEOs earn more than production workers, men earn more than women, and natives earn more than immigrants. Even workers who share the same characteristics earn substantially different wages. Moreover, in recent decades wages of low-income workers have been stagnating while wages of high-income workers have increased rapidly, widening inequality in many countries around the world.

To be able to explain these patterns requires an understanding of what determines wages and why wages differ across workers. Economists have a long history of trying to understand exactly this. The classic interpretation is that market forces determine wages. Wages are set such that labor supply equates labor demand for similar types of workers in similar types of jobs. In this interpretation, wages differ across workers for two main reasons. First, not all workers are equally productive. Workers that have higher so-called human capital are rewarded on the labor market for this in terms of higher wages. Second, jobs come with different amenities. Being a fishermen in winter is more dangerous than in summer, and wages for fishing move similarly. Already in the 18th century Adam Smith (1776) wrote about the idea that these unpleasant aspects of jobs can influence the wages that they pay.

While there is much merit to the classic interpretation – for example, more highly educated workers indeed tend to earn more – it does not capture all the complexities of reality. In the past decades there has been a shift among many economists to viewing labor markets as imperfectly competitive. They are types of markets that are riddled with frictions and information asymmetries. It is difficult to gather information about jobs, it takes time to search for a new job, and workers are normally restricted to working close to their home. This has repercussions for the wages that workers earn.¹

In this thesis I dig deeper into some of the key determinants and consequences of wages. I study different aspects of wage-setting such as the role of competition between firms in chapter 1, the role of firm productivity in chapter

¹There are many economic theories of wage determination that differ from the standard competitive model. These include efficiency wage theories (Shapiro and Stiglitz 1984), insider-outsider models (Lindbeck and Snower 2001), matching models (Mortensen and Pissarides 1999), and monopsony models (Manning 2003). Workers may also earn different wages due to discrimination, though this is normally modelled in a competitive model (see Becker 1957).

2, the role of institutions in chapter 3, and the impact that wages can have on household behavior in chapter 4. Workers can earn different wages depending on how many firms compete for their services, which I explore in chapter 1. The types of firms that individuals have access to can also drive differences in wages across groups, which I address in chapter 2. Moreover, wages matter for more than the consumption possibilities of workers. They can affect where people choose to work, something which I analyze particularly in chapter 3. They can also matter for how we choose to live our lives outside the labor market. As an illustration of this, chapter 4 studies how parents choose to allocate household chores – particularly, taking care of an ill child – when one person in the couple receives a wage increase. I proceed by outlining each chapter in more detail.

In the thesis' first chapter, titled **Employer concentration and wages for specialized workers**, I study a major reform to provide evidence on how increasing the number of employers affects workers' wages. The reform deregulated the Swedish pharmacy market. Prior to 2009, pharmacies were operated by a state-run monopoly. This monopoly was dismantled in 2009, after which entry was allowed into the market. At first sight, this involves a change on the product market: following the deregulation, both private firms and the former monopoly operate pharmacies. If, for example, profits fall as more firms enter and firms tend to share these rents with their workers, then we expect wages to fall following the deregulation. However, this type of market – and many others, like teaching or healthcare – is interesting because there is a tight link between the product market and the labor market. Put simply, the majority of pharmacists and pharmacy technicians work in pharmacies. This means that competition between firms for labor increases at the same time, which can have a counteracting effect on wages and push them up.

When firms have the power to set wages, we deviate from a setting where competition disciplines wage-setting. A topic that takes a central role in the first chapter is monopsony. In the strictest and classic sense, monopsony can be contrasted to monopoly – instead of one seller, there is one buyer. Pioneered by Robinson (1933), classic monopsony power is therefore closely linked to the number of employers on a market.² A lack of competition between firms means that workers are not as sensitive to wage changes as competitive models would suggest. This allows employers to set lower wages than they otherwise would be able to.

In the chapter, I isolate the effect of increasing labor market competition and indeed find that wages increase in response to the reduction in employer concentration. Empirical economists spend a lot of time and energy trying to establish causality: how can we know that the increase in the number of

²In more modern versions of the theory, that it takes time to search for new jobs (Burdett and Mortensen 1998; Manning 2003) or that workers' have heterogeneous tastes for jobs in ways which are not perfectly observed by employers (Card et al. 2018) are the key mechanisms explaining why firms have market power.

employers *caused* wages to increase, rather than something else, which may affect both the number of employers and wages? The "gold standard" in economics to obtain causal estimates, similar to medicine, are randomized experiments.³ In absence of these, economists often look for natural experiments to estimate causal effects. In natural experiments we can compare one group who experienced a change to another group that did not. In the first chapter, the deregulation serves as such an experiment. I study pharmacy workers who all experienced the abolition of the national monopoly, but whose experiences of changes in their local labor market conditions differ. The change in the number of employers that the workers face will depend on where in Sweden they work: in some parts of the country, the increase in the number of employers was large, while in others, the increase was modest to non-existent. I am therefore able to compare workers that experienced different changes in the number of employers over time to obtain causal estimates of the effect of competition between employers on wages.

In a world where firms have wage-setting power, not only worker characteristics but also the place of work can matter for workers' wages. Previous research points to two empirical facts. First, wage differences between firms have increased over time (Card et al. 2013; Barth et al. 2016). Put differently, it has become increasingly important where you work for your take-home pay. Second, there is substantial ethnic workplace segregation (Hellerstein and Neumark 2008). However, we know relatively little about the repercussions of this for immigrant workers, and particularly for immigrant-native differences in earnings.

In the second chapter, titled **Firms, productivity, and the immigrantnative earnings gap**, co-authored with Olof Åslund, Cristina Bratu and Stefano Lombardi, we bridge this gap in the literature. Like in the other chapters in this thesis, we take Sweden as a case study, where we have access to rich matched employee-employer data. It is interesting to study for a number of reasons. First, Sweden is a heterogeneous country, thanks largely to the considerable and diverse immigration it has experienced in recent history. In 2020, the share of foreign born was 20%, nearly double what is was in 2000 (SCB 2020). Second, the data we use show that the immigrant-native earnings gap has increased by six percentage points between 1998 and 2017, the period we study. This change is driven by changes between rather than within firms. Understanding in which types of firms immigrants and natives work, and how earnings differ for immigrants and natives in these firms, are an important piece of the puzzle to understanding immigrant–native earnings differences.

In the chapter, we focus on a particular aspect of firms – namely firm productivity – and study how this translates into differences in earnings between immigrants and natives. In line with authors before us (see Card et al. 2018), we find that more productive firms pay more. We add to the literature by fo-

³For a historical review of empirical economics, see Angrist and Pischke (2010).

cusing on the relation between immigrant-native earnings differences and firm productivity. We find that immigrants are over-represented in low-productivity firms and are less likely to move to higher productivity firms than natives are. However, immigrants that manage to climb the productivity ladder experience larger gains in earnings than natives do. This is particularly true for groups of immigrants that tend to do relatively worse on the labor market: immigrants from non-Western countries as well as those that have arrived to Sweden more recently. Overall, our work shows that firm policies are a key contributor toward understanding why earnings differ between immigrants and natives.

The third chapter in the thesis shifts its focus to teacher labor markets. Teaching has historically been a female-dominated profession, especially at lower levels of education, characterized by low wages compared to other workers with comparable skills. While good teachers are key to students doing well in school and beyond (see Chetty et al. 2014), the relatively low and compressed wages can hamper the ability to attract and retain qualified teachers. Making sure that there are sufficient numbers of good teachers is also of high importance at a time when many developed countries are facing large retirement waves among teachers.

One way to make wages more aligned to the skills that teachers have are through promotions. In the third chapter, titled **Teacher career opportunities and school quality**, co-authored with Erik Grönqvist and Lena Hensvik, we study the effects of a performance-based promotion program. In 2013, the Swedish government introduced a new career step: the career teacher. If promoted, the teacher receives a substantial wage increase of SEK 5,000 (USD 520), representing a 15% to 20% increase compared to what the teacher earned prior to being promoted. Career teachers primarily continue to teach but also undertake additional tasks, such as mentoring their teacher peers and working with the school's pedagogical development.

By making use of an institutional feature – that the reform was rolled out across schools over time – we are able to study the effects of career teachers on teachers and students. Overall, introducing promotion opportunities has a positive impact on several important school outcomes. We find that providing additional career opportunities reduces the likelihood that teachers quit, both to other schools and out of the profession, and lead schools to retain a larger share of experienced and certified teachers. We also find that student performance improves in schools that introduced the career opportunity, though the changes are modest in size. More generally, the chapter shows how wages can influence where individuals choose to work. In the case of teachers, this can also have long-term repercussions for the students that they teach.

The final chapter of this thesis builds on the same reform studied in the third chapter but shifts its focus to the home. One of the most significant labor market changes in the past century is the rapid increase in female labor force participation. As Goldin (2006) puts forward, there is much more to this than an increase in hours worked. Particularly from the late 1970s onward,

there has been a shift from women holding intermittent jobs to having careers, and from women being secondary earners in the household to actively making career decisions. Despite this significant shift, there is a persistent gender gap in earnings and wages that remains to this day. The gap is intertwined with the fact that women spend considerably more time doing household chores and caring for children than men do. This is true in many countries around the world, including in Sweden, which we study in this chapter.

In the fourth chapter, titled Spousal earnings and household dynamics: Evidence from a promotion reform, also co-authored with Erik Grönqvist and Lena Hensvik, we study how household decision-making with regards to taking care of children is affected when one parent - primarily the mother - receives a promotion. The wage increase that the promotion entails has the ability to influence time spent at home because the cost of not working is now higher. Moreover, as mothers are promoted, it may influence their bargaining position in the household. At the same time, gender norms may hinder a more even division of household chores. In the chapter we analyze the take-up of temporary parental leave (TPL, commonly referred to as VAB in Swedish) around the time of promotion to a career teacher. TPL is a benefit available in Sweden for working parents to temporarily care for ill children during work hours. While interesting to study in its own right, it also proxies well for how parents divide household chores more generally (Eriksson and Nermo 2010). To obtain causal estimates, we use the fact that promotions happened at different points in time for different parents, and compare how couples divide their time spent taking care of their ill children before and after promotion.

In line with our results in chapter 3, we find that promoted persons, relative to their partners, earn SEK 5,000 more following promotion. Because around 80% of the promoted teachers are women, this reduces the within-couple gender wage gap by a third among all promoted persons. Regarding take-up of temporary parental leave, we find that the couples respond to the new economic incentives. The promoted person reduces their take-up of leave relative to their partner once promoted. For women, promotions reduce the gender gap in temporary parental leave days taken by over half. Overall, the findings in the chapter suggest that, by improving career opportunities for women, gender inequality in the household can be reduced.

As a whole, the chapters in this thesis support the conclusion that wages are influenced by many factors beyond only human capital differences across workers. The place of work matters for wages, including labor market competition between firms. At the same time, access to good firms is unevenly distributed across workers. Moreover, institutional interventions on the labor market have the ability to impact the wages that individuals earn and the workplaces they choose to work in. They may also have repercussions beyond what was initially intended, such as affecting equality in the home.

References

- Angrist, J.D. and J.-S. Pischke (2010). "The Credibility Revolution in Empirical Economics: How Better Research Design is Taking the Con out of Econometrics". *Journal of Economic Perspectives* 24.2, pp. 3–30.
- Barth, E., A. Bryson, J.C. Davis, and R. Freeman (2016). "It's Where You Work: Increases in the Dispersion of Earnings across Establishments and Individuals in the United States". *Journal of Labor Economics* 34.S2, pp. 67– 97.
- Becker, G.S. (1957). *The Economics of Discrimination*. Chicago, Illinois: The University of Chicago Press.
- Burdett, K. and D.T. Mortensen (1998). "Wage Differentials, Employer Size, and Unemployment". *International Economic Review* 39.2, pp. 257–273.
- Card, D., A. R. Cardoso, J. Heining, and P. Kline (2018). "Firms and Labor Market Inequality: Evidence and Some Theory". *Journal of Labor Economics* 36.S1, pp. 13–70.
- Card, D., J. Heining, and P. Kline (2013). "Workplace Heterogeneity and the Rise of West German Wage Inequality". *The Quarterly Journal of Economics* 128.3, pp. 967–1015.
- Chetty, R., J.N. Friedman, and J.E. Rockoff (2014). "Measuring the Impacts of Teachers II: Teacher Value-Added and Student Outcomes in Adulthood". *American Economic Review* 104.9, pp. 2633–2679.
- Eriksson, R. and M. Nermo (2010). "Care for Sick Children as a Proxy for Gender Equality in the Family". *Social Indicators Research* 97.3, pp. 341–356.
- Hellerstein, J.K. and D. Neumark (2008). "Workplace Segregation in the United States: Race, Ethnicity, and Skill". *The Review of Economics and Statistics* 90.3, pp. 459–477.
- Lindbeck, A. and D.J. Snower (2001). "Insiders versus Outsiders". Journal of Economic Perspectives 15.1, pp. 165–188.
- Manning, A. (2003). *Monopsony in Motion: Imperfect Competition in Labor Markets*. Princeton, New Jersey: Princeton University Press.
- Mortensen, D.T and C.A. Pissarides (1999). "Job Reallocation, Employment Fluctuations and Unemployment". *Handbook of Macroeconomics*. Vol. 1, pp. 1171–1228.
- Robinson, J. (1933). *The Economics of Imperfect Competition*. London: Macmillan.
- SCB (2020). Utrikes födda i Sverige (Foreign-born in Sweden). URL: https: //www.scb.se/hitta-statistik/sverige-i-siffror/manniskornai-sverige/utrikes-fodda/ (visited on 10/13/2020).

Shapiro, C. and J.E. Stiglitz (1984). "Equilibrium Unemployment as a Worker Discipline Device". *The American Economic Review* 74.3, pp. 433–444.

Smith, A. (1776). An Inquiry into the Nature and Causes of the Wealth of Nations. London: W. Strahan and T. Cadell.

1. Employer Concentration and Wages for Specialized Workers

Acknowledgments: I am grateful for comments from my supervisors Lena Hensvik and Oskar Nordström Skans as well as from Peter Arcidiacono, Erling Barth, Davide Cipullo, Niklas Engbom, Mette Foged, Peter Fredriksson, Nils Gottfries, Georg Graetz, Sebastian Jävervall, Lawrence Katz, Ilse Lindenlaub, Pedro Martins, André Reslow, Arnaldur Stefánsson, Claudia Steinwender, Christopher Taber, Rune Vejlin, Johan Vikström, Erik Öberg and seminar participants at the EEA Congress 2020, EALE/SOLE/AASLE World Conference 2020, 76th Annual Congress of the IIPF, the Institute for Evaluation of Labor Market and Education Policy (IFAU), Research Institute of Industrial Economics (IFN), University of Copenhagen, University of Gothenburg and Uppsala University. I thank Patrik Lind for helpful data advice. Financial support from the Jan Wallander and Tom Hedelius foundation is gratefully acknowledged.

1.1 Introduction

A recent and growing literature has revived interest in the idea that employers have wage-setting power. This research has sought to answer whether market power matters for workers' wages;¹ whether antitrust policy should focus more on judging labor market effects when reviewing mergers;² and to explain important labor market trends such as the falling labor share, stagnant wage growth and rising wage inequality by changes in product market power³ or labor market power.⁴ While there can be many sources of market power, a canonical source is market concentration. Market concentration relates to the existence of only a small number of sellers or buyers on a particular market. In the case of labor markets, having access to a limited number of employers can give employers the ability to depress wages below competitive levels.

In this paper, I focus on a particular market – pharmacies – and use a major policy reform to deduce quasi-experimental evidence on how increasing the number of employers affects workers' wages. Prior to 2009, only the state-run monopoly *Apoteket* could retail pharmaceuticals in Sweden. In 2009, entry barriers were removed and private firms could enter. *Apoteket* had to privatize two thirds of its pharmacies as part of the deregulation. Since pharmacists have highly industry-specific skills, the deregulation causes changes to the number of employers on the labor market. I exploit this setting to study how wages respond when labor market concentration falls. I rely only on the variation in concentration induced by the privatization of pre-existing pharmacies in 2009 to remedy the concern that actual labor market concentration is endogenous. This variation in concentration is controlled by the policymaker. I find that wages respond negatively to local labor market concentration, with estimated elasticities ranging from -0.02 to -0.05.

This paper studies a classic question in novel way. It uses a rare natural experiment that provides geographic variation in employer concentration coupled with rich matched employee-employer data. The setting is unique and interesting for a number of reasons. First, the deregulation resulted in substantial variation in employer concentration caused by a policy-decision. Second, product prices are mainly regulated and set by the state both before and after the reform, which make them independent of product and labor market concentration. The setting allows me to isolate the effects of changes to labor market concentration. Third, wage-setting in the industry is decentralized and wages are set in individual negotiations both before and after the deregulation.

¹See Berger et al. (2019); Jarosch et al. (2019); Lamadon et al. (2019); Card et al. (2018); Benmelech et al. (2018); and Azar et al. (2020a).

²See Naidu et al. (2018) and Marinescu and Hovenkamp (2018). Naidu and Posner (2019) argue that, due to the frictional nature of labor markets, there are limitations to what antitrust policy can achieve.

³See Autor et al. (2020); De Loecker et al. (2020); and Barkai (2020).

⁴See Berger et al. (2019); Rinz (2020); and Lipsius (2018).

Fourth, the study uses high quality matched employee-employer data. I observe all workers inside and outside the pharmacy industry, including detailed information about the workers such as their educational level and educational specialization, as well as transitions they make on the labor market.

The paper consists of four main parts. In the first part of the paper, I present a simple model that relates wages to labor market concentration. The overall effect on wages of the deregulation is ambiguous *a priori*. When isolating the effects related to reduced labor market concentration, however, that model predicts that wages increase. The framework highlights that the overall effect depends on two channels: changes to labor market power and changes to product market power. A decrease in product market power is likely to decrease wages. This can, for example, arise if profits fall when there is rent-sharing. However, the increase in the number of employers can have a counteracting, and positive, effect on wages as monopsony power is reduced. Originally coined by Robinson (1933), monopsony in the strictest sense is a situation with only one buyer. In labor markets, it has more generally come to refer to a situation where individual firms face upward-sloping labor supply (Boal and Ransom 1997).⁵ In contrast to competitive labor markets, firms are able to extract rents by setting wages below the marginal revenue product of labor.

The second part of the paper uses rich employee-employer data between 2004 and 2016 to characterize aggregate changes in the pharmacy industry, which is female dominated and highly skilled. The reform led to a large aggregate reduction in labor market concentration, measured by the Herfindahl-Hirschman Index (*HHI*), from 1 to 0.25. Wages for pharmacy employees increased on aggregate by 2 to 4 percentage points upon deregulation, relative to comparable workers in other industries. The number of pharmacies increased by around 50% and employment by 10%. Overall the market became more fragmented as employees split into more pharmacies operated by more firms.

The third and fourth parts of the paper include the main results. The third part estimates how labor market concentration affects wages. The change in labor market concentration induced by the deregulation differs across local markets. Making use of this geographical variation, elasticities of wages with respect to labor market concentration (or precisely, *HHI*) are estimated to be between -0.02 and -0.05. These results are consistent with previous studies (see for example Rinz 2020 or Hershbein et al. 2019). The effects materialize within two years of the deregulation and are relatively stable until the last period of observation in 2016.

⁵For a recent review of monopsony, see Manning (2020). Given the theoretical link between labor supply elasticities to the firm and labor market power, several papers have estimated the elasticity of labor supply to the firm in particular markets and found these to be far from perfectly elastic (see Sullivan (1989), Staiger et al. (2010) and Matsudaira (2014) for nurses; Falch (2010) and Ransom and Sims (2010) for teachers; and Dube et al. (2018) for the gig economy). This has been taken as support of that at least specific labor markets are imperfectly competitive, while being agnostic as to the actual source of labor market power.

The causal interpretation is supported by a battery of checks. Log wages evolve in parallel in local markets prior to deregulation and effects are not found in a related but unaffected industry. Only variation in labor market concentration from the sale of pre-existing pharmacies in 2009, the year the market was deregulated, is used to remedy the concern that actual labor market concentration is endogenous. This depends on the privatization of pre-existing pharmacies, a process controlled by the policymaker, and is neither affected by firms' decisions to open up new pharmacies nor to mobility decisions that workers make post-deregulation.

The fourth part of the paper studies which employees in the pharmacy industry benefit most from reduced labor market concentration. Stayers rather than new hires benefit with higher wages. That is, conditional on joining a new employer, the results suggest that there is no additional return to making the transition in a labor market with relatively low labor market concentration. This result should be interpreted with caution, however, as the likelihood of moving is itself affected by the deregulation. The positive wage effects are also estimated to be prevalent primarily for those with more industry experience and longer tenure. Individual characteristics do not appear to matter for the wage returns. Instead, similar positive wage returns from reduced labor market concentration are estimated for employees of different age, country of birth, gender, educational level and educational specialization.

The paper is related to three main strands of literature. First, it contributes to the literature that studies the effect of labor market concentration on wages. Whether there exists such a relationship is an old question, reviewed in Boal and Ransom (1997). More generally, this concerns imperfect competition in labor markets under the assumption that labor market concentration captures labor market power (see Manning (2011) for an overview). The question of whether labor market concentration affects wages, which according to Manning (2020) was originally studied by Bunting (1962), was revived recently by Azar et al. (2020a) and Benmelech et al. (2018), who estimate a negative relationship between these variables in a U.S. context.⁶ Empirically, these papers tend to exploit broad changes in concentration at the region by industry or occupation level. A key concern is that many factors affect both market concentration and wages (Berry et al. 2019), and it is hard in data to identify exogenous shifts in concentration. This paper contributes to the literature by focusing on a particular market where there is a policy change that has market structure implications.

⁶This negative relationship has been confirmed in many other studies including Hershbein et al. (2019), Qiu and Sojourner (2019), Lipsius (2018), Rinz (2020) and Schubert et al. (2020) using U.S. data; in Martins (2018) using data from Portugal; and in Marinescu et al. (2020) using data for France.

A second yet scarce strand of literature uses variation from mergers to study how market concentration affects workers.⁷ Arnold (2020) exploits mergerinduced changes in concentration in the U.S. to find negative effects of increased labor market concentration on earnings when the change in concentration is large. A related approach is taken by Prager and Schmitt (2019) who focus on hospital consolidation only. They find a negative effect on wages when the change in concentration is large and worker skills are industry-specific.⁸ While compelling, these papers suffer from the drawback that firms choose whether or not to merge. If firms merge for labor cost-saving reasons, then that could result in a spurious correlation between labor market concentration and wages. I take an alternative approach by focusing on changes in concentration induced by a policy change rather than a firm decision.

Third, the paper contributes to the literature on the wage effects of privatization and deregulation. In contrast to these papers, I isolate the wage effect of the regulatory change to that associated with changes in labor market concentration. The literature on how workers are affected by privatization is scarce. In two recent papers, Olsson and Tåg (2018) find increased unemployment incidence and duration while annual labor income and labor force participation remain unchanged for privatization events in Sweden, while Arnold (2019) finds that privatization lowered incumbents' wages substantially in Brazil. Regarding deregulation, this is often analyzed as a shock to profitability in a specific industry, which under rent-sharing would put downward pressure on wages. Peoples (1998) provides on overview of the wage effects of reducing entry barriers to specific industries. The literature predominantly finds negative industry wage effects of deregulation, including in the U.S. airline (Card 1998, Hirsch and Macpherson 2000), trucking (Rose 1987, Hirsch 1988) and banking industries (Black and Strahan 2001).⁹ This literature is often based on cross-sectional data and is, unlike this paper, unable to control for unobserved worker characteristics. An exception is Lergetporer et al. (2018), who find negative wage effects for incumbent workers after lifting entry barriers in the German crafts sector.

Overall, the results in this paper underpin that an increase in the number of employers in an industry with specialized labor increases wages. The paper lends support to the literature that finds that labor market concentration can matter for workers' wages. The rest of the paper proceeds as follows. A theoretical framework is included in Section 1.2. Key definitions and data

⁷In a related paper Hensvik (2012) studies the relation between school competition and teacher wages in Sweden. Hensvik (ibid.) focuses on how public-school hiring and wages are affected by private entry, finding that wages respond positively to the increased competition.

⁸Currie et al. (2005) consider labor market effects of mergers for hospitals without linking this explicitly to changes in market concentration. They find increases in nurse effort but no wage effects when studying hospital consolidation.

⁹The deregulation literature also finds that reducing entry barriers leads to employment growth. Bertrand and Kramarz (2002) study entry deterrence in the French retail industry.

are described in Section 1.3. Institutional details and descriptive patterns are provided in Section 1.4. Section 1.5 outlines the empirical strategy, Section 1.6 provides estimates of the wage effects of reduced labor market concentration and Section 1.7 considers which employees benefit most from this reduction. Finally, Section 1.8 concludes.

1.2 Theoretical framework

1.2.1 Sources of labor market power

In imperfectly competitive labor markets, monopsony powers can stem from many sources. The quasi-experiment studied in this paper naturally lends itself to studying the effect of labor market concentration on wages. Market concentration relates to the existence of only a small number of potential employers; that is, labor markets are thin. Due to regulatory barriers, only one firm was allowed to operate in the pharmacy industry prior to the reform. Building on a tradition in industrial organization and in antitrust policy, the intuition is that firms are interdependent and take actions of other firms operating in the same labor market into account when making employment decisions. In such a world, it may be profitable for firms to hire fewer workers and thereby set lower wages than in a perfectly competitive world. This type of argument is emphasized by classic models such as Cournot oligopsony and in empirical work by Azar et al. (2020a) and Benmelech et al. (2018), among others. Recent theoretical work by Berger et al. (2019) provides a micro-foundation to the relationship between market power and market structure. Their model allows for a large but finite number of employers, and market power arises from the ability of firms to exploit the market-share dependent upward-sloping labor supply curves to the firm.¹⁰

While I focus on a particular source of labor market power – labor market concentration – there are sources of monopsony power that arise even with a continuum of firms. These sources include differentiation and preference heterogeneity (see Bhaskar et al. 2002 and Card et al. 2018), as well as search frictions (Burdett and Mortensen 1998). Search frictions are key to the seminal dynamic monopsony models by Manning (2003).

1.2.2 A Cournot oligopsony model

In this section, I outline a simple static Cournot oligopsony model that relates wages and labor market concentration. The framework follows Arnold (2020),

¹⁰Jarosch et al. (2019) develop a model where a different mechanism gives rise to a relation between market concentration and wages. Market power stems from employer size where each employer recognizes it can exploit its power by eliminating its own vacancies from the worker's outside option and thereby not compete with itself. In the model, employment is not directly affected by market structure. It therefore departs from the Robinson-style models.

who exploits mergers and acquisitions to study how labor market concentration affects workers.¹¹ There are *F* firms in a market *m*, indexed f = 1, ..., Fand total employment on the market is $L_m = \sum_{f=1}^{F} l_f$, where l_f is firm *f*'s level of employment. For simplicity, labor is assumed to be the only input into production. Each firm maximizes its objective function by choosing its employment, l_f , taking the labor demand of other firms on the same market as given. The market wage $w_m(L_m)$ depends on total employment in the market. This is a posted wage; there is no wage bargaining in this model.

$$\max_{l_f} R_f(l_f) - w_m(L_m)l_f \tag{1.1}$$

 $R_f(l_f)$ is the firm's revenue function. This will depend on product market factors, like price and quantity of goods sold, as well as productivity parameters. More generally, $R_f(l_f)$ can be interpreted as any concave function that is increasing in l_f . This covers the case of the public monopsonist that may have additional objectives beyond maximizing revenue minus cost. This leads to the following first-order condition:

$$\underbrace{\frac{\partial R_f(l_f)}{\partial l_f}}_{\Omega_f \equiv MRPL_f} - \underbrace{\left[w_m(L_m) + \frac{\partial w_m(L_m)}{\partial l_f}l_f\right]}_{MLC_f} = 0$$
(1.2)

Notice that, in the absence of labor market power, the firm's labor decision would not affect wages and wages would be set to equal the marginal revenue product of labor (*MRPL*). The first order conditions can be re-written as follows:

$$\Omega_f - w_m(L_m) \left[1 + \frac{s_f}{\varepsilon_m} \right] = 0 \tag{1.3}$$

where $s_f = \frac{l_f}{L_m}$ is firm *f*'s employment share in market *m* and ε_m is the marketlevel labor supply elasticity, $\varepsilon_m = \frac{\partial L_m}{\partial w_m(L_m)} \frac{w_m(L_m)}{L_m}$. Multiplying each side by s_f and then summing the first order conditions across all firms, we can rearrange to find an expression for market wages:

$$w_m = \underbrace{\left[\frac{\varepsilon_m}{HHI_m + \varepsilon_m}\right]}_{\sigma_m} \Omega_m \tag{1.4}$$

where $HHI_m = \sum_f (s_f)^2$ is the Herfindahl-Hirschman Index and $\Omega_m = \sum_f s_f \Omega_f$ is the employment-weighted average of the marginal revenue product of labor. HHI_m can take values in the interval (0, 1] where values approaching 0 represent perfect competition and 1 represents only one employer in the market.

¹¹Arnold (2020) decomposes the effects of mergers into three components: monopsony effects, market power effects and productivity effects. It draws on classic Cournot oligopsony results, among others outlined in Boal and Ransom (1997) and Naidu and Posner (2019).

Unlike in the competitive model, workers will only get a fraction (denoted σ_m) of the average marginal revenue product. The model implies that higher concentration is negatively associated with wages, holding all else constant. Moreover, if market power only stems from market concentration, then the wage approaches the competitive wage as the number of firms increases.

Channels through which deregulation affects wages

In the simple model outlined above, the deregulation may impact wages through changes in labor market power (operating through σ_m) and through changes in productivity, product market power or objectives as ownership shifts from public to private (operating through Ω_m). Taking logs, the average treatment effect of the deregulation on log wages \tilde{w}_m can be written as follows. *post* refers to post-deregulation and *pre* to pre-deregulation:

$$E[\widetilde{w}_{m,post} - \widetilde{w}_{m,pre}] = E[\widetilde{\sigma}_{m,post} - \widetilde{\sigma}_{m,pre}] + E[\widetilde{\Omega}_{m,post} - \widetilde{\Omega}_{m,pre}]$$
(1.5)

While decreases to labor market concentration will increase wages, decreases to Ω_m are, on the other hand, likely to put downward pressure on wages. To the extent that these two channels are correlated, I am likely to estimate lower bound effects of reduced labor market concentration on wages.

Focusing on the Ω_m component, increased product market competition will in general put downward pressure on prices, and under rent-sharing, also on wages.¹² While the price of prescription drugs, the dominant product category, is regulated, this is not the case for non-prescription drugs and retail items which represents around 25% of revenue. Indeed, earlier deregulation studies (Black and Strahan 2001) have used deregulation as a shock to profitability and find support for the rent-sharing channel as wages fall post-deregulation. Peoples (1998) highlights how labor earnings may fall after deregulation as the bargaining power of workers falls. Recalling that the reform considered in this paper also involves privatization and that the revenue function can be interpreted as any function that is increasing in l_f , Arnold (2019) finds that state-owned enterprises pay significant wage premiums over private firms, also suggesting that changes to Ω_m could put downward pressure on wages. This is consistent with state-owned enterprises having wider objectives than only maximizing profits (Haskel and Szymanski 1993). Indeed, trade unions often fear privatization will lower wages and much academic literature has assumed privatization has negative effects on wages.¹³

¹²The model in Section 1.2.2 implies that, if decreases to product market power increase employment, then wages also increase. This is not the case with wage bargaining, where lower revenue per worker results in lower wages.

¹³Even so, there is no general theory of how deregulation and privatization will affect wages, and certain product market channels could push wages up. Many papers find that efficiency and profits increase once state owned enterprises are privatized (see Megginson and Netter (2001) for a review), which could put upward pressure on wages if private firms share rents at least as

To isolate labor market power effects (captured by σ_m) from productivity or product market aspects (captured by Ω_m), a sufficient condition is that changes to Ω_m are independent of changes to σ_m , conditional on included controls. For example, if changes to labor market power are correlated with changes to product market power, and if higher product market power has a positive effect on wages, then I am likely to underestimate the effect of labor market concentration on wages. Similarly, if public monopsonists overpay compared to private monopsonists, then the state-to-market quasi-experiment that this paper relies on will also underestimate the effects of reduced labor market concentration. In support of this assumption, firstly note that decisions about product pricing and campaigns are normally taken nationally and the product ranges at pharmacies are relatively homogeneous (KKV 2013). The empirical strategy only exploits within-industry changes. Moreover, while pharmacies do sell products with unregulated prices, around three quarters of pharmacies' revenues are from products with nationally regulated prices. That product prices are regulated both before and after the reform make them independent of product and labor market concentration.

1.3 Definitions, data and sample

1.3.1 Definitions

In order to calculate concentration measures, it is necessary to define what a market is. The definition of the labor market should capture the set of potential employers for a worker. Because workers are tied to their workplace, labor markets tend to be local. In this paper, a local labor market (LLM) m is defined by the interaction of the industry for dispensing chemists and commuting zones (CZ). The industry for dispensing chemists is identified by workplace industry codes included in the data.¹⁴ Commuting zones encompass all industries in a geographic area and are taken from Statistics Sweden who define CZs using commuting patterns.

In line with the theoretical framework, labor market concentration is measured using the Herfindahl–Hirschman Index (*HHI*) in employment shares. This measure captures concentration among pharmacies. For all workplaces that are operating as dispensing chemists, a unique workplace identifier is assumed to be a pharmacy. A firm is defined using firm identifiers provided in the data as a collection of pharmacies. *HHI* is calculated separately by year *t* and LLM *m* as the sum of squared employment shares s_f across firms in each

much as public firms do. Earle and Shpak (2019) summarize why wages may rise or fall as a result of privatization.

¹⁴Industry (SNI) code 52.310 is used until 2007 and 47.730 from 2008 onward. There is a one-toone mapping between these codes. To remedy potential miss-classifications of workplaces that may occur especially around the time of the deregulation, I (iteratively) assume that a workplace is a dispensing chemist if it was classified as a dispensing chemist in the previous year.

local pharmacy market:

$$HHI_{mt} = \sum_{f=1}^{F} s_{fmt}^2 \tag{1.6}$$

An *HHI* approaching 0 corresponds to perfect competition while an *HHI* equal to 1 corresponds to a single employer.¹⁵ A higher value means higher concentration and thus lower competition. *HHI* is a canonical measure of labor market concentration, used among others in Benmelech et al. (2018), Lipsius (2018) and Rinz (2020), who calculate *HHI* using employment shares, and in Azar et al. (2020a), who calculate *HHI* using vacancy shares. In addition, *HHI* is widely used in merger control (see the U.S. DoJ and FTC's as well as the European Commission's horizontal merger guidelines) as a measure of market power.

A relevant question is whether industries capture reasonable employment opportunities for workers, and whether workers travel across commuting zone borders to work. In support of that CZs are a reasonable geographical denomination, 92.5% of employees in the pharmacy market work and reside in the same CZ. A commonly used alternative to industry is occupation.¹⁶ In this paper industry is used instead of occupation both because the deregulation took place at the industry level, and because the data lacks complete and consistent occupation information. Out of the employees that worked at a pharmacy in 2004, only one third have worked in a different five-digit industry after 2004. To the extent that industry is too narrow to represent the employment opportunities for workers, *HHI* in both the pre-deregulation and post-deregulation period will be too high; the identification relies on differences in *HHI* over time.

1.3.2 Data

The data is drawn from a panel of annual register data collected by Statistics Sweden. I have access to full-population data, meaning that I can identify educated pharmacists and non-pharmacists working both inside and outside the pharmacy industry. The main dataset is a matched employer–employee register (*RAMS*) that includes firm, workplace and person identifiers as well as gross labor earnings and the months worked at each workplace for all gainfully

¹⁵The measure assumes that pharmacies operated by the same firm in the same LLM do not compete for workers. In support of this, the estimated returns to being a new hire in a pharmacy in the same LLM between 2004 and 2008 are marginal. For the full sample, the wage returns to joining are not statistically different from 0. Restricting to those who join from another pharmacy, the returns are estimated to be 1.46%. Restricting further to those who move from another pharmacy but are not managers, the return is 1.01%.

¹⁶Previous work has defined markets both using geography–industry (Berger et al. 2019, Lipsius 2018 Benmelech et al. 2018 and Rinz 2020) and geography–occupation (Azar et al. 2020a, Azar et al. 2020b and Qiu and Sojourner 2019).

employed individuals. For those employees with more than one workplace, only one workplace per employee and year is kept, defined as the workplace in November where the individual has the highest annual earnings.¹⁷ Using year, firm and workplace identifiers, firm and workplace characteristics such as workplace industry and ownership structure are matched in. Wages and occupations are taken from structural earnings statistics. Using person and year identifiers, the employer-employee data is linked to demographic registers (*Louise* and *Födelseland*) that include variables such as year and country of birth, gender, education level and field of educational specialization. Commuting zones (*lokala arbetsmarknader*) are taken from Statistics Sweden and are matched in based on the municipality of the workplace.¹⁸ Financial data, only available until 2015, is also matched in at the firm and year level.

As outlined in Section 1.3.1, the pharmacy market is delineated from the full-population data using the workplace industry code for dispensing chemists. Pharmacists are identified using information on education. The demographic registers not only has information on the level of education that individuals hold but also detailed information on which field the education is in, based on the Swedish educational nomenclature (*SUN*). Consequently pharmacists can be identified using the individual's educational specialization in pharmacy together with the level of education the individual has.¹⁹

The main wage measure is monthly full-time adjusted wages in Swedish crowns (SEK), measured between September and November each year. Wages are available for all individuals in the public (non-market) sector and for an annual random stratified sample in the private (market) sector.²⁰ The main analysis uses log full-time equivalent wages as the outcome. These data are accurate but, due to the sampling design, incomplete (see Table 1.1 for summary statistics). To ensure the results are not driven by extreme outliers, the data is trimmed so that wages 50% below the 1st percentile or 50% above the 99th percentile of the monthly full-time adjusted national wage distribution every year are excluded from the regressions. Sensitivity analyses, included in the appendix, instead use gross monthly earnings (defined as gross annual labor earnings divided by the number of months worked at the primary em-

¹⁷Using November is in line with Sweden's official statistics, in turn based on ILO's methodology. Before identifying the main workplace, I drop employees that cannot be linked to a physical workplace and therefore obtain a false workplace code. I also exclude self-employed, who have a different employment relationship to employees.

¹⁸The boundaries of commuting zones are revised periodically. To maintain a consistent measure of commuting zones throughout the time period considered, commuting zones from 2013 are used.

¹⁹Pharmacists are defined using education rather than occupation to have complete coverage in the data. Occupation in the registry data is only available for around 50% of workers.

²⁰Approximately 50% of private sector employees are included in the sample. The sample is stratified by industry and firm size, with an oversampling of larger firms.

ployer) from the matched employer-employee data. The earnings measure has complete coverage in the data.²¹

1.3.3 Main sample

The main sample consists of all employees (pharmacists, pharmacy technicians and non-pharmacists) who are employed at a pharmacy as their main place of work. The sample period is restricted to the years 2004 to 2016, which means pre-trends can be analyzed alongside post-deregulation effects. As the employment relationship is different for self-employed, self-employed are excluded from the sample.²²

Table 1.1 includes summary statistics for the full sample period (2004 to 2016) as well as pre-deregulation and post-deregulation. The pharmacy industry is highly skilled and female dominated: 88% are women and 67% have at least post-secondary education. Half the share of employees are educated pharmacists. The statistics also highlight that there have been compositional changes over time that will be important to control for in the empirical analysis. Workers are on average younger and have slightly less tenure and experience in the post-period. The share with post-secondary education has increased over time and the share of foreign born, defined as being born outside of Sweden, has increased sharply from 12% pre-deregulation to 23% post-deregulation.

1.4 Institutional setting and descriptive patterns

1.4.1 Introducing competition on the pharmacy market

Between 1971 and 2009, *Apoteket* (the National Corporation of Swedish Pharmacies) had the exclusive right to retail prescription and non-prescription pharmaceuticals in Sweden. *Apoteket* was a state-owned enterprise whose primary purpose was to ensure a nationwide pharmaceutical supply system. In 2009, a regulatory reform ended the monopoly. The main implication was that *Apoteket* lost its monopoly rights and private firms could enter.

²¹There are two main differences between wages and earnings: (i) due to the sampling design of the official statistics, wages are predominantly available for larger firms, and (ii) earnings are not full-time equivalent. Like for wages, the earnings measure is restricted to drop earnings more than 50% below the 1st percentile or 50% above the 99th percentile of the monthly national wage distribution every year.

²²Self-employment among pharmacists today is low. Goldin and Katz (2016) find that the fraction of pharmacists in the U.S. who are self-employed has decreased from 40% in 1966 to just under 5% in 2011. Figure A.1 shows the share of self-employment among all educated pharmacists in Sweden between 2004 and 2016. Under 4% of pharmacists are self-employed. There is an increase in self-employment in conjunction with the deregulation. This is not explored further in this paper.

	2004	2016	200	4-2008	200	9-2012	201	3-2016
	Mean	Std. dev.	Mean	Std. dev.	Mean	Std. dev.	Mean	Std. dev.
Female	0.88	(0.32)	0.91	(0.29)	0.88	(0.32)	0.86	(0.35)
Age (years)	43.79	(13.96)	45.60	(13.20)	43.32	(14.32)	42.13	(14.23)
Age < 30	0.20	(0.40)	0.16	(0.37)	0.22	(0.41)	0.23	(0.42)
$\mathrm{Age} \geq 50$	0.40	(0.49)	0.47	(0.50)	0.38	(0.49)	0.33	(0.47)
Foreign born	0.17	(0.38)	0.12	(0.32)	0.17	(0.37)	0.23	(0.42)
Post-secondary	0.67	(0.47)	0.64	(0.48)	0.66	(0.47)	0.71	(0.45)
Pharmacist	0.49	(0.50)	0.51	(0.50)	0.48	(0.50)	0.47	(0.50)
Tenured	0.43	(0.50)	0.50	(0.50)	0.41	(0.49)	0.37	(0.48)
Industry experienced	0.73	(0.44)	0.77	(0.42)	0.73	(0.44)	0.69	(0.46)
Non-missing wage	0.69		0.84		0.54		0.67	
Monthly wage (2004 SEK)	26,193	(8, 166)	24,246	(7,404)	26,151	(8,096)	29,083	(8,429)
Monthly earnings (2004 SEK)	23,806	(10,353)	22,097	(9, 198)	23,707	(9,955)	26,059	(11,614)
Number of employee-year obs.	159,863		59,392		49,277		51,194	
<i>Note:</i> The table shows summary statpost-period, split into two parts (2005 tenure at a pharmacy. Industry experit	tistics for all e 9–2012 and 20 enced hold at 1	mployees in th 13–2016). Fore east five years o	e pharmacy in eign born are th of experience fi	dustry for the i nose born in a c rom the pharma	full period (20 country other tl acy industry.	04–2016), the _F han Sweden. Te	re-period (200 nured hold at])4–2008) and the east five years of

Table 1.1. Summary statistics, pharmacy industry

33

The timeline of the deregulation is included in Figure 1.1. A special inquiry was commissioned in December 2006 to submit proposals regarding deregulating the market. *Apoteket*'s exclusive rights were abolished through the implementation of three bills between September 2008 and November 2009. Only the second bill (Regeringen 2009b Prop 2008/09:145) is pivotal to the experiment that this paper relies on. The bill, which passed in parliament in April 2009 and into law in July 2009, made it possible for private firms to operate pharmacies. Following this law change, *Apoteket* sold the majority of its pharmacies to private owners but remained in the market. The first private pharmacies opened to customers in January 2010.²³



Figure 1.1. Timeline of regulatory changes

Entry into the market took place through two channels. First, two thirds of *Apoteket*'s pharmacies were privatized. Second, firms could open new pharmacies, subject to obtaining a permit from the Swedish Medical Products Agency. Only variation from the first channel is used in the empirical strategy. Based on public records, 466 out of *Apoteket*'s 946 pharmacies were sold to private firms and a further 150 pharmacies were transferred to a separate state-run company, *Apoteksgruppen*, where entrepreneurs could enter as majority owners (Nya apoteksmarknadsutredningen 2017). The privatization involved sales of pharmacies in eight clusters during 2009.²⁴ The number of clusters and composition of pharmacies within each cluster was formed by the policymaker. The stated aim was to promote competition in and after the bidding process and to achieve competitive neutrality between public and private owners (Riksrevisionen 2012).

The share of employment at a public owner in the pharmacy industry fell from 100% to 30% and the number of pharmacies increased by over 400 fol-

²³The other two bills are not directly relevant to retail pharmacies. The first bill (Regeringen 2008 Prop 2007/08:142) involves a separate product market – the supply of drugs to hospitals. Prior to 2008, only *Apoteket* or the caregiver could supply inpatient pharmaceuticals. From September 2008, inpatient drugs can be publicly procured from other suppliers. The third bill, implemented in November 2009, implied that certain non-prescription drugs can retail in new locations, such as supermarkets (Regeringen 2009a Prop 2008/09:190). Price competition for non-prescription drugs therefore intensified by breaking up the monopoly and by retailing non-prescription drugs in new locations.

²⁴There were also bids for individual pharmacies in *Apoteksgruppen*. They were first received in 2010 (see Riksrevisionen 2012).
lowing the deregulation. Likewise, the number of firms increased substantially, reaching over 150 within two years of the deregulation. Labor market concentration also fell substantially. Average *HHI* in the pharmacy market, weighted by employment in each local labor market, fell from 1 in prederegulation to just over 0.25 in 2016.

The variation in *HHI* that I use in the empirical strategy is based on observed ownership changes in the data in 2009.²⁵ In this year, most of the privatization occurred but no additional firms had yet entered and no new pharmacies had been opened. Figure 1.2 maps *HHI* by local labor market in 2009. Recall that the industry was a monopoly pre-2009. Thus light-colored areas with relatively low levels of *HHI* in 2009 have experienced the largest reductions in employer concentration, or similarly the largest increases in competition. In line with earlier literature (see Rinz 2020), urban areas are least concentrated post-deregulation and rural areas are most concentrated. Nevertheless, there is substantial variation in *HHI* across local labor markets.²⁶

²⁵I observe firm and workplace identifiers over time, as well as which employees work at each workplace and firm. I neither have access to data on the pharmacies included in each cluster nor the winners of the bids.

 $^{^{26}}$ The reform variation is also illustrated in Figure A.3, which shows the distribution of *HHI* in 2009 across LLMs and employees. Table A.1 includes pre-reform summary statistics separately for local markets where the change in *HHI* is high, medium or low. The groups are similar along many dimensions, like the gender and age composition of the workers, but they also differ along some dimensions, like share of foreign born. LLM fixed effects control for permanent differences between local markets in the empirical estimation.



Figure 1.2. HHI by local labor market (2009)



Figure 1.3. Employment and number of pharmacies in pharmacy industry

1.4.2 Employees

Figure 1.3 shows the number of employees and the number of pharmacies in the pharmacy industry annually between 2004 and 2016, based on data in the main sample. Prior to the deregulation, there were 12,000 individuals working in the industry. This has increased by just under 10%, to 13,000, post-deregulation. The number of pharmacies has increased by around 50%, from 900 to nearly 1,400.²⁷ It follows that the number of employees per pharmacy has decreased and that the market has become fragmented: there are many more pharmacies operated by many more firms, but not equally more employees.

A simple monopsony model predicts that employment should respond positively to reduced monopsony power. This is in line with what we see in the data. Similarly, if the market experienced a market-wide labor demand shock once deregulated, a competitive model would also predict that employment

²⁷The number of pharmacies in the data is similar but generally slightly higher than that reported by the Dental and Pharmaceutical Benefits Agency (*TLV*), see TLV (2018) Figure 2. This could arise if the industry code for dispensing chemists is wider than that used by *TLV*, and/or if some pharmacies organize the pharmacy under more than one workplace identifier. Similarly, the number of employees is higher than that reported by the industry organization *Sveriges Apoteksförening*. In addition to the above, this could for example arise if fewer employee categories are included in the industry organization's reporting.

increases. The latter, however, is unlikely. Labor demand is derived from product demand, and the demand for pharmaceuticals is unlikely to jump discontinuously in 2009. I return to this in Section 1.6.3 below.

Type of pharmacy employees

Pharmacists make up 50% to 60% of the workforce in pharmacies, pharmacy technicians around 25% and non-pharmacists, including managers, the remainder. Pharmacies must have a pharmacist on duty during opening hours. They have an occupational license, issued by the National Board of Health and Welfare.²⁸ In 2016, 64% of pharmacists worked in a pharmacy. The second largest industry for pharmacists at the five-digit level was manufacture of pharmaceutical preparations (6%) and the third largest wholesale of pharmaceutical goods (5%).²⁹ Pharmacy technicians primarily work with sales and advice on non-prescription drugs and retail items, and can assist with dispensing medicines. They are tied to pharmacies – over 95% of pharmacy technicians work there – but are harder to identify in data as they lack unifying educational backgrounds or occupational codes.³⁰ Prior to the deregulation, *Apoteket* internally trained pharmacy technicians. Post-deregulation, there are vocational degrees. In 2016, 62% had upper secondary schooling or less.

1.4.3 Wages

Despite being a state-owned enterprise, employees negotiate wages individually both pre-deregulation and post-deregulation. The pharmacy market is fully covered by collective agreements. Wages are not specified in the collective agreements but are set flexibly in annual wage negotiations between the employee and their manager.

Aggregate changes in wages

The deregulation may on aggregate result in either increases or decreases in wages, as outlined in Section 1.2. To descriptively gauge aggregate changes in wages for pharmacy employees over time, workers in the pharmacy industry are compared either to the school sector (preschools and compulsory schools) or to the labor market as a whole. The school sector is used because spillovers

²⁸Formally, there are two types of pharmacists: those who hold at least a Master's degree in Pharmacy (*apotekare*) and those who hold a Bachelor's degree in Pharmacy (*receptarie*). The legal requirements apply to either type of pharmacist, and there are only small differences in tasks performed. Consequently no distinction is made between the two categories in this paper. ²⁹Table A.2 includes the top 5 industries by employment in 2016 for educated pharmacists.

³⁰From 2014, there is an occupational code for pharmacy technicians. The occupational code prior to 2014 is broader and does not only encompass pharmacy technicians.

to this sector are unlikely, and because it shares important features with the pharmacy industry:³¹

$$\ln(w_{it}) = \alpha_1 Pharmacy_{it} + \alpha_2 Pharmacy_{it} \times Post_t + \lambda_t + \beta X_{it} + \varepsilon_{it}$$
(1.7)

The coefficient of interest is α_2 . This summarizes the aggregate change in wages in pharmacies in the post- compared to pre-period, relative to changes in wages on the whole labor market and in schools. Individual-level controls, X_{it} , for age, gender, foreign-born and education are included in the specification to make the comparison between similar workers. In certain specifications, CZ fixed effects are included to control for permanent regional wage differences, or individual by CZ fixed effects to control for individual heterogeneity in wages. Estimates of α_2 in Table 1.2 indicate that wages on aggregate increased by 2 to 4 percentage points more for workers in pharmacies in the post-period (from 2009 onward) than they did in the rest of the labor market or for individuals working in schools.³²

	(1)	(2)	(3)
Panel A: Full labor market			
Pharmacy \times Post	0.030***	0.034***	0.023***
	(0.008)	(0.009)	(0.002)
R^2	0.394	0.419	0.936
Ν	26,587,246	26,587,246	25,613,134
Panel B: Pharmacies and schools			
Pharmacy \times Post	0.040***	0.042***	0.020***
	(0.006)	(0.006)	(0.002)
R^2	0.646	0.663	0.958
Ν	3,256,605	3,256,605	3,113,725
Year FE	Yes	Yes	Yes
Controls	Yes	Yes	Yes
CZ FE		Yes	
Person \times CZ FE			Yes

Table 1.2. Aggregate changes in ln(wage)

Note: This table provides the estimates of α_2 from estimating equation (1.7). Controls are age (in five categories), gender, foreign born and level of education (in five categories). Standard errors are clustered by commuting zone and reported in parentheses.

³¹The school sector is defined using workplace industry codes 85.321 and 80.100 (SNI92), 80.101 and 80.102 (SNI02) and 85.100 and 85.201 (SNI07). Both public and private organizations run schools. Similar to pharmacies, it is highly skilled and female dominated (see Figure A.2, which compares workers in the pharmacy industry to workers in schools and the full labor market). Teachers are certified and, from 2011, there is an occupational license.

³²Comparing changes in mean (deflated) wages over time gives a similar picture. Wages have increased 2 (3.7) percentage points more in pharmacies than in the full labor market (schools).

1.5 Empirical strategy

The deregulation provides variation in market concentration that is oftentimes hard to observe. While the national pharmacy market was deregulated, the change in local labor market concentration varied over the country. The empirical strategy exploits the fact that the deregulation gave rise to differences in the size of changes in labor market concentration across local labor markets within the pharmacy industry. Recognizing that actual *HHI* is potentially endogenous, it only makes use of the variation in concentration from the privatization of pre-existing pharmacies. The empirical strategy is outlined in detail below.³³

I would like to estimate regressions of the following form to understand the effect that labor market concentration has on wages:

$$\ln(w_{imt}) = \alpha \ln(HHI_{mt}) + \lambda_m + \lambda_t + \beta X_{imt} + \varepsilon_{imt}$$
(1.8)

 λ_m are local labor market (LLM) fixed effects, λ_t are year fixed effects and X_{imt} are additional controls. The concern is that actual HHI_{mt} is potentially endogenous, leading to biased estimates of the coefficient of interest, α . For example, market concentration depends on firms' location choices. If unobserved factors affect both the choice of where to locate pharmacies as well as wages in those locations in ways which are not controlled for, $\hat{\alpha}$ will be biased. Another potential concern is that *HHI* is based on employment shares, which relate to individuals' decisions of where to work. The reform coincides with increased aggregate mobility on the labor market (see Figure 1.7) and wages are a key component in the labor supply decision.³⁴

To address these concerns, I adopt an approach where, instead of using actual changes to concentration, only the change that arises from the privatization of pharmacies that exist when the market is deregulated is used. This is the part of the variation that was within the policymakers' control. Reduced form regressions of the following form are estimated for everyone who is employed at a pharmacy:

$$\ln(w_{imt}) = \gamma [\ln(HHI_{m,2009}) \times Post_t] + \lambda_m + \lambda_t + \beta X_{imt} + \varepsilon_{imt}$$
(1.9)

As above, λ_m are LLM fixed effects and λ_t are year fixed effects. X_{imt} are additional controls for age, gender, foreign-born, level of education, pharmacist, tenure and industry experience. $\ln(HHI_{m,2009})$ is log HHI in 2009 and *Post*_t is a dummy variable equal to one from 2009 onward. Each local labor market

³³The empirical models are estimated in Stata using the Multi-Way Fixed Effects estimator (Correia 2017).

³⁴Schubert et al. (2020) stress that wage–HHI regressions are likely to be biased if workers' outside options are ignored. This is likely to be less of a concern in this setting for at least two reasons. First, pharmacists and pharmacy technicians are closely tied to pharmacies, so outside options that require their skills are limited. Second, permanent differences in outside options across local labor markets are controlled for in the specifications.

receives a constant value of $\ln(HHI_{m,2009})$, such that $\ln(HHI_{m,2009}) \times Post_t$ captures treatment intensity at the local market level from the year of deregulation onward. The specification exploits the full variation in market concentration based on the sale of pre-existing pharmacies as a predictor of the actual change in concentration. $HHI_{m,2009}$ is neither affected by firms' decisions to open up new pharmacies nor to any mobility decisions that workers make post-deregulation, but is instead a function of the deregulatory design (see Section 1.4 for details).³⁵

The local labor market fixed effects control for permanent differences across local markets and year fixed effects for general time trends in industry wages, including those due to inflation. Importantly, national effects of deregulating the market will be absorbed in the time fixed effects. The empirical strategy allows me to capture the effects of changes to labor market concentration without confounding it with effects related to product market concentration, which are likely to be national.³⁶ The local labor market fixed effects are important both because wages are likely to differ throughout the country for reasons unrelated to the reform, and because firms generally want to locate pharmacies in highly populated areas. Finally, as wages are particularly driven by individual characteristics, certain specifications use person by local labor market fixed effects instead of local labor market fixed effects. This keeps composition constant. The identifying variation then comes from workers who have stayed in their local labor market over time and consequently experienced different levels of concentration pre- and post deregulation.

Equation (1.9) is a type of difference-in-difference specification with different treatment intensities that, in addition to correct functional form, relies on parallel trends in log wages across local markets in the absence of the deregulation. To support this assumption, I also estimate event versions of equation (1.9):

$$\ln(w_{imt}) = \sum_{t \neq 2008} \gamma_t \ln(HHI_{m,2009}) \mathbb{1}[year = t] + \lambda_m + \lambda_t + \beta X_{imt} + \varepsilon_{imt} \quad (1.10)$$

Estimating responses by year serve a dual purpose. This both shows whether there are pre-treatment trends in log wages relative to 2008 (the omitted year) and the time trajectories of post-treatment effects. In Section 1.6.3 I include extensive additional robustness checks.

³⁵We may worry that there is some measurement error in employment in 2009 as employment seems to dip slightly in that year, see Figure 1.3. The results are robust to instead calculating HHI based on pharmacies and employment from 2004 but ownership structures from 2009, see Figure B.2.

³⁶Qiu and Sojourner (2019) highlight the importance of distinguishing between concentration in product and in labor markets. Kroft et al. (2020) study the construction industry to find that firms in this industry enjoy rents both due to markups of prices and markdowns of wages.

1.6 Results

1.6.1 Wage effects

Descriptive patterns

Before formally estimating the effect of labor market concentration on wages, I show descriptively how wages have evolved over time in local markets that experienced different changes in employer concentration. Figure 1.4 plots residualized log wages separately by change in *HHI*. Pharmacy workers are split into three categories: those that work in local markets where the change in *HHI* is high, medium or low. Wages are residualized by age, gender, foreign born and level of education in Panel A. To control for general time trends in wages, year fixed effects are also included in Panel B.

The figure suggests that wages evolve in parallel in markets that ended up experiencing different changes in concentration prior to deregulation. Notice that mean wages are not the same even though *HHI* is the same everywhere in the pre-period. Wages are set flexibly, and differences in wages are driven by factors other than employer concentration. Upon deregulation, wages grow faster in markets that experience larger reductions in labor market concentration.



(a) Without year fixed effects (b) With year fixed effects Figure 1.4. Residualized log wages by change in labor market concentration

Note: This plots mean residualized log wages separately by three groups of changes in labor market concentration: "High" (where $HHI_{m,2009}$ is less than or equal to the 25th percentile of the distribution of $HHI_{m,2009}$ by LLM), "Medium" (where $HHI_{m,2009}$ is above the 25th but less than or equal to the 50th percentile), and "Low" (where $HHI_{m,2009}$ above the 50th percentile). In Panel A, log wages are residualized by age, gender, foreign born and level of education. In Panel B, year fixed effects are additionally included.

Estimations

Columns (1) to (3) in Table 1.3 report the results of estimating equation (1.8) in Panel A, and equation (1.9) in Panel C. Columns (4) and (5) include the first stage and IV results of instead using $\ln(HHI_{m,2009}) \times Post_t$ as an instrument for $\ln(HHI_{mt})$. Focusing first on the OLS results, the point estimates are negative and statistically significant at the 1% level. Taken at face value, they imply elasticities of wages with respect to labor market concentration between -0.016

(see 0.003) and -0.048 (se 0.007). The reduced form estimates in Panel C are between -0.022 (se 0.006) and -0.048 (se 0.011), and the elasticities implied by the IV estimation are between -0.020 (se 0.006) and -0.046 (se 0.008).³⁷ The similarity in estimates across estimation techniques arises because the change in market concentration in the year of deregulation is highly predictive of actual changes in market concentration on the pharmacy market. The first stage coefficient is between 1.023 and 1.065, and the F-statistic between 80.45 and 107.21, depending on whether person by LLM fixed effects are used or not. Figure A.4 in the appendix plots actual *HHI* against $\ln(HHI_{m,2009}) \times Post_t$. The two measures are highly correlated: the R^2 of regressing $\ln(HHI_{m,t})$ on $\ln(HHI_{m,2009}) \times Post_t$ is 0.91.

	(1)	(2)	(3)	(4)	(5)
		OLS		Γ	V
Panel A: OLS & IV					
$ln(HHI_{mt})$	-0.045***	-0.048***	-0.016***	-0.046***	-0.020***
	(0.009)	(0.007)	(0.003)	(0.008)	(0.006)
R^2	0.217	0.552	0.924		
Panel B: First stage					
$ln(HHI_{m,2009}) \times Post_t$				1.023***	1.065***
				(0.114)	(0.103)
F-statistic				80.45	107.21
Panel C: Reduced form					
$ln(HHI_m 2009) \times Post_t$	-0.033**	-0.048***	-0.022***		
(11,200)) 1	(0.013)	(0.011)	(0.006)		
R^2	0.216	0.551	0.924		
Ν	110,825	110,722	104,968	110,722	104,968
Year FE	Yes	Yes	Yes	Yes	Yes
LLM FE	Yes	Yes		Yes	
$Person \times LLM \ FE$			Yes		Yes
Controls		Yes		Yes	

Table 1.3. Effect of labor market concentration on ln(wage) – OLS, RF and IV results

Note: This provides the results of estimating equations (1.8) and (1.9) for log wages. OLS and IV results are presented in Panel A. The first stage in Panel B show the results of regressing $\ln(HHI_{mt})$ on the instrument $\ln(HHI_{m,2009}) \times Post_t$ and exogenous regressors. Controls are age (in five categories), gender, foreign born, pharmacist, level of education (in five categories), tenure (in three categories) and industry experience (in three categories). Standard errors are clustered by LLM and reported in parentheses. The F-statistic is the Kleibergen-Paap Wald rk F-statistic.

To put the estimated effects into perspective, a local market that ended up at the 25th percentile of market concentration instead of the 75th percentile following the deregulation (an *HHI* of 0.28 instead of an *HHI* of 1 in the LLM-year distribution) would have wages that are 2.5 to 6 percent higher.

³⁷Similar but somewhat more imprecise results are found when instead using log earnings as the outcome, see Appendix Table A.3.

The negative effect that market concentration has on wages echoes a growing literature in labor economics.³⁸ The results are very similar to those found by Hershbein et al. (2019) and Rinz (2020), who report elasticities between -0.01 and -0.05, and somewhat smaller than those found by Azar et al. (2020a) who report elasticities between -0.03 (OLS) and -0.14 (IV). The somewhat smaller magnitude found in this paper could reflect that the public monopsonist did not depress wages as much as its market power would allow it to. It could also be that labor market institutions, such as the high degree of collective bargaining and the fact that this is a highly specialized group of labor, limit the firm's bargaining power pre-deregulation. In line with this, Jarosch et al. (2019)'s model implies that the elasticity of wages with respect to concentration becomes smaller when worker bargaining power increases, i.e. wages are less sensitive to concentration when workers have more bargaining power. Moreover, the estimates in this paper can be regarded as lower-bound estimates of the effect of labor market concentration on wages, as discussed in Section 1.2.2.

In the remainder of the discussion, I focus on the reduced form results from estimating equation (1.9). To check pre-reform parallel trends in log wages as well as post-reform differences in effects, Figure 1.5 plots estimates of γ_t from equation (1.10). The results support that wages evolve in parallel in local labor markets prior to deregulation, responding positively to the lower concentration post-deregulation.³⁹ As expected, there is no response to the deregulation in 2009: while the deregulation came into force in 2009, the new firms formally began trading in 2010. The positive wage effects rise between 2009 and 2010, and again between 2010 and 2011, and at least at the 5% level they are not equal between 2009 and 2011. From 2011 onward, the estimated effects are relatively stable. This stability over time is also supported by regressions of equation (1.9) when the pre-2009 period is grouped and the post-2009 period is split into two (2010–2012 and 2013–2016). The coefficients on the two sub-periods post-deregulation are not found to be statistically significantly different from each other.

³⁸See Arnold (2020); Azar et al. (2020a); Benmelech et al. (2018); Hershbein et al. (2019); Jarosch et al. (2019); Lipsius (2018); Martins (2018); Prager and Schmitt (2019); Qiu and Sojourner (2019); Rinz (2020); Schubert et al. (2020).

³⁹Figure 1.4 suggests that the evolution of mean wages slows down in 2008. The parallel trends are robust to instead using 2004 as the base year.



Figure 1.5. The effect of labor market concentration on $\ln(wage)$ over time

Note: This plots estimates of γ_i from equation (1.10) with 95% confidence intervals. Panel A also controls for age (in five categories), gender, foreign born, pharmacist, level of education (in five categories), workplace tenure (in three categories) and industry experience (in three categories).

1.6.2 Composition effects

The results in Table 1.3 show that there are differences in point estimates across models. The estimates of γ that are only identified from differences in concentration across LLMs over time, without any controls, are -0.033. When adding controls, the point estimates increase to nearly -0.048. This suggests that the composition of workers bears at least partial importance for the estimated wage effects.⁴⁰ The estimates that use within-individual variation over time and therefore fully control for composition are around -0.02. That these estimates are slightly smaller suggests that high-wage workers become employed in markets where concentration changed more. Put differently, the lower concentration that materializes post-deregulation both results in higher wages for workers conditional on worker quality (seen by the regressions that include individual by local labor market fixed effects), and changes the composition of workers toward higher-paid workers.⁴¹

To gauge how composition changes following the deregulation, Figure 1.6 plots the estimated γ -coefficients from estimating equation (1.9) without controls for five indicator outcomes: female, foreign born, post-secondary education, pharmacist and young workers (defined as being below age 40). The results indicate that there is no statistically significant effect at the 5% level on the share with post-secondary education. Local markets where where concen-

⁴⁰That labor market concentration can affect the types of workers hired is highlighted by Hershbein et al. (2019), who find that local labor market concentration is negatively correlated with wages but positively correlated with skill demand.

⁴¹This result is also supported by the fact that when the sample is restricted to incumbents, the estimated wage effects from reduced labor market concentration are -0.035 (compared to -0.048 for the full sample), in a model that includes year fixed effects, local labor market fixed effects and controls.

tration was reduced the most have experienced faster growth in the share of foreign born and the share of young employees. On the contrary, the share of women and pharmacists has declined more in these markets, though the effect is only statistically significant at the 5% level for women, not pharmacists.



Figure 1.6. Compositional changes

Note: The figure plots estimated γ -coefficients with 95% confidence intervals from the following model for five indicator outcomes: $Y_{imt} = \gamma [\ln(HHI_{m,2009}) \times Post_t] + \lambda_m + \lambda_t + \varepsilon_{imt}$. Young is defined as being below age 40.

1.6.3 Robustness checks

Robustness checks are included in Appendix B. To assess the stability of the estimates, a set of robustness checks estimates equation (1.9) using different samples or controls. The γ -estimates are included in Appendix Table B.1. In column (1) local labor market fixed effects and controls are included and in column (2) individual by local labor market fixed effects are included. Baseline results are shown in Panel A.

We may be worried that the relationship between labor market concentration and wages is biased if labor demand changes in conjunction with concentration. Boal and Ransom (1997) highlight that a negative correlation between labor market concentration and wages cannot be taken as evidence against competitive labor markets unless total labor demand is fixed. If the deregulation coincides with a market-level increase in labor demand, we may see increases in wages as well as the number of firms who enter, leading to lower levels of market concentration. This would bias $\hat{\gamma}$ away from 0.⁴² To alleviate this concern, I include controls for log number of pharmacy employees at the LLM-year level.⁴³ The results of this estimation are included in Panel B of Table B.1. The point estimates at between -0.020 and -0.046 are very similar to the baseline results of -0.022 to -0.048, indicating that the results are not driven by labor demand effects. We also expect demand effects, to the extent that they exist, to be moderated over time. The event analysis presented in Figure 1.5 shows that the wage effects from reduced labor market concentration persist also over the longer term, which supports that the estimated effects do not reflect a demand shock.

In Panel C of Table B.1, controls for log value added per employee at firmlevel are included to remedy the concern that productivity changes coupled with rent-sharing may drive observed wage effects. In Panel D, urban areas (precisely, local markets that encompass the three largest cities in Sweden) are excluded to ensure that the effects are not driven by these markets alone. The share of managers employed in the pharmacy industry increases from 2% to between 6% and 8% post-deregulation. In Panel E, managers are excluded from the sample to ensure that the results are not driven by the increase in the number of managers. In Panel F only the public sector is included. This is to remedy the concern that the effects are driven by the entry of private firms only, that may have different objectives to public firms. Finally, in Panel G controls are included for mean log wages at the CZ and year level, to remedy the concern that wages may in general be growing faster in certain commuting zones. The estimates are remarkably stable across the empirical models, and the result that the reduction in market concentration positively influences wages is robust to these checks. The most conservative estimates that hold composition constant find estimates between -0.013 and -0.023.

While the analysis above displays reassuring pre-trends (see Figure 1.5) and is robust to controlling for log wages in the CZ (see Panel G of Table B.1), we may nevertheless worry that the regressions are picking up spurious changes within these local labor markets rather than effects related to the competitive changes. To alleviate this concern, a placebo analysis is performed where event specification (1.10) is estimated for the school sector. Figure B.1 plots γ_t separately for the school sector and the pharmacy market. $HHI_{m,2009}$ is defined as previously. Reassuringly, the analysis shows that employees in schools have not experienced corresponding wage growth as in the pharmacy industry, supporting that the estimated wage effects are real rather than spuri-

⁴²Given that labor demand is derived from product demand, and that there is little reason to expect demand for pharmaceuticals to be affected by the deregulation, this alleviates the concern that there is a concurrent labor demand shock.

⁴³An alternative is to control for log number of pharmacies per LLM. The correlation between log number pharmacies and log employment is 0.98, thus I do not include both at the same time. Instead controlling for the number of pharmacies also yields statistically significant and negative estimates of employer concentration on wages at the 5% level.

ous. The analysis for the school sector suggests that wages have grown slightly faster over time in markets where concentration has changed more, but these effects are much smaller than those estimated in the pharmacy industry. Taken together, the results support that reducing market concentration leads to a positive wage effect not observed in general in local labor markets.

1.7 Heterogeneity in estimated effects

In this section I explore who benefits most from reduced labor market concentration. First, I study new hires and stayers in the pharmacy industry. I consider how mobility is affected in Section 1.7.1, and how wage effects of reduced labor market concentration differ for new hires and stayers in Section 1.7.1. Second, I consider whether the estimated wage effects differ by individual characteristics such as being female or foreign born in Section 1.7.2.

1.7.1 New hires and stayers

Mobility effects

To understand the types of career options that individuals face, it is instructive to consider transitions on the labor market. The rate at which workers change jobs is informative of the extent of competition between employers, as the ability of workers to leave for another employer limits the wage-setting power that employers hold. Mobility can be measured in different ways. Mobility to firms (where one firm can have many workplaces, or in this case, pharmacies) captures the fraction of employees who change firms or transition from nonemployment, while mobility to workplaces captures the fraction of employees that change workplace (which can either be within the same firm, across firms or from non-employment). For wages, mobility between firms is likely important: changing firms involves changing employers, and is an opportunity to renegotiate wages. Mobility between workplaces can also be important for wages and can represent a career transition.

To gauge whether the opportunity to move has been affected by the deregulation, Figure 1.7 plots the fraction of new hires to a firm (Panel A) or pharmacy (Panel B) across all employees working at a pharmacy.⁴⁴ This shows that mobility in the pharmacy industry as a whole increased following the deregulation. A new hire is defined to be an employee who works at the firm (pharmacy) in year *t* but did not work there the year before, in t - 1. To capture genuine recruits rather than moves that happen due to restructuring, the restriction that it is not a move if over 50% of the employees at the firm (pharmacy) in year *t* come from the same firm (pharmacy) in t - 1 is applied. Prior to the deregulation, the fraction of new firm hires was 10%. Because there was only

⁴⁴New hire is used interchangeably with mover, joiner or recruit in the text.

one firm in the pharmacy industry, this represents transitions from firms in other industries and from non-employment. Firm mobility fluctuates between 25% and 30% post-deregulation, spiking at 31% in 2011. Recruits to pharmacies is higher than to firms, at 20% pre-deregulation. Post-deregulation, recruits to pharmacies increases by around ten percentage points, to 30%. Just like firm mobility, pharmacy mobility peaks in 2011 at 36%, supporting that there is a reshuffling of employees in close conjunction with the deregulation. The increase in mobility between pharmacies is not only driven by the opening of new pharmacies but is also present at pre-existing pharmacies (though the changes are less stark; see Appendix Figure A.5).



Figure 1.7. Mobility in pharmacy industry

Note: Panel A and B show the annual share of employees that join a new firm and join a new pharmacy respectively.

To assess how the fraction of new hires relates to changes in labor market concentration, Table 1.4 reports estimates from equation (1.9) where the outcome is joining a firm (Panel A) or joining a pharmacy (Panel B). In column (1) the full sample is included. The results suggest that the likelihood of joining a new firm is positively affected by reduced labor market concentration: the point estimate of -0.046 implies that the likelihood of moving increases by 0.44 percentage points when *HHI* decreases by 10%. The effects for joining a new pharmacy are not statistically significantly different from zero. This implies that the share who move pharmacy does not increase more in areas where the change in concentration is high compared to low.

New hires can either come from another employer or from non-employment. Manning (2003) explains that a simple measure of monopsony power is the share of recruits from non-employment. If workers who quit are easily replaced by new hires from non-employment, the threat of quitting is limited and we expect employers' wage setting powers to be high. In column (2), attention is limited to the sample of new hires and the outcome is a dummy that indicates whether the new hire came from non-employment. The results suggest that the likelihood of joining a firm from non-employment is negatively related to the reduction in labor market concentration, but the effects are only statistically significant at the 10% level. A 10% decrease in HHI is related to a 0.78 percentage point decrease in the likelihood of joining a firm from nonemployment, or similarly a 0.78 percentage point increase in the likelihood of making a firm-to-firm transition.

	(1)	(2)
	Join	Non-employment
Panel A: Firm hires		
$ln(HHI_{m,2009}) \times Post_t$	-0.046***	0.082*
	(0.008)	(0.042)
R^2	0.367	0.166
Ν	159,633	31,140
Panel B: Pharmacy hires		
$ln(HHI_{m,2009}) \times Post_t$	0.006	-0.015
	(0.007)	(0.019)
R^2	0.454	0.204
Ν	159,633	41,083
Sample	Full	New hires
Year FE	Yes	Yes
LLM FE	Yes	Yes
Controls	Yes	Yes

Table 1.4. Effect of labor market concentration on hiring

Note: This provides estimates of γ from equation (1.9) for the likelihood of joining a new firm (Panel A) or a new pharmacy (Panel B). In column (1) the outcome is an indicator variable for joining the pharmacy or firm from anywhere, and in column (2) it is an indicator for joining from non-employment, conditional on being a new hire to the firm or pharmacy. Column (1) uses the full sample and column (2) only new hires. Controls are included for age, gender, foreign born, pharmacist, level of education, tenure and industry experience. Standard errors are clustered by LLM and reported in parentheses.

Wage effects

Next, I consider wage effects in the pharmacy industry for new hires and stayers. The analysis should be interpreted with caution: it is based on comparisons of a selected sample, and we saw above that the likelihood of moving firm is itself related to labor market concentration. Table 1.5 shows results of estimating equation (1.9) separately for four sub-groups: stayers, new hires, new hires from another firm (which can be inside or outside the pharmacy industry), and new hires from non-employment. As before, a new hire is defined to be an employee who works at the firm in year t but does not work there in t - 1.⁴⁵ A stayer is defined to be someone who is neither a new hire in year t nor in year t + 1.

⁴⁵As previously, the restriction that it is not a move if over 50% of the employees at the firm (pharmacy) in year *t* come from the same firm (pharmacy) in t - 1 is applied.

The results show that stayers (compared to other stayers) experience positive wage gains from the reduced labor market concentration while no statistically significant is found for joiners (compared to other joiners). Put differently, conditional on moving to a new firm, there is no additional return to moving in a local market with relatively low levels of concentration.⁴⁶ The effect for new hires is not distinguishable from zero irrespective of whether they are coming directly from another firm or from non-employment (see columns 3 and 4).⁴⁷

	(1)	(2)	(3)	(4)
	Stay	Join	Firm-to- firm	Non- employment
$ln(HHI_{m,2009}) \times Post_t$	-0.048***	-0.023	0.011	-0.026
. , .	(0.013)	(0.027)	(0.042)	(0.034)
R^2	0.560	0.555	0.539	0.516
Ν	90,554	14,123	9,388	4,732
Year FE	Yes	Yes	Yes	Yes
LLM FE	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

Table 1.5. Effect of labor market concentration on $\ln(wage) - firm$ stayers and new *hires*

Note: This provides estimates of γ from equation (1.9) for different sub-samples, specified in the column headings. Columns 3 and 4 break out the new hires into two mutually exclusive types: those coming from another firm or those coming from non-employment. Controls are included for age, gender, foreign born, pharmacist, level of education, tenure and industry experience. Standard errors are clustered by LLM and reported in parentheses.

Employees' tenure and industry experience is related to the career moves that they make. Figure 1.8 instead considers the effects of labor market concentration separately for employees with different levels of tenure and industry experience. Tenure is defined as the number of years that the employee has spent at the same pharmacy since 2000. Industry experience is analogously defined as the number of years that the employee has spent working in the pharmacy industry since 2000. The results indicate that positive wage effects of lower labor market concentration materialize predominantly for those with longer tenure and more industry experience. This echoes the results for stayers

⁴⁶Similar estimates are found for pharmacy hires and stayers, see Appendix Table A.4.

⁴⁷The results should not be interpreted as saying that there are no wage gains to moving. Table A.5 compares wages for joiners and non-joiners in the same firm or pharmacy before and after deregulation. This abstracts from the role of labor market concentration. Focusing on firm hires in Panel A, the results indicate that new hires to the firm in the pre-period have lower wages than those already working in the firm. In the post-period, the pattern is reversed. Similar results are found when making comparisons in the same pharmacy.

found above. These workers are also likely to be the hardest to replace, given that labor supply in this market is likely to be relatively inelastic.



Figure 1.8. Effect of labor market concentration on $\ln(wage)$ – tenure and industry experience

Note: This plots estimated γ -coefficients with 95% CI from estimating equation (1.9) for subsamples, as specified by the labels. Full results are included in Table A.6.

1.7.2 Individual characteristics

Next, I turn to consider how the wage effects from reduced labor market concentration differ by individual characteristics. Just like the analysis for movers, the results should be interpreted with caution since the deregulation has involved compositional changes (see Section 1.6.2). Figure 1.9 shows the results of estimating equation (1.9) by five sub-groups based on gender, country of birth, level of education, being an educated pharmacist and age. The results are remarkably stable across groups. Overall they suggest that all groups benefit from reduced labor market concentration.

Gender and country of birth: If the monopsonist engages in monopsonistic discrimination (Robinson 1933), which arises from the fact that firms can set lower wages to groups of individuals with more inelastic labor supply to the firm, we expect wages for women and foreign born to respond more strongly to the change in concentration.⁴⁸ The estimates in Figure 1.9 do not, however, suggest that this is the case. There are several plausible explanations for this. A public monopsonist may be less likely to discriminate than

⁴⁸There are several papers on firm-level monopsony and the gender pay gap (see Hirsch 2009, Barth and Dale-Olsen 2009 and Webber 2016) and the immigrant-native pay gap (see Hirsch and Jahn 2015).

a private monopsonist and the collective bargaining institutions may dampen potential discrimination across groups. Moreover, the majority of employees are women both before and after the deregulation.⁴⁹



Figure 1.9. Effect of labor market concentration on $\ln(wage)$ – individual characteristics

Note: This plots estimated γ -coefficients with 95% CI from estimating equation (1.9) for subsamples, as specified by the labels. Full results are included in Table A.7.

Age: Another dimension along which employers potentially discriminate is age. In the gray-colored bars in Figure 1.9, the sample is split into employees in the pharmacy industry that are below 40 or 40 and above. The point estimates suggest that both younger and older workers experience positive wage effects of similar magnitudes from the reduction in market concentration.

Pharmacists and level of education: We expect the deregulation to change competition for specialized labor but not necessarily non-specialized labor. As Boal and Ransom (1997) point out, concentration of hospitals is not likely to monopsonize the market for hospital housekeepers. Turning to the analysis based on education, the results suggest that both those with at least some post-secondary education and those educated as pharmacists have gained similarly from the reduction in concentration. At the same time, the results also support that those with less education and non-educated pharmacists have gained from the deregulation. While initially surprising, this can arise for several reasons. First, as explained in Section 1.4, the group of non-pharmacists consists to a large extent of pharmacy technicians. This group is tied to the pharmacy industry and is consequently likely to experience positive wage gains from re-

⁴⁹This is also in line with Goldin and Katz (2016) who find that the pharmacy profession in the U.S. is among the most egalitarian profession of all.

duced market concentration. The majority of pharmacy technicians have less than post-secondary education. Second, the similarity in point estimates for pharmacists and non-pharmacists could relate to the bargaining power of pharmacists before the deregulation. If their bargaining power is high – pharmacies are not allowed to operate without a pharmacist on site – this can limit the firm's labor market power in the pre-period and attenuate the estimated gain from reduced labor market concentration for pharmacists. In the absence of this bargaining power, we might expect larger returns to reduced labor market concentration for pharmacists than for less specialized labor.

1.8 Conclusion

This paper studies a classic question in novel way. It uses a rare natural experiment that provides geographic variation in employer concentration coupled with rich matched employee-employer data to estimate the effect of reduced labor market concentration on wages. In 2009, the Swedish state-run pharmacy monopoly was deregulated. Entry barriers were lifted and private firms entered the market. Because pharmacists and pharmacy technicians primarily work in pharmacies, the reform provides a stark change in employer concentration for employees in the industry. The change in labor market concentration differs across local markets. Making use of this geographic variation in a difference-in-difference design, elasticities of wages with respect to labor market concentration are estimated to be between -0.02 and -0.05. The empirical strategy relies only on the variation in concentration induced by the sale of pre-existing pharmacies in 2009 to remedy the concern that actual labor market concentration is endogenous. This variation in concentration arises directly from the privatization of pharmacies, a process controlled by the policymaker, but does not relate to the opening of new pharmacies post-reform. The positive wage effects from reduced labor market concentration are robust to extensive checks and are not observed in a similar, but unaffected, industry.

Regarding who benefits from reduced labor market concentration, similar positive wage returns are found for employees of different age, country of birth, gender, educational level and educational specialization. Effects differ for individuals who make different career moves: the positive wage effects primarily arise for stayers, but not new hires, and those with longer tenure and more experience. This does not mean that wages do not respond positively to changing employer. Rather, conditional on changing employer, the results suggest that it does not matter for wages whether the move is made in a local market where the change in labor market concentration is high or low. The result should be interpreted with caution, however, as the likelihood of moving is itself affected by the deregulation.

Overall, the paper lends support to the growing literature that finds that labor market concentration can matter for workers' wages. It finds that wages respond positively to reduced employer concentration in a context with high industry-specificity in skills. This is likely to be true in many other settings where labor is similarly tied, such as teachers or health professionals.

References

- Arnold, D. (2019). "The Impact of Privatization of State-Owned Enterprises on Workers". Unpublished manuscript.
- (2020). "Mergers and Acquisitions, Local Labor Market Concentration, and Worker Outcomes". Job Market Paper.
- Autor, D., D. Dorn, L. F. Katz, C. Patterson, and J. Van Reenen (2020). "The Fall of the Labor Share and the Rise of Superstar Firms". *Quarterly Journal* of Economics 135.2, pp. 645–709.
- Azar, J., I. Marinescu, and M. Steinbaum (2020a). "Labor Market Concentration". *Journal of Human Resources* forthcoming.
- Azar, J., I. Marinescu, M. Steinbaum, and B. Taska (2020b). "Concentration in US Labor Markets: Evidence from Online Vacancy Data". *Labour Economics* 66.
- Barkai, S. (2020). "Declining Labor and Capital Shares". *Journal of Finance* 75.5, pp. 2421–2463.
- Barth, E. and H. Dale-Olsen (2009). "Monopsonistic Discrimination, Worker Turnover, and the Gender Wage Gap". *Labour Economics* 16, pp. 589–597.
- Benmelech, E., N. Bergman, and H. Kim (2018). Strong Employers and Weak Employees: How Does Employer Concentration Affect Wages? Tech. rep. 24307. National Bureau of Economic Research.
- Berger, D., K. Herkenhoff, and S. Mongey (2019). *Labor Market Power*. Tech. rep. 25719. National Bureau of Economic Research.
- Berry, S., M. Gaynor, and F. Scott Morton (2019). "Do Increasing Markups Matter? Lessons from Empirical Industrial Organization". *Journal of Economic Perspectives* 33.3, pp. 44–68.
- Bertrand, M. and F. Kramarz (2002). "Does Entry Regulation Hinder Job Creation? Evidence from the French Retail Industry". *The Quarterly Journal* of Economics 117.4, pp. 1369–1413.
- Bhaskar, V., A. Manning, and T. To (2002). "Oligopsony and Monopsonistic Competition in Labor Markets". *The Journal of Economic Perspectives* 16.2, pp. 155–174.
- Black, S. and P. Strahan (2001). "The Division of Spoils: Rent-Sharing and Discrimination in a Regulated Industry". *American Economic Review* 91.4, pp. 814–831.
- Boal, W. and M. Ransom (1997). "Monopsony in the Labor Market". *Journal* of *Economic Literature* 35.1, pp. 86–112.
- Bunting, R. L. (1962). *Employer Concentration in Local Labor Markets*. University of North Carolina Press.

- Burdett, K. and D.T. Mortensen (1998). "Wage Differentials, Employer Size, and Unemployment". *International Economic Review* 39.2, pp. 257–273.
- Card, D. (1998). "Deregulation and Labor Earnings in the Airline Industry". *Regulatory Reform and Labor Markets*, pp. 184–247.
- Card, D., A. R. Cardoso, J. Heining, and P. Kline (2018). "Firms and Labor Market Inequality: Evidence and Some Theory". *Journal of Labor Economics* 36.S1, pp. 13–70.
- Correia, S. (2017). *Linear Models with High-Dimensional Fixed Effects: An Efficient and Feasible Estimator*. Tech. rep. Working Paper.
- Currie, J., M. Farsi, and W. B. Macleod (2005). "Cut to the Bone? Hospital Takeovers and Employment Contracts". *Industrial and Labor Relations Review* 58.3, pp. 471–493.
- De Loecker, J., J. Eeckhout, and G. Unger (2020). "The Rise of Market Power and the Macroeconomic Implications". *Quarterly Journal of Economics* 135.2, pp. 561–644.
- Dube, A., J. Jacobs, S. Naidu, and S. Suri (2018). Monopsony in Online Labor Markets. Tech. rep. 24416. National Bureau of Economic Research.
- Earle, J. S and S. Shpak (2019). "The Impact of Privatization on Employment and Earnings". *IZA World of Labor*.
- Falch, T. (2010). "The Elasticity of Labor Supply at the Establishment Level". *Journal of Labor Economics* 28.2, pp. 237–266.
- Goldin, C. and L. Katz (2016). "A Most Egalitarian Profession: Pharmacy and the Evolution of a Family-Friendly Occupation". *Journal of Labor Economics* 34.3, pp. 705–746.
- Haskel, J. and S. Szymanski (1993). "Privatization, Liberalization, Wages and Employment: Theory and Evidence for the UK". *Economica* 60, pp. 161–182.
- Hensvik, L. (2012). "Competition, Wages and Teacher Sorting: Lessons Learned From a Voucher Reform". *The Economic Journal* 122.561, pp. 799–824.
- Hershbein, B., C. Macaluso, and C. Yeh (2019). "Concentration in U.S. Local Labor Markets: Evidence from Vacancy and Employment Data". Unpublished manuscript.
- Hirsch, B. (2009). "The Gender Pay Gap Under Duopsony: Joan Robinson Meets Harold Hotelling". Scottish Journal of Political Economy 56.5, pp. 543– 558.
- Hirsch, B. and E. J. Jahn (2015). "Is There Monopsonistic Discrimination Against Immigrants?" *ILR Review* 68.3, pp. 501–528.
- Hirsch, B. T. (1988). "Trucking Regulation, Unionization, and Labor Earnings: 1973-85". *Journal of Human Resources* 23.3, pp. 296–319.
- Hirsch, B. T. and D. A. Macpherson (2000). "Earnings, Rents, and Competition in the Airline Labor Market". *Journal of Labor Economics* 18.1, pp. 125–155.

- Jarosch, G., J. S. Nimczik, and I. Sorkin (2019). *Granual Search, Market Structure, and Wages*. Tech. rep. 26239. National Bureau of Economic Research.
- KKV (2013). Swedish Competition Authority's Decision in Oriola KD's acquisition of Medstop. 246/2013.
- Kroft, K., Y. Luo, M. Mogstad, and B. Setzler (2020). Imperfect Competition and Rents in Labor and Product Markets: The Case of the Construction Industry. Tech. rep. 666. University of Toronto, Department of Economics.
- Lamadon, T., M. Mogstad, and B. Setzler (2019). Imperfect Competition, Compensating Differentials and Rent Sharing in the U.S. Labor Market. Tech. rep. 25954. National Bureau of Economic Research.
- Lergetporer, P., J. Ruhose, and L. Simon (2018). *Entry Barriers and the Labor* Market Outcomes of Incumbent Workers: Evidence from a Deregulation Reform in the German Crafts Sector. Tech. rep. 7274. CESifo.
- Lipsius, B. (2018). "Labor Market Concentration Does Not Explain the Falling Labor Share". Job Market Paper.
- Manning, A. (2003). *Monopsony in Motion: Imperfect Competition in Labor Markets*. Princeton, New Jersey: Princeton University Press.
- (2011). "Imperfect Competition in the Labor Market". *Handbook of Labor Economics*. Vol. 4, pp. 973–1041.
- (2020). "Monopsony in Labor Markets: A Review". ILR Review, pp. 1–24.
- Marinescu, I. and H. J. Hovenkamp (2018). "Anticompetitive Mergers in Labor Markets". Unpublished manuscript.
- Marinescu, I., I. Ouss, and L.-D. Pape (2020). Wages, Hires, and Labor Market Concentration. Tech. rep. 13244. IZA.
- Martins, P. S. (2018). Making Their Own Weather? Estimating Employer Labour Market Power and Its Wage Effects. Tech. rep. 95. CGR.
- Matsudaira, J. (2014). "Monopsony in the Low-Wage Labor Market? Evidence from Minimum Nurse Staffing Regulations". *The Review of Economics and Statistics* 96.1, pp. 92–102.
- Megginson, W. L. and J. M. Netter (2001). "From State to Market: A Survey of Empirical Studies on Privatization". *Journal of Economic Literature* 39.2, pp. 321–389.
- Naidu, S. and E. A. Posner (2019). "Labor Monopsony and the Limits of the Law". Unpublished manuscript.
- Naidu, S., E. A. Posner, and E. G. Weyl (2018). "Antitrust Remedies for Labor Market Power". Unpublished manuscript.
- Nya apoteksmarknadsutredningen (2017). Kvalitet och säkerhet på apoteksmarknaden (Quality and safety in the pharmacy market) (SOU 2017:15).
- Olsson, M. and J. Tåg (2018). *What is the Cost of Privatization for Workers?* Tech. rep. 1201. IFN.
- Peoples, J. (1998). "Deregulation and the Labor Market". *Journal of Economic Perspectives* 12.3, pp. 111–130.

- Prager, E. and M. Schmitt (2019). *Employer Consolidation and Wages: Evidence from Hospitals*. Tech. rep. Washington Center for Equitable Growth.
- Qiu, Y. and A. Sojourner (2019). *Labor-Market Concentration and Labor Compensation*. Tech. rep. 12089. IZA.
- Ransom and Sims (2010). "Estimating the Firm's Labor Supply Curve in a "New Monopsony" Framework: Schoolteachers in Missouri". *Journal of Labor Economics* 28.2, pp. 331–355.
- Regeringen (2008). Sjukhusens läkemedelsförsörjning (Hospitals' supply of medicine). Regeringens proposition 2007/2008:142.
- (2009a). Handel med vissa receptfria läkemedel (Trade in some non prescription medicines). Regeringens proposition 2008/09:190.
- (2009b). Omreglering av apoteksmarknaden (Reregulation of the pharmacy market). Regeringens proposition 2008/09:145.
- Riksrevisionen (2012). Statens försäljningar av apotek (The state's sales of pharmacies). RIR 2012:10.
- Rinz, K. (2020). "Labor Market Concentration, Earnings, and Inequality". *Journal of Human Resources* forthcoming.
- Robinson, J. (1933). *The Economics of Imperfect Competition*. London: Macmillan.
- Rose, N. (1987). "Labor Rent-Sharing and Regulation: Evidence from the Trucking Industry". *Journal of Political Economy* 95.6, pp. 1146–1178.
- Schubert, G., A. Stansbury, and B. Taska (2020). "Monopsony and Outside Options". Unpublished manuscript.
- Staiger, D., J. Spetz, and C. Phibbs (2010). "Is There Monopsony in the Labor Market? Evidence from a Natural Experiment". *Journal of Labor Economics* 28.2, pp. 211–236.
- Sullivan, D. (1989). "Monopsony Power in the Market for Nurses". *Journal of Law and Economics* 32, pp. 135–178.
- TLV (2018). 2017 års uppföljning av apoteksmarknadens utveckling (2017 follow-up of the development on the pharmacy market). 286/2018.
- Webber, D. A. (2016). "Firm-Level Monopsony and the Gender Pay Gap". *Industrial Relations* 55.2, pp. 323–345.

Appendix A: Additional description and results



Figure A.1. Self-employment among pharmacists

Note: The figure shows the share of all educated pharmacists that are self-employed.



(c) Share female

Figure A.2. Worker composition in pharmacy, school sector and whole labor market



(a) LLMs

(b) Employment



Note: The figure shows the distribution of $HHI_{m,2009}$ across LLMs (Panel A) and employees (Panel B). The three lines mark the 25th, 50th and 75th percentiles of the distribution of $HHI_{m,2009}$ across LLMs. They are 0.306, 0.433 and 1 respectively.



Figure A.4. Scatter of $\ln(HHI_{mt})$ vs. $\ln(HHI_{m,2009}) \times Post_t$



Figure A.5. Share new hires to pharmacy (pre-reform pharmacies)

		HHI _{m,2009} interval	
	\leq 25th	>25 th & \leq 50th	>50th
	(high)	(medium)	(low)
Panel A: Characteristics of LLMs			
Employment	433.44	138.68	34.64
Pharmacies	31.16	12.38	3.59
Population ('000)	341.19	100.70	25.08
$HHI_{m,2009}$	0.25	0.35	0.76
$HHI_{m,2009}$ (empl. weighted)	0.25	0.35	0.55
Number of LLM-year obs.	95	85	185
Number of LLMs	19	17	37
Panel B: Characteristics of employees			
Female	0.91	0.91	0.89
Age (years)	45.64	45.36	45.80
Age < 30	0.16	0.17	0.17
Age ≥ 50	0.47	0.46	0.49
Foreign born	0.14	0.09	0.04
Post-secondary	0.66	0.61	0.58
Pharmacist	0.52	0.50	0.48
Tenured	0.50	0.47	0.52
Industry experienced	0.79	0.76	0.71
Monthly wage (2004 SEK)	24,650	23,492	23,102
Monthly earnings (2004 SEK)	22,459	21,449	20,975
Number of employee-year obs.	41,177	11,788	6,408
Number of employees (2008)	8,182	2,366	1,273

 Table A.1. Means by change in labor market concentration (2004–2008)

Note: The table shows means across LLMs (panel A) or employees (panel B) for the prederegulation period (2004 to 2008), separately by three groups of changes in labor market concentration: "High" (where $HHI_{m,2009}$ is less than or equal to the 25th percentile of the distribution of $HHI_{m,2009}$ by LLM), "Medium" (where $HHI_{m,2009}$ is above the 25th but less than or equal to the 50th percentile), and "Low" (where $HHI_{m,2009}$ above the 50th percentile). The 25th (50th) percentile of $HHI_{m,2009}$ is 0.306 (0.433). Foreign born are born in a country other than Sweden. Tenured hold at least five years of tenure at a pharmacy. Industry experienced hold at least five years of experience from the pharmacy market. $HHI_{m,2009}$ (empl. weighted) is the mean value of HHI 2009, weighting by employment in 2008.

 Table A.2. Top 5 industries for pharmacists (2016)

	%
Dispensing chemist	63.90
Manufacture of pharmaceutical preparations	6.16
Wholesale of pharmaceutical goods	4.61
Specialized hospital somatic activities	3.60
Inspection, control, permit & licensing activities of central & local gov't	3.16

Note: The table shows the top 5 industries for pharmacists in Sweden in 2016 by share of employment. Pharmacists are identified by their educational level and specialization. Industries are defined by five-digit SNI codes.

Table A.3. The effect of market concentration on $\ln(earnings) - OLS$, RF and IV results

	(1)	(2)	(3)	(4)	(5)
		OLS		Ι	V
Panel A: OLS & IV					
$ln(HHI_{mt})$	-0.035**	-0.051***	-0.034***	-0.029**	-0.019*
	(0.014)	(0.012)	(0.012)	(0.013)	(0.010)
R^2	0.082	0.306	0.726		
Panel B: First stage					
$ln(HHI_{m,2009}) \times Post_t$				1.035***	1.063***
				(0.108)	(0.091)
F-statistic				91.66	136.80
Panel C: Reduced form					
$ln(HHI_m 2009) \times Post_t$	-0.003	-0.030**	-0.020*		
(,,,,200)) , ,	(0.012)	(0.015)	(0.012)		
R^2	0.081	0.305	0.726		
N	137,664	137,511	131,149	137,511	131,149
Year FE	Yes	Yes	Yes	Yes	Yes
LLM FE	Yes	Yes		Yes	
Person \times LLM FE			Yes		Yes
Controls		Yes		Yes	

Note: This table provides the results of estimating equations (1.8) and (1.9) for the outcome log earnings. OLS and IV results are presented in Panel A. The first stage in Panel B show the results of regressing $\ln(HHI_{mt})$ on the instrument $\ln(HHI_{m,2009}) \times Post_t$ and exogenous regressors. Controls are age (in five categories), gender, foreign born, pharmacist, level of education (in five categories), tenure (in three categories) and industry experience (in three categories). Standard errors are clustered by LLM and reported in parentheses. The F-statistic is the Kleibergen-Paap Wald rk F-statistic.

	(1)	(2)	(3)	(4)
	Stay	Join	Plant-to- plant	Non- employment
$ln(HHI_{m,2009}) \times Post_t$	-0.049***	-0.023	-0.022	-0.041
. , .	(0.014)	(0.015)	(0.013)	(0.028)
R^2	0.569	0.540	0.532	0.513
Ν	79,208	21,139	16,679	4,458
Year FE	Yes	Yes	Yes	Yes
LLM FE	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes

Table A.4. Effect of labor market concentration on $\ln(wage)$ – pharmacy stayers and new hires

Note: This provides estimates of γ from equation (1.9) for the outcome ln(*wage*) for different sub-samples, specified in the column headings. A joiner is someone who is working at the pharmacy in year *t* but not in year *t*-1, while a stayer is someone who is neither a joiner in year *t* or *t*+1 (see text for full details). Columns 3 and 4 break out the new hires into two mutually exclusive types: those coming from another pharmacy or those coming from non-employment. Controls are included for age, gender, foreign born, pharmacist, level of education, tenure and industry experience. Standard errors are clustered by LLM and reported in parentheses.

	(1)	(2)
Panel A: Firm hires		
Join firm	-0.064***	-0.047***
	(0.004)	(0.006)
Join firm \times Post	0.070***	0.058***
	(0.008)	(0.004)
R^2	0.545	0.573
Ν	110,722	93,872
Year \times firm FE	Yes	Yes
Panel B: Pharmacy hires		
Join pharmacy	-0.014***	-0.007**
	(0.003)	(0.003)
Join pharmacy \times Post	0.026***	0.022***
	(0.008)	(0.007)
R^2	0.674	0.691
Ν	109,778	92,555
$\text{Year} \times \text{pharmacy FE}$	Yes	Yes
Sample	Full	Incumbents
Controls	Yes	Yes

Table A.5. Wages – new hires vs. non-new hires

Note: The table compares wages for new hires and non-new hires in the same firm or pharmacy in the pharmacy industry. It provides estimates of θ_1 and θ_2 from estimating $\ln(w_{it}) = \theta_1 Join_{ipt} + \theta_2 Join_{ipt} \times Post_t + \lambda_{pt} + \beta X_{it} + \varepsilon_{it}$, where *p* is either the firm or the pharmacy. A joiner to a firm (pharmacy) is someone who is working at the firm (pharmacy) in year *t* but not in year *t*-1 (see text for full details). In column (1) the full sample is included, while in column (2) only incumbents are included, defined to be employees who are working in the pharmacy industry in the pre-period. Controls are included for age (in five categories), gender, foreign born, pharmacist, level of education (in five categories), tenure (in three categories) and industry experience (in three categories). Standard errors are clustered by LLM and reported in parentheses.

	(1)	(2)	(3)
Tenure (years):	<3	3–4	≥ 5
$ln(HHI_{m,2009}) \times Post_t$	-0.010	-0.036**	-0.049***
	(0.017)	(0.015)	(0.018)
Mean log wage (pre)	10.05	10.08	10.1
R^2	0.544	0.566	0.566
Ν	34,007	22,190	54,525
Industry experience (years):	<3	3–4	≥ 5
$ln(HHI_{m,2009}) \times Post_t$	-0.006	-0.042	-0.050***
	(0.032)	(0.027)	(0.013)
Mean log wage (pre)	9.88	9.93	10.12
R^2	0.502	0.542	0.554
Ν	9,375	12,200	89,145
Year FE	Yes	Yes	Yes
LLM FE	Yes	Yes	Yes
Controls	Yes	Yes	Yes

Table A.6. Heterogeneous effects of labor market concentration on $\ln(wage)$ – tenure and experience

Note: This provides estimates of γ from equation (1.9) for different sub-samples, specified in the column headings. Controls are age (in five categories), gender, foreign born, pharmacist, level of education (in five categories), tenure (in three categories) (only in the bottom panel) and industry experience (in three categories) (only in the top panel). Standard errors are clustered by LLM and reported in parentheses.

	(1)	(2)	(3)	(4)	(5)	(9)
		Female	Foi	reign born		Young
	ou	yes	ou	yes	ou	yes
$ln(HHI_{m,2009}) \times Post_t$	-0.067	-0.047***	-0.042***	-0.037*	-0.045***	-0.042**
	(0.052)	(0.008)	(0.010)	(0.019)	(0.014)	(0.017)
Mean log wage (pre)	10.18	10.08	10.09	10.04	10.14	9.96
R^2	0.528	0.571	0.553	0.593	0.538	0.569
N	11,094	99,626	92,654	18,067	71,109	39,612
	Post	t secondary	Pł	narmacist		
	ou	yes	ou	yes		
$ln(HHI_{m,2009}) \times Post_t$	-0.042**	-0.052***	-0.046**	-0.043***		
	(0.016)	(0.011)	(0.019)	(0.007)		
Mean log wage (pre)	9.88	10.19	9.93	10.22		
R^2	0.480	0.391	0.493	0.400		
Ν	35,443	75,279	52,975	57,747		
Year FE	Yes	Yes	Yes	Yes	Yes	Yes
LLM FE	Yes	Yes	Yes	Yes	Yes	Yes
Controls	Yes	Yes	Yes	Yes	Yes	Yes
<i>Note:</i> This provides estimates age 40. Controls are age (i experience (in three categor	ates of γ from equation in five categories), ites). Standard error	ation (1.9) for differen gender, foreign born, J prs are clustered by LL	tt sub-samples, speci- pharmacist, level of e. M and reported in par	fied in the column hes ducation (in five categ rentheses.	adings. Young is defi ories), tenure (in thre	ined to be someone below se categories) and industry

Appendix B: Robustness checks



Figure B.1. Placebo analysis of the effect of market concentration on wages

Note: The figure plots γ_t from Equation (1.10) with 95% confidence intervals for regressions that are run separately for the pharmacy industry and for the school sector (preschools and compulsory schools). The outcome is log wages. Included in X_{imt} are age (in five categories), gender, foreign born and level of education (in five categories). Standard errors are clustered by LLM.



Figure B.2. The effect of $\ln(HHI_{m,2009})$ on $\ln(wage) - 2004$ pharmacies

Note: The figure plots γ_t from Equation (1.10) with 95% confidence intervals. *HH1* is based on employment and pharmacies from 2004 and ownership structures from 2009. Included in X_{intt} are age (in five categories), gender, foreign born, pharmacist, level of education (in five categories), tenure (in three categories) and industry experience (in three categories). Standard errors are clustered by LLM.

	(1)	(2)	
Panel A: Baseline			
$ln(HHI_{m,2009}) \times Post_t$	-0.048***	-0.022***	
, , ,	(0.011)	(0.006)	
R^2	0.551	0.924	
Ν	110,722	104,968	
Panel B: Control for nr emplo	vyees		
$ln(HHI_{m,2009}) \times Post_t$	-0.046***	-0.020***	
	(0.010)	(0.006)	
R^2	0.551	0.924	
Ν	110,722	104,968	
Panel C: Control for value ad	ded		
$ln(HHI_{m,2009}) \times Post_t$	-0.040***	-0.017***	
,	(0.011)	(0.006)	
R^2	0.544	0.926	
Ν	99,052	93,588	
Panel D: Omit urban areas			
$ln(HHI_{m,2009}) \times Post_t$	-0.039***	-0.014*	
- , -	(0.011)	(0.008)	
R^2	0.607	0.921	
Ν	58,254	55,326	
Panel E: No managers			
$ln(HHI_{m,2009}) \times Post_t$	-0.051***	-0.023***	
,	(0.010)	(0.006)	
R^2	0.566	0.913	
Ν	104,041	98,272	
Panel F: Public sector only			
$ln(HHI_{m,2009}) \times Post_t$	-0.051***	-0.013***	
. , .	(0.019)	(0.004)	
R^2	0.541	0.945	
Ν	87,237	82,697	
Panel G: Control for mean we	ige		
$ln(HHI_{m,2009}) \times Post_t$	-0.047***	-0.021***	
,	(0.011)	(0.006)	
R^2	0.551	0.924	
N	110,722	104,968	
Year FE	Yes	Yes	
LLM FE	Yes	**	
Person \times LLM FE		Yes	
Controls	Yes		

Table B.1. *Robustness – Effect of labor market concentration on* ln(*wage*)

Note: This provides robustness checks for estimating equation (1.9). Column (1) controls for age, gender, foreign born, pharmacist, education, tenure and industry experience. Both columns control for log number of employees per LLM (Panel B), log value added per employee (Panel C), and mean log wages on the whole labor market by CZ and year (Panel G). Value added is available until 2015. Urban areas are LLMs that encompass Stockholm, Gothenburg and Malmö (the three largest cities in Sweden). There is some measurement error which reports a too high share in the public sector particularly between 2009 and 2011. Omitting these years leads to somewhat larger point estimates in absolute terms. Standard errors are clustered by LLM and reported in parentheses.
2. Firms, Productivity, and the Immigrant-Native Earnings Gap

Co-authored with Olof Åslund, Cristina Bratu and Stefano Lombardi

Acknowledgments: We thank Lena Hensvik, Oskar Nordström Skans, Rune Vejlin as well as seminar participants at IFAU and Uppsala University for helpful comments on this paper.

2.1 Introduction

The extent, sources, and remedies of immigrant labor market disadvantages seen in many Western economies are topics of intense scholarly and political debates. A vast literature documents wage and employment gaps relative to native workers, greater among recent migrants but present for decades.¹ Lack of country-specific human capital, such as language skills, is a commonly proposed explanation to why outcomes are particularly poor among the recently arrived. Other sources of inequalities are less immigrant-specific, but highly relevant. Substantial bodies of work investigate factors like employer discrimination, residential segregation, and the importance of networks and contacts.²

A much smaller but growing literature considers the role of firms and their hiring and pay-setting practices. These studies are partly motivated by two empirical observations: (i) there are significant and growing differences in wages and earnings across firms and establishments (Nordström Skans et al. 2007; Card et al. 2013; Barth et al. 2016), and firm practices increase inequality across groups in general (Card et al. 2016; Card et al. 2018; Gerard et al. 2018); (ii) there is substantial origin-based workplace segregation, which is also correlated with economic outcomes of individuals and groups (Hellerstein and Neumark 2008; Åslund and Nordström Skans 2010).

This study focuses on a particular aspect of firms, namely firm productivity. Consistent with models of rent-sharing that imply a positive relationship between wages and firm productivity (Manning 2011), our population-wide linked employer-employee data for Sweden show that persistent measures of firm value added per worker are strongly related to firm-level earnings. We study how this measure of productivity relates to the allocation and workplace mobility of immigrant and native workers, the group-specific earnings gains from working at more productive firms, and how these factors relate to overall immigrant-native earnings gaps.

Sweden provides an interesting case for several reasons. First, Sweden is a diverse country, thanks largely to its immigration. Over the past decades the country has experienced substantial and varied economic and humanitarian immigration, bringing the fraction of foreign-born close to 20 percent in 2020 (SCB 2020). Second, the overall immigrant-native employment differentials are among the greatest in the OECD, and our data show that the raw earnings ratio decreased from about 0.92 to 0.86 between 1998 and 2017. However, in line with existing evidence on within- and across-firm wage dispersion, the within-firm earnings gap is smaller and has not increased over time. Third, the country's low wage dispersion and high degree of unionization and collective bargaining may imply different roles for firm policies than in less regulated labor markets.

¹See Kerr and Kerr (2011), Borjas (2014), and Duleep (2015) for overviews.

²See e.g. Neumark (2018) on discrimination, Gobillon et al. (2007) and Chetty et al. (2020) on segregation, and Dustmann et al. (2016) on networks.

Previous studies show that between-workplace variation explains significant shares of immigrant-native earnings gaps (Barth et al. 2012; Damas de Matos 2017; Dostie et al. 2020). There is also evidence that workplaces are related to the assimilation process. Eliasson (2013) finds that most earnings convergence occurs within establishments rather than through transitions between workplaces. Ansala et al. (2020) find that workplace conditions are strongly related to entry job earnings and subsequent performance among immigrants. Arellano-Bover and San (2020) find that recent migrants tend to work in low-paying firms, while access to higher-paying firms over time explains a significant fraction of immigrant-native pay differences.³

In addition to an impact of sorting over "good" and "bad" firms, the immigrant native earnings gap can also be influenced by systematic across-group differences in the premium from working in a specific type of firm. The latter phenomenon can for example arise if firms have more market power over immigrants than natives.⁴ Manning (2020) highlights that when labor markets are less competitive, wages will be more closely linked to reservation wages than to worker productivity. Hirsch and Jahn (2015) find that immigrants supply labor to the firm less elastically than natives. Bassier et al. (2020) show that the degree of monopsony power is higher in low-wage labor markets.

To capture persistent differences in firm productivity, we rank firms based on average productivity over our sample period, 1998 to 2017. Average productivity comes from a regression of log value added per worker on firm and year fixed effects. The procedure allows us to classify firms into a tractable number of groups, and is similar in spirit to Bonhomme et al. (2019) who bin firms into classes via k-means clustering. We use productivity percentiles instead of clusters, and argue that there is value in basing the analysis on an easily observable measure of productivity. We find no indication that this measure is influenced by the composition of workers in the firm, including the share of immigrants the firm employs.

We begin by documenting the association between productivity and earnings at the firm level. Based on the firm productivity ranking, we find a positive and gradually increasing relationship between firm productivity and earnings starting from the 20th percentile of the firm productivity distribution. Using an AKM model (Abowd et al. 1999) to control for worker heterogeneity, we show an almost linear association between estimated firm earnings premiums and the productivity rank across the entire distribution of firms. AKM results also suggest that firm effects explain more of the earnings variation among immigrants than natives,⁵ and that there is little sorting of high-earning migrants

³Orefice and Peri (2020) find that immigration can also drive positive assortative matching between workers and firms, with consequences for the overall wage structure.

⁴Closely linked are mechanisms related to employer discrimination and immigrants having worse bargaining positions.

⁵When we estimate AKM models separately for immigrants and natives, firm effects account for 11% (22%) of native (immigrant) earnings variation.

into high-earning firms. In other words, where you work appears to be even more important for immigrant workers than for native workers.

When looking at the distribution of workers across firm types, we find that foreign-born workers are over-represented in low-productive firms and under-represented in the upper part of the productivity distribution relative to natives. Comparing the 1998–2009 period to the 2010–2017 period, we find that this sorting has become stronger over time. This pattern holds also when controlling for compositional changes in the immigrant population, even though the increased under-representation in high-productive firms is related to changes in source countries. In general, people born outside Western countries (and thus less likely to be economic migrants) are relatively more concentrated in low-productive firms.

Average earnings are higher among natives than immigrants in all firm productivity deciles. The raw difference is greater where the immigrant share of the workforce is higher (about 15 log points in decile 2 to 4, 16% foreignborn) compared to the highest deciles (5 log points, 8%). Estimates controlling for worker fixed effects suggest that the earnings returns to working in a firm of high persistent productivity are substantial and positive for both groups, but greater for immigrants. For example, for natives there is a 7 log point difference between working in the fifth compared to the first decile. For immigrants, the difference is over 10 log points. Relating within-employment spell (and thus within-firm) variation in value added per worker to individual earnings gives a similar picture. A doubling of firm value added per worker is estimated to raise native earnings by around 2% and immigrant earnings by around 3%. The differential returns are primarily found at lower levels of firm productivity and are primarily driven by immigrants from non-Western countries. Western migrants have returns very close to those of natives.

Given that firm productivity matters for worker outcomes, it is relevant to study transitions between firm types and to ask who gets access to highproductivity firms. For all starting productivity deciles, natives are more likely to have moved upwards five years later. The immigrant-native difference in mobility is about 5 percentage points up to and including the 7th productivity decile, which is substantial relative to a baseline mobility of 10–15 percent. Differences across subgroups of the foreign-born also suggest that upward mobility is linked to overall socioeconomic positions. First, upward mobility is more common among immigrants with 10 or more years in the host country than among those who arrived more recently. Second, Western migrants transition to more productive firms more often than non-Western migrants. Third, upward mobility is less prevalent when working in peer-dense firms than when one has more native colleagues.

Finally, we present a decomposition analysis to evaluate the contribution of firm productivity pay premiums to the immigrant-native earnings gap. The average premium is the sum of sorting across deciles, and a pay-setting component for working in a given decile (relative to working in the lowest productivity decile). This average is 1 percentage point higher for immigrants, amounting to 8% of the earnings gap in the overall sample. Importantly, however, sorting and pay-setting work in opposite directions. Assuming migrants had the same returns to firm productivity as natives, their over-representation in less productive firms increases the earnings gap by 10%. The sorting component is particularly striking for women (20% of the gap) and those up to the age of 30 (about 30%), but for both of these groups the overall earnings gap is also much smaller (around 4%). On the other hand, if the allocation across firm types would have been the same among immigrant and native workers, the higher returns among immigrants would have reduced the gap by around 18%.

In sum, we find that: (i) more productive firms pay higher wages; (ii) earnings gains from firm productivity are greater for immigrant groups with poor average labor market positions, especially in the lower part of the productivity distribution; (iii) immigrants are more concentrated in low-productivity firms, and have lower rates of upward mobility. Higher returns are consistent with greater monopsony power over immigrants in the lower part of the firm productivity distribution. A greater earnings gap in the lower end means a steeper gradient to the pay offered by more productive firms. Our findings are also consistent with greater mismatch between firms and workers for immigrants. The relative concentration in low-productive firms and lower upward mobility rates fit predictions of ethnically segregated and segmented labor markets (Reich et al. 1973; Massey and Denton 1993).⁶

Since immigrants gain more from entering better firms but do so less frequently, it seems likely that that there are thresholds for immigrants to climb the productivity ladder. There are of course many potential candidates for such thresholds; for example, language barriers or manager hiring practices following ethnic delineations (Åslund et al. 2014). From a policy perspective, it is particularly striking that immigrant groups with poor labor market positions deviate the most from natives in sorting, mobility, and returns. This speaks against voluntary sorting due to worker preferences, and signal the individual and societal gains from more equal employer access.

The rest of the paper proceeds as follows. In Section 2.2 we outline the data and main sample. In Section 2.3 we present descriptive evidence on the native-immigrant earnings gap, focusing on the role of firms in explaining this gap. Section 2.4 explains how we measure firm productivity. Our main results are included in Section 2.5, where we analyze the sorting of immigrants and natives across the firm productivity distribution, as well as the earnings returns associated with working in more productive firms. This also considers mobility up the productivity distribution and presents a decomposition of the earnings premium. Finally, Section 2.6 concludes.

⁶The broader literature on transitions of workers up the job ladder and on worker sorting across firms also finds differences across groups of workers (see Haltiwanger et al. 2018).

2.2 Data and main sample

We use data for the entire Swedish population over the period 1998 to 2017, combining information from several administrative registers collected by Statistics Sweden. The RAMS matched employer–employee database is used to construct an employer-employee panel with yearly information on firm size in November, industry and total earnings paid by the firm. Statistics Sweden's business register on firm-level accounts provides information on value added (VA) for private firms. VA is defined as total value added at each production stage net of costs for intermediate goods and services, and is equal to total revenues minus intermediate consumption of goods and services.⁷ We complement this information with a rich set of socioeconomic characteristics from the Louise/Lisa database.

We restrict our sample to people aged between 18 and 65 who are employed at firms in the private sector that have at least two employees in November. We drop the self-employed. For each employee we compute total annual earnings, job tenure, and total number of months worked at the primary employer.⁸ All monetary values are deflated to 2010 Swedish Kronor (SEK) and winsorized at the 99th percentile of their respective yearly distribution. We drop individual histories if log-earnings in any year are three standard deviations or more above the sample mean. In addition, we drop observations where earnings from the primary employer are lower than the Price Base Amount (PBA) for 2010.⁹ Our main outcome of interest is monthly earnings from the primary employer, obtained by dividing annual earnings by the number of months worked.

Immigrants are defined as foreign-born with two foreign-born parents. People born abroad to at least one Swedish-born parent are excluded from the sample. We also present results where immigrants are divided into "West" (i.e. Western Europe, USA and Australia) and "Rest of the world" based on country of birth.¹⁰

⁷Firm accounts are available until 2015. We eliminate from the starting sample firm observations for which VA information is missing. This results in about 12% employee-year observations being omitted. Table A.2 in the Appendix shows that the sample statistics are virtually unaffected.

⁸The primary employer is defined as the firm paying the highest yearly earnings. To compute job tenure we use data from 1985 onward.

⁹PBA is used to calculate benefits and fees in Sweden. Ruist (2018) argues that an earnings level equal to three times the PBA is a threshold for being self-supporting. Therefore, one PBA is a rather low threshold.

¹⁰"West" consists of the Nordics (Denmark, Finland, Norway and Iceland but not Sweden), Western Europe (Ireland, the UK, Germany, Greece, Italy, Malta, Monaco, Portugal, San Marino, Spain, the Vatican Sate, Andorra, Belgium, France, Liechtenstein, Luxembourg, the Netherlands, Switzerland and Austria), Canada, USA, Australia and New Zealand. "Rest of World" are countries that are not in the West.

2.2.1 Summary statistics

Table 2.1 shows summary statistics for our main analysis sample. Panel A shows worker-level characteristics. At the start of the sample period, 9% are immigrants. Out of these, 51% were born in the West while 49% were born in the Rest of the World. The share of immigrants has doubled over the sample period. This change is driven primarily by an increase in non-Western migrants. 18% of employees work at firms that are either completely native-segregated or completely immigrant-segregated. Considering how this varies by immigrants and natives, 6% of immigrants work at all-immigrant firms, and 20% of natives work at all-native firms.

The sample is more male than the population overall -64% are men - and the average age is 40 years old.¹¹ Over time, the educational level has increased on average, reflected by a decline in the share with only compulsory education and an increase in the share with upper secondary education. Moreover, earnings (in 2010 prices) have increased from 21,000 SEK to 28,000 SEK between 1998 and 2017.

Turning to firm characteristics in Panel B, there are approximately 145,000 firms per year in the sample. The mean share of immigrants per firm has evolved roughly in line with the mean share of immigrants in the population. The mean masks the fact that there is a large portion of completely segregated firms: 62% of firms are native-segregated and 5% of firms are immigrant-segregated. The share of firms that are native-segregated has declined over time while the share that are immigrant-segregated has increased. Regarding firm size, there are on average 22 employees per firm (the median firm size is 6). Firm size has increased somewhat over time, where mean firm size was just under 21 in 1998 and 29 in 2017.

2.3 Immigrant-native earnings ratio

We start by analyzing the immigrant-native earnings gap and how it has evolved over time. Figure 2.1 reports the yearly immigrant-native earnings ratio between 1998 and 2017. It is given by the exponential of the coefficient θ_t from the following yearly regressions:

$$ln(e_{it}) = c + \theta_t imm_i + \beta_t X_{it} + \lambda_{jt} + \varepsilon_{it}$$
(2.1)

 $ln(e_{it})$ are log monthly earnings for worker *i* in year *t* at the primary employer. Included in X_{it} are controls for age, age squared, gender, level of education (dashed line), as well as industry dummies $\lambda_{ind,t}$ (dotted line) or firm fixed effects $\lambda_{f,t}$ (triangles). The figure shows that there is a persistent earnings gap between natives and immigrants in all years. The raw immigrant-native earnings ratio has been declining over time, meaning that the gap in earnings

¹¹The male concentration follows from the fact that we are focusing on the private sector only.

	1998-2017	1998	2017
	(1)	(2)	(3)
Panel A: Employees			
Immigrant	0.13	0.09	0.18
Immigrant from West	0.04	0.05	0.03
Immigrant from Rest of World	0.09	0.05	0.15
Native-segregated firms	0.17	0.22	0.11
Immigrant-segregated firms	0.01	0.00	0.01
Male	0.64	0.66	0.62
Age	40.29	39.93	40.22
$Age \leq 30$	0.27	0.27	0.29
$Age \ge 50$	0.27	0.26	0.28
Education, compulsory	0.16	0.25	0.11
Education, secondary	0.55	0.54	0.53
Education, upper secondary	0.29	0.21	0.35
Education, missing	0.01	0.01	0.01
Monthly earnings (2010 SEK)	24,624	20,838	28,004
No. of employees $ imes$ year	46,870,657	1,883,222	2,504,695
Panel B: Firms			
Fraction immigrants at employer	0.13	0.09	0.19
Yearly employer size	22.38	20.55	29.00
Share native-segregated firms	0.62	0.71	0.49
Share immigrant-segregated firms	0.05	0.03	0.07
No. of firms \times year	2,883,741	129,322	115,210

 Table 2.1. Summary statistics, analysis sample

has widened by approximately six percentage points between 1998 and 2017. Adjusting for age, gender, and education, the gap has widened even more, at approximately seven percentage points. The results with industry dummies indicate that the widening gap is present also within industries. However, adding firm fixed effects, the earnings ratio is higher in all years and even rises slightly over the period. It follows that earnings differences are substantially lower within than between firms. This suggests that the increase in the earnings gap is driven by differences between rather than within firms (see e.g. Tomaskovic-Devey et al. 2020), which motivates focusing on firms as one key element to understanding differences in earnings between immigrants and natives.¹²



Figure 2.1. Immigrant-native earnings ratio

That the place of work is important for earnings is also confirmed when we estimate AKM models separately for immigrants and natives. Full results, including an outline of the AKM model, are included in Appendix Table A.1. The standard deviation of the estimated person effects are nearly twice as big as the standard deviation of the estimated firm effects for both natives and immigrants, indicating that a large share of earnings differences across workers depends on worker characteristics that are rewarded equally across firms. This

¹²In Appendix Figure A.1, we plot the earnings ratio of immigrants compared to natives separately for immigrants from Western countries and from Rest of World countries. The West/native earnings ratio is flat and equal to approximately 0.97 throughout the period, once we include controls. The Rest of World/native earnings ratio is much lower, approximately between 0.8 and 0.85 throughout the period. The overall earnings gap within firms has been closing over time when we compare the Rest of World immigrants to natives. Thus, at least some of the overall trends are driven by differences in the composition of migrants, bearing in mind that the share of Rest of World migrants has increased over time (see Table 2.1).

is in line with the large body of work that finds that skill differences across workers are important in explaining earnings. Even so, the results suggest that firms are also a key component in explaining earnings differences across workers.

Decomposing the overall variance of log earnings, firm effects are found to account for 11% of the overall variance in earnings for natives, and 22% for immigrants (see the bottom of Table A.1). The covariance of person and firm effects accounts for a further 6% for natives and only just over 1% for immigrants, suggesting that at least for immigrants, there is little sorting of high-wage workers to high-wage firms.

For the remainder of the paper, we focus on a measure of firm productivity based on value added per worker, as opposed to the AKM firm fixed effects. This choice allows us to give an explicit interpretation to the relationship between firms and the immigrant-native pay gap, as well as include segregated firms in all results, a feature which is relatively common among immigrants and natives (see Table 2.1).¹³ Nonetheless, as we show in the next section, the firm fixed effects from an AKM model are positively correlated with firm productivity based on value added per worker.

2.4 Firm productivity ranking

2.4.1 Firm productivity and earnings

In the previous section, we showed that there are substantial immigrant-native earnings gaps, particularly between firms. Differences in firm productivity could potentially explain this finding.¹⁴ This section presents our procedure for capturing firm value added, describes how this measure of persistent firm productivity is related to firm size and industry, and analyzes its association with earnings at the firm level.

In order to obtain a measure of persistent firm productivity that abstracts from fluctuations due to the business cycle and productivity shocks, we start by estimating the following model:

$$\ln(VA/N)_{ft} = \lambda_f + \lambda_t + \varepsilon_{ft} \tag{2.2}$$

where the fixed effects λ_f capture the permanent component in firm-level productivity and λ_t account for yearly effects common across all firms.

We estimate equation (2.2) using all firms for which information on both value added and firm size is available in at least two years in the matched

¹³We also avoid using firm fixed effects due to known possible incidental parameter problems when estimating the AKM (Andrews et al. 2008). Recent papers have proposed ways to tackle these problems (Kline et al. 2019; Bonhomme et al. 2020), but in our context, due to the smaller size of the group of immigrants, the bias may be particularly severe, further complicating how to interpret the results.

¹⁴For an extensive overview of why productivity differs between firms, see Syverson (2011).

employer-employee panel during 1998–2015, after restricting firm size to be larger than one.¹⁵ Next, we use the empirical distribution of the $\hat{\lambda}_f$ firm effects to rank firms into deciles or percentiles. Similar to the clustering procedure used by Bonhomme et al. (2019), grouping firms in this way aims to improve the tractability of our analyses. By construction, each firm's position in the productivity distribution is fixed over time. Even though value added per worker is only available until 2015, we thus obtain a measure of persistent productivity until 2017 (as long as the firm exists in earlier years).

To understand what types of firms are found in each productivity decile, Figure 2.2 shows the distribution of employees by firm size (Panel a) and by industry (Panel b) within each productivity decile of the $\hat{\lambda}_f$ distribution.



(a) Firm size (b) Industry Figure 2.2. Employment distribution within decile rank, by firm size or industry (1998–2017)

We group firms into five size bands: up to 9 employees (micro), up to 50 employees (small), up to 250 employees (medium), below 1000 employees (large) and 1000 employees and above (very large). Panel (a) shows that small firms tend to be the most productive ones, whereas the first productivity decile primarily consists of the largest firms.¹⁶ This confirms the importance of controlling for firm size in the analysis. Panel (b) additionally shows that virtually all industries are found in each productivity decile. Thus, working in more productive firms does not only reflect working in specific industries. Instead, the whole range of firm productivity types tends to be represented in the different industries.

Figure 2.3, Panel (a), shows log earnings (residualized by firm size band) averaged by each percentile of the firm productivity distribution. To understand how the common component of firm earnings relates to firm productivity above and beyond individual worker heterogeneity, in Panel (b) we plot the firm fixed effects obtained from a pooled AKM regression against productiv-

¹⁵Dropping the firm size restriction leaves the results qualitatively unaffected.

¹⁶This stands in contrast to the commonly proposed theoretical prediction that firm size and productivity are positively related, see for example Card et al. (2018).

ity percentiles. The results in Panel (a) indicate that, from the 20th percentile onward, there is a positive relationship between (residualized) earnings and productivity ranking.¹⁷ The common firm-specific pay premium captured by firm fixed effects in the AKM model relates positively to the firm productivity ranking over the distribution of firm productivity percentiles.



Figure 2.3. More productive firms pay higher earnings

Panel (b) of Figure 2.3 does not show a "hockey-stick" pattern as found by Card et al. (2016); that is, we do not find that firms in the bottom percentiles have similar earnings premiums. This arises because a large number of employees are clustered in a few unproductive firms at the bottom of the distribution. Consistently, when we instead rank firms by weighting by the number of employees, the relationship between firm FE and productivity ranking is flat in the bottom part of the distribution and positively sloped afterwards (Figure B.2 in the Appendix). Moreover, when we plot AKM firm fixed effects directly against mean log valued added per worker, we also find the characteristic "hockey-stick" patterns (Figure B.1). In the Appendix, we show that our main results in Section 2.5 below are robust to using the alternative employee-weighted ranking.

2.4.2 Robustness of the firm ranking

Table 2.2 compares the firm productivity ranking used in the main analysis with alternative ranking procedures. Panel A shows rank correlation coefficients for alternative specifications. Panel B reports the share of firms classi-

¹⁷Higher mean earnings in firms in the very lowest deciles could arise for several reasons. First, controlling for five firm size bands might insufficiently control for the firm-size wage premium. Second, the linear relationship in Panel (b) of Figure 2.3 indicates the presence of "high-wage workers" in low-productive firms. A third possibility is that large firms (often found in lower deciles) may in general offer more full-time contracts, implying higher earnings due to more hours worked. The latter, however, is not supported by the data: average or median percentage of full-time contracts doesn't vary by firm size or value added per worker (graphs not included).

fied higher or lower in the ranking by at least 10 percentiles as compared to the baseline. Reassuringly, we find that the baseline ranking is robust to including additional controls or using different methods to generate the ranking.

Columns (1) to (3) of Table 2.2 show results when either producing the ranking by industry or controlling for the share of immigrants, or both. In general, this leaves the ranking qualitatively unaffected. For the ranking done by industry, despite some 18.5% firms being classified as belonging to a lower decile of the firm productivity distribution, the correlation with the baseline ranking appears to be substantial (0.93).

Log value added per worker may to some extent mechanically reflect that high-skilled workers are concentrated in certain firms; i.e. firm productivity may be a function of worker productivity. Column (4) reports results when we re-estimate equation (2.2) by including staff-composition characteristics averaged at the firm-year level (share of men, share of workers in each education category, average tenure at the firm, share of immigrants).¹⁸ Also in this case the productivity ranking is virtually unaffected.

In Column (5) we rank firms using AKM fixed effects from an earnings regression. Consistent with the correlation between the AKM firm fixed effects and the firm productivity ranking shown in Figure 2.3, the two alternative ranking measures are positively correlated (correlation of 0.44). Recall that the AKM firm fixed effects capture firm-level premiums that might be attributable to several other time-fixed components on top of persistent productivity. In this sense, the observed degree of correlation between the AKM-and the productivity-based rankings appears to be sizable.¹⁹

The main analysis uses a firm-weighted ranking. An alternative is to use an employee-weighted measure of firm productivity. Since many large firms have low per-worker value added, the first few deciles of the firm-weighted ranking are extended into several deciles of the employee-weighted ranking. Our main results included in the next section are consistent with this pattern.

Finally, we have additionally tested the robustness of the ranking in various ways (detailed results are available upon request). Recall that in the baseline results, firm size is computed using the matched employer-employee panel. If we instead use a measure of firm size from the business statistics register, results are similar. This is also the case if we remove employees whose earnings are below the yearly price base amount from the analysis.

¹⁸Note that in order to alleviate some of these concerns, throughout our main analysis on the returns to working in more productive firms, we condition on individual-level fixed effects.

¹⁹Results are unaffected if we correlate the AKM ranking with an employee-weighted ranking.

	Industry	Share of immi- grants	Industry and share of immi- grants	Staff composi- tion	AKM FE ranking
	(1)	(2)	(3)	(4)	(5)
Panel A: Correlat	ion with base	line ranking			
	0.9294	0.9981	0.9276	0.9825	0.4430
Panel B: Share of	firms moving	in the rankin	g		
moving down	0.1856	0.0001	0.1844	0.0229	0.3369
moving up	0.0730	0.0062	0.0851	0.0345	0.2980
No. of firms	370,692	370,692	370,692	355,676	339,010

Table 2.2. Robustness of firm ranking

Notes: Panel A reports Spearman's rank correlations between the baseline productivity ranking percentiles and the following alternative measures: Column (1): ranking firms by industry; Column (2): controlling for the yearly share of immigrants at the firm; Column (3): ranking firms by industry and controlling for the share of immigrants at the firm; Column (4): controlling for education categories, gender, age, tenure, share of immigrants averaged at the firm-year level; Column (5): ranking by using AKM firm fixed effects from an earnings regression. Panel B reports the share of firms moving at least 10 percentiles in the ranking as compared to the baseline.

2.5 Sorting and returns to working in productive firms

The relationship between productivity and earnings for immigrants compared to natives can operate through several channels. First, it could be related to sorting: immigrants and natives may work in different types of firms. Second, it could be related to pay setting: in firms of a given productivity, immigrants and natives may be offered or negotiate different earnings (Dostie et al. 2020; Card et al. 2016). This can for example arise because firms have market power over workers and consequently firms are able to set lower wages to groups of individuals with more inelastic labor supply to the firm. Motivated by this, we consider sorting and earnings returns for immigrants and natives in this section.

2.5.1 Distribution of workers in the firm productivity distribution

Figure 2.4a presents the distribution of immigrants and natives across the firm productivity distribution. It shows that a large part of workers are employed in relatively low productivity firms. Figure 2.4b, which omits the first two percentiles, additionally shows that the distribution of immigrants is skewed to the left: immigrants are relatively more concentrated in low productivity firms, while natives are more concentrated in high productivity firms.²⁰





Table 2.1 showed that, over the period we are studying, there has been an increase in the fraction of immigrants. This may have affected sorting over time. To investigate this, we break the data into the sub-periods 1998–2009 and 2010–2017. We then estimate the following linear probability model separately by sub-period p:

$$imm_i = \alpha_p + \sum_{d=2}^{10} \beta_{dp} decile_d + \varepsilon_{ip}$$
(2.3)

²⁰We find very similar patterns when we restrict the sample of migrants to Rest of World immigrants with at least 10 years spent in Sweden who hold upper secondary education.



Figure 2.5. Sorting of immigrants across productivity deciles

where $decile_d$ refers to productivity decile. The first decile is omitted from the regressions such that the immigrant shares in a particular decile are estimated relative to the bottom decile.

Figure 2.5 plots $\hat{\beta}_{dp}$ from the regressions. It shows that immigrants have become relatively more concentrated in firms with lower productivity over time, compared to the share in the first decile. In particular, the second period estimates (shown by the dark blue squares) are above the first period (orange dots) at the bottom deciles and vice versa at the top.

One reason for the change in sorting could be due to compositional changes over the long time period considered. To account for this, we weight the second sub-period to match the first in terms of either country of birth (CoB) or years since migration (YSM) to Sweden. Weights are constructed as the ratio of the share in each respective country of birth or year since migration cell. The results (shown by the diamonds) suggest that the change in sorting is observed even when we re-weight by years since migration, though the change at the bottom deciles is somewhat attenuated. Re-weighting by country of birth (shown by the triangles), on the other hand, diminishes the decline at the top but the concentration at the bottom remains. Changes to the composition of workers with regards to their country of birth can thus explain the relative decline at the top but not the relative increase at the bottom.

In order to assess whether sorting differs for the two main groups of immigrants (West and Rest of the World), we estimate equation (2.3) separately for these groups. The first message of Figures 2.6a and 2.6b is that sorting is much stronger among non-Western immigrants (note that the scales are different). Regarding changes over time, Figure 2.6a suggests that the pattern for the Rest of the World group is similar to what we saw above: there has been a relative increase of immigrants in lower productivity firms and a relative decrease in higher productivity firms. When interpreting the results, it is important to bear in mind that the share of this group has increased in the first decile, from 8.6% to 13.8%. For Western immigrants in Figure 2.6b, the share of migrants has decreased in the first decile, from 4.9% to 4.1%. The levels suggest that there has been an aggregate increase in the share of migrants within each decile over time, both at the bottom and at the top of the productivity distribution, in line with the fact that immigration has increased over the period in general.



(a) Rest of World (b) West Figure 2.6. Sorting of immigrants across productivity deciles – by country of birth

2.5.2 Earnings returns to working in more productive firms

Immigrants are relatively more concentrated in low-productivity firms than natives, and thus make up a larger share of employees in low-productive firms than in high-productive firms. Figure 2.7 shows that in the first five deciles, immigrants account for 14 to 16% of employment within each decile. At the top deciles, immigrants account for just over 8%.²¹ Not only is there a higher share of immigrants in low-productivity firms, but the persistent raw earnings gap is particularly prevalent in lower productivity firms. Moreover, earnings of both natives and immigrants are higher in high-productivity firms, but there are fewer immigrants than natives as we go up the firm productivity distribution.

To more formally understand how earnings returns differ for immigrants and natives across the firm productivity distribution, we regress log monthly earnings at the primary employer $\ln(e_{it})$ on the firm productivity decile *deciled*

²¹Figure A.2 shows that this pattern is predominantly driven by immigrants from the Rest of the world. Immigrants from Western countries account for 4% of employees within each decile over the sample period.



Figure 2.7. Share immigrants and raw log earnings by productivity decile

and the interaction of the decile and an immigrant dummy *imm_i*:

$$\ln(e_{it}) = c + \sum_{d=2}^{10} \theta_d decile_d + \sum_{d=2}^{10} \gamma_d decile_d \times imm_i + \alpha_i + \lambda_t + \beta X_{it} + \varepsilon_{it} \quad (2.4)$$

The estimand $\hat{\theta}_d$ ($\hat{\theta}_d + \hat{\gamma}_d$) are the earnings returns to natives (immigrants) of working in relatively more productive firms, compared to natives (immigrants) working in the first productivity decile. Thus, $\hat{\gamma}_d$ is the differential return to immigrants of working in more productive firms. We include individual fixed effects α_i to control for individual heterogeneity in earnings. The identification of the return by productivity deciles.²² We include controls X_{it} for age polynomials (age squared and age cubed) and firm size band (micro, small, medium, large, very large), as well as the same controls interacted with the immigrant dummy to allow differential effects for immigrants and natives. We cluster standard errors at the firm level.

The results are included in Figure 2.8 below. Panel (a) is for the full sample of immigrants, while Panel (b) splits the comparison to natives into the Rest of the World and the West birth region groups. The blue dots show returns to natives $(\hat{\theta}_d)$ while the orange dots show returns to immigrants $(\hat{\theta}_d + \hat{\gamma}_d)$.

²²Specifications excluding worker fixed effects give similar results, but the estimated coefficients $\hat{\theta}_d$ and $\hat{\theta}_d + \hat{\gamma}_d$ are much larger in size.



(a) All immigrants



(b) By immigrant group

Figure 2.8. Earnings returns to working in more productive firms

Note: Panel (a) plots $\hat{\theta}_d$ and $\hat{\theta}_d + \hat{\gamma}_d$ from equation (2.4) for the full sample of natives and immigrants. Panel (b) plots $\hat{\theta}_d$ and $\hat{\theta}_d + \hat{\gamma}_d$ from equation (2.4) for: (i) the full sample (circles); (ii) the full sample of natives and Western immigrants (diamonds); and (iii) the full sample of natives and Rest of World immigrants (triangles). All specifications include individual fixed effects, year fixed effects and controls as specified in Section 2.5.2.

For both immigrants and natives, there is a clear positive earnings return to working in more productive firms. For example, for the full sample in Panel (a), the estimated return to natives of working in the fifth decile compared to the first is 6.7 log points, and to immigrants nearly 12 log points. The

return in the tenth decile is 20 log points for natives and 25 log points for immigrants. The returns to immigrants of working in more productive firms is generally higher than for natives. This differential return increases gradually in the bottom half of the productivity distribution, after which it remains flat at just over 5 log points. This pattern is clearest for immigrants from the Rest of the World. Immigrants from the West have earnings returns more similar to natives.²³ Recall that there is a persistent raw earnings gap between immigrants and natives across the whole productivity distribution (Figure 2.7). The results thus indicate that for immigrants who climb to more productive firms, each within-individual rung on the ladder is taller, not that immigrants have higher earnings than natives in more productive firms.

The returns to firm productivity may differ across groups of migrants. Above, we already considered how the returns differ by country of birth. Another important component is years since migration. As discussed in the introduction, a large literature suggests that immigrants become more similar to natives with years spent in the host country. Motivated by this, we re-run specification (2.4) above separately for immigrants that have spent less than 10 years in Sweden and for those who have spent at least 10 years in Sweden. The estimates from this exercise are similar to those presented above in the sense that both immigrants that have spent less than 10 years in Sweden have higher returns than natives, see Figure B.4 in the appendix. However, immigrants with less than 10 years in Sweden experience even higher returns than those who have been in the country longer. This pattern is consistent with that immigrants become more similar to natives with time spent in the country.

Within-firm variation in productivity and earnings

Our definition of productivity is based on log value added per worker, where one firm is assigned a constant productivity rank throughout the sample period. The estimates above are based on comparisons across firms. Next, we consider how within-firm variation in productivity is related to earnings.

We estimate a model allowing for different rent-sharing between immigrants and natives. Motivated by the finding that returns differ for immigrants from different countries, we let the association between earnings and firm value added vary between three groups: Rest of the World, Western, and native (reference). The analysis allows us to gauge how changes to value added at the firm over time translate into changes in earnings, and whether this differs for immigrants and natives. This keeps the sorting of workers to firms constant and controls for time-invariant worker and firm heterogeneity.

²³Results from estimating equation (2.4) using the employee-weighted ranking are included in Appendix Figure B.3.

The identifying variation comes from workers who do not switch firms (i.e. within an employment spell):

$$\ln(e_{ift}) = c + \delta_1 \ln(VA/N)_{ft} + \delta_2 \ln(VA/N)_{ft} \times RoW_i + \delta_3 \ln(VA/N)_{ft} \times Western_i + \lambda_{if} + \lambda_t + \varepsilon_{ift}$$
(2.5)

where $\ln(e_{ift})$ are log earnings for worker *i* in year *t* at main firm f, $\ln(VA/N)_{ft}$ is a time-varying measure of log value added per employee at the firm-level and λ_{if} are match fixed effects.

Table 2.3 reports $\hat{\delta}_1$, $\hat{\delta}_2$ and $\hat{\delta}_3$. The column (1) specification does not include any additional controls, while column (2) includes individual time-varying controls (age squared, age cubed, and tenure)²⁴ and in column (3) we additionally add time-varying firm controls (firm size and share of immigrants at the firm). As before, individual and firm controls are also interacted with the immigrant group to allow the coefficients to vary for natives and immigrants.

The results indicate that firms indeed share profits with their employees: a 1% increase in value added per worker is associated with approximately a 0.03% increase in earnings for natives. In line with the results above, the estimation indicates that there are positive differential returns for immigrants: immigrants from non-Western countries earn an additional 0.01% for a 1% increase in value added, while Western immigrants earn approximately half the additional return. The standard deviation in firm log value added per worker is about 1 and the within-firm standard deviation is $0.5.^{25}$ In other words, if a firm moves from low to high productivity (i.e. increases its log value added by two within-firm standard deviations), earnings are expected to increase by over 4 percent among non-Western migrants. If one is willing to extrapolate to across-firm variations, the earnings increase would be twice as large.

We also estimate equation (2.5) for firms in different productivity deciles of the persistent productivity ranking outlined in Section 2.4. In Table 2.4, column (1) includes workers at firms in the bottom half of the deciles while column (2) includes the top half. Column (3) only includes the bottom decile and column (4) only includes the top decile. The results reveal an interesting pattern. First, the differential rent-sharing for immigrants compared to natives is present in the bottom half of the firm decile distribution, but not in the top. This is consistent with Figure 2.8, which suggests that the difference in gains mainly materializes in the bottom half of the distribution, while the slopes are relatively similar in the upper half. Second, the results suggest that low-productivity firms do not share rents, consistent with the fact that low value-added firms have limited rents to share.

²⁴To compute tenure we use data back to 1985. Because we have observed workers in 1998 for fewer years than workers in 2015, the tenure variable is left-truncated. We include tenure in six bands: 1 year (omitted category), 2-3 years, 4-6 years, 7-9 years, 10-13 years and 14+ years.

²⁵The overall and between standard deviations in the firm-year sample is 0.94. The within-firm standard deviation is 0.51.

	(1)	(2)	(3)
Log VA per worker	0.032***	0.026***	0.036***
	(0.003)	(0.003)	(0.002)
Rest of World \times Log VA per worker	0.012***	0.008*	0.010***
	(0.004)	(0.004)	(0.003)
Western \times Log VA per worker	0.006***	0.004**	0.004**
	(0.002)	(0.002)	(0.002)
R^2	0.765	0.776	0.776
Ν	37,198,227	37,198,227	37,198,227
Year FE	Yes	Yes	Yes
Spell FE	Yes	Yes	Yes
Individual controls		Yes	Yes
Firm controls			Yes

 Table 2.3. Rent-sharing among immigrants and natives

Note: This table provides the results of estimating equation (2.5). Individual controls are age squared, age cubed, and tenure. Firm controls are firm size band (micro, small, medium, large, very large) and share of immigrants at the firm. Controls are also interacted with the immigrant group. Standard errors are clustered by firm and reported in parentheses.

Table 2.4.	Rent-sharing	among	immigrants	and	natives	in	different	productivity
deciles								

	(1)	(2)	(3)	(4)
Firm productivity decile	1-5	6-10	1	10
Log VA per worker	0.025***	0.055***	-0.000	0.042***
	(0.003)	(0.001)	(0.005)	(0.002)
Rest of World \times Log VA per worker	0.018***	0.001	0.005	-0.000
	(0.004)	(0.002)	(0.006)	(0.005)
Western \times Log VA per worker	0.006***	0.003	0.004	-0.001
	(0.002)	(0.002)	(0.003)	(0.004)
R^2	0.781	0.763	0.774	0.791
Ν	24,542,19	912,656,02	814,453,56	61,965,021
Year FE	Yes	Yes	Yes	Yes
Spell FE	Yes	Yes	Yes	Yes
Individual controls	Yes	Yes	Yes	Yes
Firm controls	Yes	Yes	Yes	Yes

Note: This presents the results of estimating equation (2.5), separately for firms in different firm productivity deciles. Individual controls are age squared, age cubed, and tenure. Firm controls are firm size band (micro, small, medium, large, very large) and share of immigrants at the firm. Controls are also interacted with the immigrant group. Standard errors are clustered by firm and reported in parentheses.

2.5.3 Upward mobility in the productivity distribution

The returns to working in more productive firms are higher for immigrants than natives, but immigrants are less likely to work in more productive firms. These two findings raise the question as to which immigrants actually climb up the productivity ladder.

Figure 2.9 plots the share of immigrants in various groups and natives who move up the productivity ranking, conditional on where they start when they enter our sample.²⁶ We define upward mobility as working in a higher productivity decile five years later compared to when the individual is first observed.²⁷ We see that, for all groups, the likelihood of moving up decreases with initial productivity decile. However, immigrants are less likely than natives to move at all, and this holds across the whole productivity distribution and for all groups of immigrants.²⁸



Figure 2.9. Upward mobility, natives vs. various groups of immigrants

²⁶We look at the following groups: Western immigrants, Rest of World immigrants, recentlyarrived immigrants (YSM less than 10 years), immigrants with YSM more than 10 years, immigrants who start in peer-dense firms, defined as firms with above-median share of immigrants and immigrants who start in non-peer dense firms, defined as firms with below-median share of immigrants. Note that the immigrant groups are not mutually exclusive. For example, there is likely overlap between the group of Rest of World immigrants and the group of recent arrivals, or between the group of recent arrivals and those in peer-dense firms.

²⁷The outcome variable takes the value 1 if the productivity decile five years later is *strictly* higher than in the initial year, and 0 otherwise. Since by construction the outcome does not vary for those that start off in the highest decile, we disregard these individuals.

²⁸Figure A.3 in the appendix shows these patterns for natives and the pooled sample of immigrants.

The analysis shows that upward mobility is more common among those that start in non-peer dense firms, whereas migrants starting their career in immigrant and peer dense firms largely remain in low-productive workplaces. This is consistent with the influence of ethnic segregation and segmentation. Mobility into high-productivity firms is also strongly correlated with region of birth. It is higher for Western than Rest of World immigrants, at least when transitioning between higher productivity deciles. It is also more common for immigrants with 10 or more years in the host country than among those who arrived more recently.

Could it be that mobility differences are related to differential selection of natives and immigrants? In order to test this hypothesis, we assess the native-immigrant difference in average person fixed effects across the productivity distribution.²⁹ Figure 2.10 plots this difference for four different groups, by productivity decile: i) all workers; ii) never-movers, defined as workers that never move across productivity distribution deciles throughout their histories; iii) movers, defined as those that move at least once (either up or down) during their history; and iv) upward movers, defined as above as those that move up with respect to the baseline year five years later.

Among the upward movers, the native-immigrant difference in average person fixed effects is the smallest and close to zero from the seventh decile onward. For all the other groups, natives have higher average person fixed effects across the productivity distribution, although generally diminishing as we move up the distribution. The difference is particularly striking for the never-movers. The results suggest that, in terms of their individual-specific earnings capacity, immigrant movers are similar to native movers, while immigrant stayers differ from native stayers. Somewhat speculative, the large gap among never-movers could be seen as an indication that natives stay because they are in a good position, whereas migrants have fewer options and therefore have persistently lower earnings.

²⁹Gerard et al. (2018) call this difference the "average skill gap". The person fixed effects are taken from estimating equation (2.4).



Figure 2.10. Native-immigrant average difference in person fixed effects

2.5.4 Decomposition of the immigrant-native earnings gap

Finally, we tie together the analysis by decomposing the earnings premium associated with working in more productive firms into a sorting and a paysetting component.³⁰ Building on our framework from section 2.5.2, assume that the earnings of worker *i* in group *g* in time *t* are given by:

$$\ln e_{git} = \alpha_{gi} + \theta_d^g + X_{git}\beta^g + \varepsilon_{git}$$
(2.6)

where α_{gi} is a person effect, θ_d^g is a group-specific earnings premium in productivity decile d, X_{git} is a vector of time-varying controls (age squared, age cubed, firm size and year effects) and β_g a vector of coefficients. ε_{git} captures all remaining determinants of earnings. Let D_{git} indicate whether an individual *i* in group *g* is employed in time *t*. Let \bar{X}_{It} and \bar{X}_{Nt} constitute the means of the observed covariates for employed immigrants (*I*) and natives (*N*) in year *t*, and let π_{It} and π_{Nt} denote the fractions of the two groups employed in decile *d* in year *t*. Assuming $E[\varepsilon_{git}|D_{git} = 1] = 0$, we can express mean immigrant and native earnings in the following way:

$$E[\ln e_{Iit} | D_{Iit} = 1] = E[\alpha_{Ii} | D_{Iit} = 1] + E[X_{It} | D_{Iit} = 1]\beta_I + E[\theta_d^I | D_{Iit} = 1]$$
$$E[\ln e_{Nit} | D_{Nit} = 1] = E[\alpha_{Ni} | D_{Nit} = 1] + E[X_{Nt} | D_{Nit} = 1]\beta_N + E[\theta_d^N | D_{Nit} = 1]$$

³⁰The decomposition closely follows Dostie et al. (2020), who instead decompose a firm-specific earnings premium that differs for immigrants and natives.

and the mean immigrant-native gap in year *t* is then:

$$E[\ln e_{Nit}|D_{Nit} = 1] - E[\ln e_{Iit}|D_{Iit} = 1] = E[\alpha_{Ni}|D_{Nit} = 1] - E[\alpha_{Ii}|D_{Iit} = 1] + E[X_{Nt}|D_{Nit} = 1]\beta_N - E[X_{It}|D_{Iit} = 1]\beta_I + E[\theta_d^N|D_{Nit} = 1] - E[\theta_d^I|D_{Iit} = 1]$$

Since we are interested in the part of the earnings gap explained by the productivity decile premiums, we focus on the third term. A simple decomposition (Oaxaca 1973; Blinder 1973) of the third term gives:

$$E[\theta_{d}^{N}|D_{Nit} = 1] - E[\theta_{d}^{I}|D_{Iit} = 1] = E[\theta_{d}^{N}|D_{Nit} = 1] - E[\theta_{d}^{N}|D_{Iit} = 1] + E[\theta_{d}^{N} - \theta_{d}^{I}|D_{Iit} = 1] = E[\theta_{d}^{I}|D_{Nit} = 1] - E[\theta_{d}^{I}|D_{Iit} = 1] + E[\theta_{d}^{N} - \theta_{d}^{I}|D_{Nit} = 1]$$

The sample counterpart of the expression above is given by:

$$\sum_{d} \widehat{\theta_{d}^{N}} \pi_{Ndt} - \sum_{d} \widehat{\theta_{d}^{I}} \pi_{Idt} = \underbrace{\sum_{d} \widehat{\theta_{d}^{N}} (\pi_{Ndt} - \pi_{Idt})}_{sorting} + \underbrace{\sum_{d} (\widehat{\theta_{d}^{N}} - \widehat{\theta_{d}^{I}}) \pi_{Idt}}_{pay-setting}$$
(2.7)
$$= \underbrace{\sum_{d} \widehat{\theta_{d}^{I}} (\pi_{Ndt} - \pi_{Idt})}_{sorting} + \underbrace{\sum_{d} (\widehat{\theta_{d}^{N}} - \widehat{\theta_{d}^{I}}) \pi_{Ndt}}_{pay-setting}$$

where the $\widehat{\theta_d^g}$ terms, $g \in \{N, I\}$ are estimated from equation (2.6).

We find the first expression to be the most intuitive. The contribution of the productivity decile premiums to the immigrant-native earnings gap is given by a weighted average of the differences in employment shares of immigrants and natives (weighted by the earnings premium of natives per decile) and a weighted average of the differences in decile earnings premiums (weighted by the share of immigrants per decile). The first component in the expression in equation (2.7) (sorting) shows the effect of differences in sorting across the productivity distribution, assuming immigrants were paid the same premiums as natives. It will be positive if natives are more likely to work in more productive firms which pay higher premiums. The second component (pay-setting) shows the differential pay-setting across the productivity distribution, given the distribution of immigrants across productivity deciles. Our results from the previous section consistently show higher earnings premiums for immigrants across the decile distribution, hence we expect this term to be negative.

In the estimations we omit the first decile such that decile premiums are estimated relative to this decile. A sufficient condition to perform the decomposition is that firms in the first decile pay zero earnings premiums to both immigrants and natives. Column (3) of Table 2.4 suggests that firms in the bottom decile of the productivity distribution do not share rents with any of their workers, which strengthens the validity of this assumption. The paysetting component is sensitive to this assumption, while the sorting component is invariant to it. If it does not hold – and in particular if natives earn a higher premium relative to immigrants in the first decile – then the estimated negative pay-setting effect will move toward zero or even become positive.

Table 2.5 shows the decomposition results for the overall group of natives and immigrants and for sub-samples categorized by gender, age, region of origin and education level. Column (1) shows the mean log earnings gap between immigrants and natives in different groups. Columns (2) and (3) show the mean decile premium received by natives and immigrants, respectively. Column (4) gives the difference between column (2) and column (3), while columns (5) and (6) show the sorting and within-decile pay-setting effects, respectively.³¹ Starting with the first row, we see that, on average, the decile premium immigrants get is slightly higher than the decile premium natives get (6.5 vs. 5.5 percent), which is in line with our results from Figure 2.8. The difference of 1 percentage point reduces the overall gap by 8%. As expected, the sorting effect in column (5) is positive and accounts for around 10% of the earnings gap. The pay-setting term, instead, reduces the gap by around 18%.³²

In the next rows, we show how the decomposition results vary for different subgroups. Similar patterns emerge: the pay-setting effect is always negative and larger than the sorting effect. We highlight a few noteworthy patterns. The earnings gap is much larger for men than for women, with sorting explaining a higher proportion of the overall gap for women than for men (20% vs. 9%). In terms of age groups, the biggest earnings gap is observed for the group aged between 31 and 50, and it is slightly lower for the group aged 50 and above, in line with earnings assimilation with time spent in the country. Western immigrants in our sample have an earnings advantage over natives, primarily explained by the pay-setting effect. The mean decile premium for Rest of World immigrants is lower than for Western immigrants due to their concentration in the bottom deciles of the firm productivity distribution. Finally, we note that since we do not fit separate models for these different subgroups, these results do not account for the fact that firms may set different earnings premiums for the different subgroups. As we saw above, premiums may differ for immigrants depending on their country of birth. This aspect is also particularly important in the case of men and women, as previous research finds that firm pay differentials explains an important part of the gender gap (Card et al. 2016; Bruns 2019).

³¹Table A.3 repeats the decomposition exercise when doing the employee-weighted ranking of firms. The results are consistent with the fact that when using the employee-weighted ranking the bottom half of the productivity distribution is now characterized by low decile premiums.

³²The signs on these effects are in line with those in Dostie et al. (2020), who decompose *firm-specific* as opposed to decile-specific premiums using a similar method; the magnitudes are not directly comparable.

	, v	-				
	Earnings	Mean decile	Mean decile	Premium	Sorting	Pay- setting
	gap	premium natives	premium immigran	gap its		
	(1)	(2)	(3)	(4)	(5)	(6)
All	0.124	0.055	0.065	-0.010	0.013	-0.023
By gender						
Male	0.158	0.061	0.071	-0.010	0.014	-0.025
Female	0.044	0.045	0.055	-0.010	0.009	-0.020
By age group						
Up to age 30	0.041	0.051	0.061	-0.010	0.012	-0.022
Between 31 and 50	0.198	0.058	0.067	-0.009	0.014	-0.023
50 and above	0.134	0.056	0.064	-0.009	0.013	-0.022
By region of origin						
West	-0.040	0.055	0.075	-0.020	0.005	-0.025
Rest of World	0.192	0.055	0.061	-0.006	0.016	-0.022
By education						
Compulsory	0.132	0.058	0.060	-0.002	0.020	-0.022
Upper secondary	0.177	0.052	0.065	-0.013	0.009	-0.022
Post Secondary	0.099	0.056	0.066	-0.009	0.014	-0.023

 Table 2.5. Decomposition of immigrant-native earnings gap

Notes: Column (1) shows the mean log earnings gap between immigrants and natives in different groups. Columns (2) and (3) show the mean decile premium received by natives and immigrants, respectively. Column (4) gives the difference between column (2) and column (3). We decompose the gap in column (4) into a between-decile sorting effect (column (5)) and a differential within-decile pay-setting effect (column (6)). The sorting effect is weighted by the native premium, and the pay-setting effect by the immigrant shares.

2.6 Conclusion

The role of firms in determining immigrant-native earnings differentials is potentially important but under-explored. When firms differ in their wage-setting practices, the average earnings of immigrants relative to that of natives may depend both on the firms in which immigrants and natives work, and on how immigrants fare in a given firm.

This paper focuses on firm productivity as a potential key to understanding labor market differences across groups of workers. We use population-wide linked employer-employee data for Sweden between 1998 and 2017 to study the sorting and earnings of immigrants and natives across the firm productivity distribution. Firm productivity is measured using persistent value added per worker, which we rank into bins. We find no indication that this measure is influenced by the composition of workers in the firm, including the share of immigrants that the firm employs.

We reach four main findings. First, immigrants are more concentrated in low-productivity firms relative to natives and have lower rates of upward mobility in the firm productivity distribution. This is in line with previous research that finds that immigrants do not have access to the same workplaces as natives. Second, more productive firms pay higher wages to both immigrants and natives. Third, the earnings gains from working in more productive firms are greater for immigrants than natives. That is, the within-individual increase in earnings for a worker who climbs the productivity ladder is steeper for immigrants than for natives, especially in the lower half of the productivity distribution. This result is particularly striking for immigrants with poor average labor market outcomes: immigrants who are born in non-Western countries and also immigrants who arrived in Sweden more recently. Fourth, decomposing the contribution of firm productivity pay premiums on the immigrant-native earnings gap, we find that the premiums reduce the earnings gap by 8%. Sorting and pay-setting work in opposite directions: the over-representation of immigrants in less productive firms widens the gap by 10%, while the relatively higher premiums that immigrants earn reduces the gap by 18%.

Immigrant labor market assimilation is a major policy concern in many countries, and the academic literature is voluminous. However, so far, the number of studies analyzing the contribution of firms' hiring patterns and paysetting policies to immigrant-native differentials is small. Our results clearly suggest that a better understanding of firm-level factors is needed. Productivity, technology, and competition, as well as their interactions with structural change and institutions, appear to be relevant areas for further research.

References

- Abowd, J. M., F. Kramarz, and D. N. Margolis (1999). "High Wage Workers and High Wage Firms". *Econometrica* 67.2, pp. 251–333.
- Andrews, M. J., L. Gill, T. Schank, and R. Upward (2008). "High wage workers and low wage firms: negative assortative matching or limited mobility bias?" *Journal of the Royal Statistical Society: Series A (Statistics in Society)* 171.3, pp. 673–697.
- Ansala, L., O. Åslund, and M. Sarvimäki (2020). "Immigration History, Entry Jobs, and the Labor Market Integration of Immigrants". *Journal of Economic Geography* forthcoming.
- Arellano-Bover, J. and S. San (2020). "The Role of Firms in the Assimilation of Immigrants". Unpublished.
- Åslund, O., L. Hensvik, and O. Nordström Skans (2014). "Seeking Similarity: How Immigrants and Natives Manage in the Labor Market". *Journal of Labor Economics* 32.3, pp. 405–441.
- Åslund, O. and O. Nordström Skans (2010). "Will I See You at Work? Ethnic Workplace Segregation in Sweden, 1985–2002". *ILR Review* 63.3, pp. 471– 493.
- Barth, E., B. Bratsberg, and O. Raaum (2012). "Immigrant Wage Profiles Within and Between Establishments". *Labour Economics* 19.4, pp. 541–556.
- Barth, E., A. Bryson, J.C. Davis, and R. Freeman (2016). "It's Where You Work: Increases in the Dispersion of Earnings across Establishments and Individuals in the United States". *Journal of Labor Economics* 34.S2, pp. 67– 97.
- Bassier, I., A. Dube, and S. Naidu (2020). *Monopsony in Movers: The Elasticity of Labor Supply to Firm Wage Policies*. Tech. rep. 27755. National Bureau of Economic Research.
- Blinder, A. S. (1973). "Wage Discrimination: Reduced Form and Structural Estimates". *The Journal of Human Resources* 8.4, pp. 436–455.
- Bonhomme, S., K. Holzheu, T. Lamadon, E. Manresa, M. Mogstad, and B. Setzler (2020). *How Much Should we Trust Estimates of Firm Effects and Worker Sorting?* Tech. rep. 27368. National Bureau of Economic Research.
- Bonhomme, S., T. Lamadon, and E. Manresa (2019). "A Distributional Framework for Matched Employer Employee Data". *Econometrica* 87.3, pp. 699– 739.
- Borjas, G. J. (2014). Immigration Economics. Harvard University Press.

- Bruns, B. (2019). "Changes in Workplace Heterogeneity and How They Widen the Gender Wage Gap". *American Economic Journal: Applied Economics* 11.2, pp. 74–113.
- Card, D., A. R. Cardoso, J. Heining, and P. Kline (2018). "Firms and Labor Market Inequality: Evidence and Some Theory". *Journal of Labor Economics* 36.S1, pp. 13–70.
- Card, D., A.R. Cardoso, and P. Kline (2016). "Bargaining, Sorting, and the Gender Wage Gap: Quantifying the Impact of Firms on the Relative Pay of Women". *The Quarterly Journal of Economics* 131.2, pp. 633–686.
- Card, D., J. Heining, and P. Kline (2013). "Workplace Heterogeneity and the Rise of West German Wage Inequality". *The Quarterly Journal of Economics* 128.3, pp. 967–1015.
- Chetty, R., N. Hendren, M. R. Jones, and S. R. Porter (2020). "Race and Economic Opportunity in the United States: an Intergenerational Perspective". *The Quarterly Journal of Economics* 135.2, pp. 711–783.
- Damas de Matos, A. (2017). "Firm Heterogeneity and Immigrant Wage Assimilation". *Applied Economics Letters* 24.9, pp. 653–657.
- Dostie, B., J. Li, D. Card, and D. Parent (2020). *Employer Policies and the Immigrant-Native Earnings Gap*. Tech. rep. 27096. National Bureau of Economic Research.
- Duleep, H. O. (2015). "The Adjustment of Immigrants in the Labor Market". *Handbook of the Economics of International Migration*. Vol. 1, pp. 105– 182.
- Dustmann, C., A. Glitz, U. Schönberg, and H. Brücker (2016). "Referral-based Job Search Networks". *The Review of Economic Studies* 83.2, pp. 514–546.
- Eliasson, T. (2013). Decomposing Immigrant Wage Assimilation the Role of Workplaces and Occupations. Tech. rep. 2013:7. IFAU.
- Gerard, F., L. Lagos, E. Severnini, and D. Card (2018). Assortative Matching or Exclusionary Hiring? The Impact of Firm Policies on Racial Wage Differences in Brazil. Tech. rep. 25176. National Bureau of Economic Research.
- Gobillon, L., H. Selod, and Y. Zenou (2007). "The Mechanisms of Spatial Mismatch". Urban Studies 44.12, pp. 2401–2427.
- Haltiwanger, J., H. Hyatt, and E. McEntarfer (2018). "Who Moves Up the Job Ladder?" *Journal of Labor Economics* 36.S1, S301–S336.
- Hellerstein, J.K. and D. Neumark (2008). "Workplace Segregation in the United States: Race, Ethnicity, and Skill". *The Review of Economics and Statistics* 90.3, pp. 459–477.
- Hirsch, B. and E. J. Jahn (2015). "Is There Monopsonistic Discrimination Against Immigrants?" *ILR Review* 68.3, pp. 501–528.
- Kerr, S. P. and W. R. Kerr (2011). *Economic Impacts of Immigration: A Survey*. Tech. rep. 16736. National Bureau of Economic Research.

- Kline, P., R. Saggio, and M. Sølvsten (2019). *Leave-out Estimation of Variance Components*. Tech. rep. 26244. National Bureau of Economic Research.
- Manning, A. (2011). "Imperfect Competition in the Labor Market". *Handbook* of Labor Economics. Vol. 4, pp. 973–1041.
- (2020). "Monopsony in Labor Markets: A Review". ILR Review, pp. 1-24.
- Massey, D. and N. A. Denton (1993). *American Apartheid: Segregation and the Making of the Underclass*. Harvard University Press.
- Neumark, D. (2018). "Experimental Research on Labor Market Discrimination". *Journal of Economic Literature* 56.3, pp. 799–866.
- Nordström Skans, O., P.-A. Edin, and B. Holmlund (2007). *Wage Dispersion Between and Within plants: Sweden 1985-2000*. Tech. rep. 13021. National Bureau of Economic Research.
- Oaxaca, R. (1973). "Male-Female Wage Differentials in Urban Labor Markets". *International Economic Review* 14.3, pp. 693–709.
- Orefice, G. and G. Peri (2020). *Immigration and Worker-Firm Matching*. Tech. rep. 26860. National Bureau of Economic Research.
- Reich, M., D. M. Gordon, and R. C. Edwards (1973). "A Theory of Labor Market Segmentation". *The American Economic Review* 63.2, pp. 359–365.
- Ruist, J. (2018). Tid för integration: En ESO-rapport om flyktingars bakgrund och arbetsmarknadsetablering (Time for integration. An ESO-raport about refugee backround and labor market integration.) 2018:3. ISBN: 978-91-38-24801-0.
- SCB (2020). Utrikes födda i Sverige (Foreign-born in Sweden). URL: https: //www.scb.se/hitta-statistik/sverige-i-siffror/manniskornai-sverige/utrikes-fodda/ (visited on 10/13/2020).
- Syverson, C. (2011). "What Determines Productivity?" *Journal of Economic Literature* 49.2, pp. 326–365.
- Tomaskovic-Devey, D., A. Rainey, D. Avent-Holt, N. Bandelj, I. Boza, D. Cort, O. Godechot, G. Hajdu, M. Hällsten, L.F. Henriksen, A.S. Hermansen, F. Hou, J. Jung, A. Kanjuo-Mrčela, J. King, N. Kodama, T. Kristal, A. Křížková, Z. Lippényi, S.M. Melzer, E. Mun, A. Penner, T. Petersen, A. Poje, M. Safi, M. Thaning, and Z. Tufail (2020). "Rising Between-Workplace Inequalities in High-Income Countries". *Proceedings of the National Academy of Sciences* 117.17, pp. 9277–9283.

Appendix A: Additional description and results

We estimate AKM models (Abowd et al. 1999) of the following form, separately for two groups g: immigrants and natives (see e.g. Dostie et al. 2020).

$$\ln(e_{git}) = \alpha_{gi} + \psi_{f(g,i,t)}^g + X_{git}\beta^g + \varepsilon_{git}$$
(2.8)

 α_{gi} captures individual time-invariant skills and other factors that are rewarded equally across all firms; $\psi_{f(g,i,t)}^{g}$ captures a group-specific firm pay premium that is rewarded equally across individuals in a group within the same firm; X_{git} are time-varying individual controls; and the error term ε_{git} captures random match effects, human capital shocks, and other unobservables.³³ A summary of the estimated parameters and model fit are included in Table A.1 below.

³³The firm- and worker-specific fixed effects are separately identified by job-to-job transitions of workers across firms. Cross-firm mobility is therefore crucial for identification (see e.g. Card et al. (2013)). Under exogenous mobility, both job-to-job transitions and the job assignment process depend solely on time-invariant unobservable characteristics of workers and firms, along with time-varying observables in X_{git} . Provided that the exogenous mobility assumption holds, the model requires a sufficiently high number of transitions in order to consistently estimate its variance components. Failing to have information on enough worker transitions can lead to substantial bias (see e.g. Andrews et al. (2008)), which in empirical applications motivates the pooling of multiple years of data for the model estimation under the assumption that firm-fixed effects are time-constant.

	Natives (1)		Immigrants
Standard deviation of log earnings	0.593		0.592
Number of person-year observations	45,873,468		6,784,236
Summary of parameter estimates			
Number of person effects	4,564,683		991,165
Number of firm effects	431,937		206,291
Std. dev. person effects (across persyr. obs.)	0.390		0.404
Std. dev. firm effects (across persyr. obs.)	0.200		0.277
Std. dev. Xb (across persyr. obs.)	0.211		0.201
Correlation of person/firm effects	0.130		0.020
RMSE of model	0.325		0.321
Adjusted R-squared of model	0.663		0.643
Correlation native/immigrant firm effects		0.626	
Share of variance of log earnings due to:			
Person effects	0.433		0.466
Firm effects	0.113		0.219
Covariance of person and firm effects	0.058		0.013
Xb and associated covariances	0.096		0.009
Residual	0.300		0.294

Table A.1. Summary of estimated AKM models for natives and immigrants

Notes: Results from two-way fixed effects models estimated separately for natives (column 1) and immigrants (column 2). Models include year dummies interacted with education dummies, and quadratic and cubic terms in age interacted with education dummies. The correlations of native and immigrant firm effects are calculated for the subset of dual-connected firms.

	J					
		Full sample			Analysis samp	ole
	1998-2017	1998	2017	1998-2017	1998	2017
	(1)	(2)	(3)	(4)	(5)	(9)
Panel A: Employees						
Immigrant	0.13	0.09	0.19	0.13	0.09	0.18
Immigrant from West	0.04	0.05	0.03	0.04	0.05	0.03
Immigrant from Rest of World	0.09	0.04	0.15	0.09	0.05	0.15
Native-segregated firms	0.17	0.21	0.12	0.17	0.22	0.11
Immigrant-segregated firms	0.01	0.00	0.02	0.01	0.00	0.01
Male	0.63	0.65	0.61	0.64	0.66	0.62
Age	40.53	40.24	40.26	40.29	39.93	40.22
Age ≤ 30	0.26	0.26	0.29	0.27	0.27	0.29
$\mathrm{Age} \geq 50$	0.28	0.27	0.28	0.27	0.26	0.28
Education, compulsory	0.15	0.23	0.11	0.16	0.25	0.11
Education, secondary	0.54	0.54	0.52	0.55	0.54	0.53
Education, upper secondary	0.30	0.22	0.36	0.29	0.21	0.35
Education, missing	0.01	0.01	0.01	0.01	0.01	0.01
Monthly earnings (2010 SEK)	24,706	20,828	27,966	24,624	20,838	28,004
No. of employees \times year	52,984,851	2,279,920	2,948,946	46,870,657	1,883,222	2,504,695
Panel B: Firms						
Fraction immigrants at employer	0.14	0.09	0.22	0.13	0.09	0.19
Yearly employer size	23.32	22.30	25.46	22.38	20.55	29.00
Share native-segregated firms	0.62	0.71	0.50	0.62	0.71	0.49
Share immigrant-segregated firms	s 0.06	0.03	0.10	0.05	0.03	0.07
No. of firms \times year	3,319,820	156,840	163,138	2,883,741	129,322	115,210

Table A.2. Summary statistics, full sample compared to analysis sample



(a) West

(b) Rest of World

Figure A.1. Immigrant-native earnings ratio - by region of birth



Figure A.2. Share immigrants in productivity decile – by country of birth



Figure A.3. Upward mobility, natives vs. immigrants
	Earnings	Mean decile	Mean decile	Premium	Sorting	Pay- setting
	gap	premium natives	premium immigran	gap its		
	(1)	(2)	(3)	(4)	(5)	(6)
All	0.124	0.076	0.078	-0.002	0.012	-0.014
By gender						
Male	0.158	0.082	0.084	-0.003	0.014	-0.016
Female	0.044	0.065	0.067	-0.002	0.009	-0.011
By age group						
Up to age 30	0.041	0.072	0.075	-0.002	0.011	-0.013
Between 31 and 50	0.198	0.078	0.080	-0.002	0.013	-0.015
50 and above	0.134	0.075	0.076	-0.001	0.013	-0.014
By region of origin						
West	-0.040	0.076	0.086	-0.011	0.006	-0.016
Rest of World	0.192	0.076	0.074	0.002	0.015	-0.013
By education						
Compulsory	0.132	0.079	0.073	0.006	0.020	-0.013
Upper secondary	0.177	0.071	0.077	-0.007	0.007	-0.014
Post secondary	0.099	0.077	0.078	-0.001	0.014	-0.015

Table A.3. Decomposition of immigrant-native earnings gap – Employee-weightedranking

Notes: Column (1) shows the mean log earnings gap between immigrants and natives in different groups. Columns (2) and (3) show the mean decile premium received by natives and immigrants, respectively. Column (4) gives the difference between column (2) and column (3). We decompose the gap in column (4) into a between-decile sorting effect (column (5)) and a differential within-decile pay-setting effect (column (6)). The sorting effect is weighted by the native premium, and the pay-setting effect by the immigrant shares.

Appendix B: Additional robustness checks





Figure B.1. AKM firm FE by log value added per worker (100 bins)



(a) Log earnings against productivity ranking (b) AKM firm FE against productivity ranking Figure B.2. More productive firms pay higher earnings (employee-weighted)

Earnings returns by productivity decile - weighted ranking



(a) All immigrants (b) By immigrant group

Figure B.3. Earnings returns to working in more productive firms (weighted ranking) Note: The figure plots $\hat{\theta}_d$ and $\hat{\theta}_d + \hat{\gamma}_d$ from equation (2.4) using the employee-weighted ranking of firms.

Earnings returns by productivity decile – by years since migration



Figure B.4. Earnings returns to working in more productive firms - YSM

Note: The figure plots $\hat{\theta}_d$ and $\hat{\theta}_d + \hat{\gamma}_d$ from equation (2.4), where the immigrant group is split by their years since migration (YSM).

3. Teacher Career Opportunities and School Quality

Co-authored with Erik Grönqvist and Lena Hensvik

Acknowledgments: We are grateful for comments from Peter Fredriksson, David Figlio, Claudia Goldin, Nils Gottfries, Lisa Höckel, Jesse Rothstein, Rune Vejlin and seminar participants at IFAU, UCLS, Uppsala University, Harvard University, IZA Summer School in Labor Economics as well as for helpful research assistance by Kristina Sibbmark.

This chapter is an updated version of IFAU working paper Grönqvist et al. (2020).

3.1 Introduction

Good teachers are an important input to schooling.¹ Still, most teacher labor markets are characterized by problems to recruit and retain talented teachers (Corcoran et al. 2004; Bacolod 2007; Fredriksson and Öckert 2007; Grönqvist and Vlachos 2016; Leigh and Ryan 2008). The compressed wage distribution – relative to other occupations – is often put forth as a primary driving force behind these problems, pointing to a greater need for policy-makers to find ways to tie the pay structure more closely to teacher quality.

There are several ways to make teacher compensation more responsive to teacher skills (see Jackson 2012 for a discussion). In this paper we study how the teacher labor market is affected by improved career opportunities, taking advantage of a unique promotion program for Swedish teachers. In response to deteriorating results in international comparisons like PISA and TIMMS, the Swedish government introduced a career step for experienced and skilled teachers by providing separate funding for a new position called 'career teachers' (Regeringen 2013b). Career teachers receive a substantial monthly wage increase (5,000 SEK (520 USD), which corresponds to nearly 20 percent of mean pre-reform wages) and continue to teach, but are also tasked to work with the school's pedagogical development, like being a mentor or initiating and leading development projects (Statskontoret 2015), to spread their competences to teacher peers.

The intention of the program was to make the teaching profession more attractive by rewarding skilled teachers, thereby increasing the wage dispersion, and to improve student outcomes by motivating, retaining and attracting high quality teachers (Regeringen 2013b). While similar types of career steps also exist in England, New Zealand, Australia, Scotland and Poland (Regeringen 2013c), we know little about their impact on teachers and students.

We address five central questions. First, who is promoted to become a career teacher? Second, what is the pass-through of the stipulated wage increase relative to non-promoted teachers' wages? Third, is there an effect of the career step on teachers' separations from their school, and from the profession overall? Fourth, is there an effect on the composition of teachers? Last, is there an impact on student performance?

The career teacher reform was rolled-out over four years, and the number of career teacher positions increased gradually. The allocation of positions to school districts was in proportion to their student population. School districts had discretion to assign career teacher positions to individual schools within

¹There is a large and growing literature documenting that teachers matter for both short term student outcomes, like test scores (Rockoff 2004; Rivkin et al. 2005; Leigh 2010; Chetty et al. 2014), and for longer term outcomes like college attendance and earnings (ibid.). It has, however, been difficult to find observable characteristics that are important for student outcomes (Jackson et al. 2014). Factors like education, cognitive ability, and personality, which are found to be important in other parts of the labor market, are only of marginal importance (Hanushek and Rivkin 2006; Rockoff et al. 2011; Grönqvist and Vlachos 2016).

the district, and at the schools it was delegated to the local principals to identify and recruit skilled teachers to the new position.²

We start by providing documentation of how the reform was implemented. This analysis suggests that school districts allocated the number of teacher promotions across schools in relation to school size. No other pre-determined observable school characteristics predict this allocation. Thus, even if school districts were free to allocate the promotions across schools, they did not target the promotions to schools with, for example, high teacher turnover or low student performance.

Considering who was promoted, the most salient pattern is that promotions within schools were given to teachers from the higher wage deciles of a compressed wage distribution, conditional on observable teacher characteristics. We find full pass-through of the state-funded stipulated wage increase onto wages. Thus, the reform increased wage dispersion both across and within schools, and there is no indication of compensatory behavior towards non-promoted teachers in regular wage negotiations.³

In terms of teacher mobility, promotions can both attract and retain individuals with a higher innate ability and induce individuals to exert more effort (see Lazear and Shaw (2007) and Oyer and Schaefer (2011) for summaries of the personnel economics literature). We expect the career step to make the current job more attractive for promoted teachers, but as promotion signals quality these teachers also become more attractive to other schools. For non-promoted teachers, on the other hand, the likelihood of quitting may go up if individuals care about their relative position (see e.g. Card et al. 2012; Dube et al. 2019). To quit may also be a rational response to a signal that you will not be promoted, but this signal is also observed by other schools. Quit rates may go down for non-promoted teachers if the career step improves school quality and the professional work environment. Based on tournament theory we would also expect the chance of becoming a career teacher to motivate marginal – not yet promoted – teachers to exert more effort (Lazear and Rosen 1981).⁴ It is less clear how motivation is affected by being promoted.

²The career step reform can be thought of as a performance-based promotion program where talented teachers are awarded a pay rise (Jackson 2012). Given the difficulties to identify good teachers based on observable characteristics, this leaves the principal with a substantial amount of discretion. Still, there is evidence suggesting that principals can identify teacher skills and that teacher assessments can predict high quality teachers (Rockoff and Speroni 2011; Cantrell et al. 2008), and that detailed screening measures used in recruitment (e.g. written assessments, interviews, and sample lessons) strongly predict teacher job performance (Jacob et al. 2018).

³In Sweden, teacher wages are set individually by the local principal. The idea is that competence, responsibilities and performance should determine the wage. Still, wages are very compressed and there is a strong equity norm among teachers. There can thus be pressure on principals to compensate non-promoted teachers in the regular wage revision.

⁴However, for promotions perceived as unfair in Chinese schools, Li (2019) finds an erosion of work moral and higher quitting probability among non-promoted teachers.

We study separations, teacher composition and student performance at the school level by exploiting the *timing* of the introduction of promotions in a difference-in-difference design. Our main identifying assumption is that schools' adoption of career teachers is unrelated to trends or school-level shocks that affect the outcomes. In line with this assumption, we find that no factors besides the number of students systematically influences the timing of participation. Our results suggest that schools with career teacher promotions have lower teacher turnover, both in general and in terms of teachers leaving the profession. This result is driven mainly by more senior teachers and applies both to teachers who were promoted and to those who were not promoted. In addition, the teaching pool improves in schools that participate in the reform as they are able to retain a higher share of certified and experienced teachers and teachers who themselves have higher compulsory school grades. Finally, we find non-negligible positive effects on student test scores in Math, English and Swedish in grades 3 and 6.

Our findings contribute to the policy-debate about how to improve school quality by tightening the link between teacher pay and performance. To the best of our knowledge, it is the first assessment of the effectiveness of performance based promotions. Most of the previous literature has focused either on smaller bonus programs (e.g. Clotfelter et al. 2008) or general salary increases (e.g. Figlio 1997; Gilpin 2012; Leigh 2012; Falch 2011; Hendricks 2014). Two recent studies consider the impact of changes from fixed to more flexible pay schemes. Biasi (2018) finds higher effort and teacher quality in school districts retaining more rigid pay schemes. In contrast, Willén (2019) finds no support for changes in teacher composition or student outcomes when individualized wage-setting was introduced to teachers in Sweden in the 1990s.

Besides providing mixed results, these studies highlight difficulties for policymakers to achieve a widening of the wage distribution through more local autonomy over teacher pay. In Wisconsin, around half of the school districts chose to maintain the fixed salary-scheme despite their new autonomy (Biasi 2018). Similarly, the wage decentralization reform in Sweden led to a relatively modest increase in wage dispersion among teachers (Willén 2019).⁵ The promotion program we study is different from these flexible pay reforms in several aspects: First, the stipulated wage increase is financed by earmarked funding from the central government, putting no strain on the local budget. Second, it restricts the number of teachers that can be promoted, preventing principals to raise wages for everyone. Third, it requires new responsibilities from the promoted teachers, which could make differential pay seem more acceptable. Consistent with this notion, we find that nearly all school districts

⁵Consistent with this, Söderström (2010) finds that the switch from centralized to individualized wage-setting increased entry wages, and wage dispersion late in the career.

chose to participate in the promotion program and that the reform had the intended first-stage impact on the actual wage distribution.

An important contribution of our study is also that we are able to go beyond the impact on teacher quality and also consider the impact on student performance. The direct evidence on the link between teacher pay and student outcomes is mixed with a focus on general pay raises rather than on increased wage dispersion.⁶ Our findings suggest that improved career opportunities for teachers in the form of performance-based promotions could be an important tool for policy-makers who aim to improve educational performance.

Finally, drawing on survey data for around a third of the school districts, we are able to shed some light on how the impact of the promotions vary with the local design of the career teacher positions. The data contain information about nine different career teacher responsibilities, which we use to identify two distinct career teacher types: "Coaching and mentoring CTs" and "Teaching development CTs". Our results highlight that teaching development is more important in reducing teacher turnover, but there is no differential impact on student results. Hence, while it is difficult based on the available data to conclude if there is a "best practice" in career teacher behavior our results indicate that increasing pay alone may not be enough to get the desired impact on teacher selection.

The paper proceeds as follows. It begins by describing the career teacher reform and the data in Section 3.2. In Section 3.3 we show how the career teacher promotions were allocated across schools and teachers, and in Section 3.4 we analyze the pass-through of the stipulated wage increase onto wages. Section 3.5 contains analyses of teacher turnover, teacher composition, and student performance. Finally, Section 3.6 concludes.

⁶Several studies from the last two decades also find a positive relationship between general teacher wages and student outcomes (see for example, Loeb and Page 2000; Dolton and Marcenaro-Gutierrez 2011; Britton and Popper 2016; Alva et al. 2017). On the other hand, Ree et al. (2018), who study an unconditional salary increase in Indonesia, and Cabrera and Webbink (2018), who study the impact of a wage policy program in Uruguay, find only modest to no effects of teacher pay on student outcomes. These studies focus on general pay raises rather than on performance-based promotions or increased wage dispersion. For performancerelated pay schemes, surveys by Jackson et al. (2014) and Neal (2011) find the earlier literature to be inconclusive. More recently, Dee and Wyckoff (2015) study a program where teachers are rated on a composite measure of teacher performance (value added in test scores and detailed classroom observations), and find that low performing teachers with the risk of dismissal and high performing teachers with the chance of financial rewards improved their performance. The US program most similar to the Swedish reform we study is probably the Minnesota Q Comp program, which encouraged Minnesota school districts to adopt locally designed payfor-performance schemes and complementary HRM practices. Sojourner et al. (2014) evaluate the impact of the reform using a difference-in-difference strategy, finding positive effects of the program on student performance. As in our case, local districts received additional funding to participate in the program.

3.2 Institutional setting and data

Teachers are formally hired by the school district ("huvudman") which, in the case of the public sector, is the municipality.⁷ In practice, principals at schools often make the hiring decision and set wages. Teachers in Sweden are covered by collective agreements, where teacher wage bargaining is decentralized and individualized, primarily set in negotiations between the teacher and the principal. This decentralized pay setting scheme came into place in 1996, replacing a centralized system where pay was set according to nationally-determined pay scales based on type of teaching and years of experience (see Willén (2019) and Söderström (2010) for details).

3.2.1 The career teacher reform

In July 2013 the Swedish government initiated a policy that introduced a new career step: the career teacher.⁸ It is financed by ear-marked government funding. The reform, akin to a performance-based promotion (Jackson 2012), targets talented teachers. Its intention is to improve student performance by improving teachers' career opportunities. Broadly speaking, it aims to make the teaching profession more attractive by raising wages for skilled teachers, thereby increasing wage dispersion in the profession, and by taking better advantage of teachers' competencies (Regeringen 2013b). Conceptually, it may entail both a sorting effect, if teachers that otherwise leave the school stay to a larger extent, and an effort effect. In addition, as outlined in more detail below, the reform may also entail spillovers within schools from promoted teachers onto their teacher peers.

The Swedish government also introduced other policy initiatives at about the same time as the career teacher reform. Two further education programs for teachers, the Boost for Mathematics and the Boost for Reading, were initiated in 2013 and 2015 respectively, and a more general program to increase teacher pay, the Teachers' Salary Boost, was initiated in 2016. The Salary

⁷Schooling is provided both by the public and by the private sector. The main public provider is the municipality. During the 2016/2017 academic year, 85 percent of compulsory school students attended public schools run by one of 290 municipal providers while 15 percent of compulsory school students attended voucher schools run by one of the 729 non-public providers (Skolverket 2018a). Children are free to choose which school to attend and incur no tuition fees regardless of provider. If a school is over-subscribed, proximity is the main guiding principle for allocation of places in public schools. We restrict our attention to public compulsory schools in our main analysis.

⁸The reform is regulated in Regulation 2013:70, see Regeringen (2013a). Formally, it introduces two career steps: career teachers ("förstelärare") and lecturers ("lektor"). Career teachers is a position for highly skilled teaching practitioners whereas lecturers is a position for teachers with an academic degree (licentiate or PhD) with a partly different job description. Since only around 1% of the promoted teachers are lecturers, we exclude teachers that are ever lecturers from our sample.

Boost also aimed at increasing wage dispersion, and implied a smaller unconditional wage increase to about half of the teaching pool financed by earmarked money from the state. In sensitivity analyses (see Panel F of Tables C.1 and C.2) we add controls for the three boost reforms. For these reforms to invalidate our results their implementation needs to interact with the career teacher reform. The conclusions regarding the effect of the career teacher reform do not change when we control for the boost reforms.

Design and roll-out of reform

The number of career teaching positions are allocated annually to school districts based on the national share of students across all educational tiers in the school district.⁹ For example, if a school district has 5 percent of students, it is allocated 5 percent of the career teaching positions. School districts in turn decide how to allocate the positions across schools within school districts, and to individual teachers. The total number of available positions each year is decided by the size of the total state grant. In 2013 the earmarked funding could finance around 4,000 positions across all educational tiers and types of school districts.¹⁰ This increased to around 14,000 by 2014 (Skolverket 2014). By 2016, the number of available positions was around 16,000 (Skolverket 2016). However, while the allocation of positions available to school districts is rulesbased, the school district need not acquire all the funding reserved to it. In 2013, approximately 75 percent of the funding (for approximately 3,000 positions) was acquired (Skolverket 2014). This increased to around 90 percent by 2016 (Skolverket 2016). For municipal compulsory schools, the group that we study, there are 290 school districts, all of which are *potentially* treated from 2013 onward.¹¹ Out of the 290 school districts, approximately 70 percent in our sample participate in 2013, where participation is defined by having at least one career teacher. By 2014, 97 percent of the school districts participate and in 2016, all but one school district participates. There is thus limited variation in participation across school districts over time. Figure 3.1 plots the school district's share of career teachers against the school district's share of students. The relationship is approximately linear, at least after 2013, which is in support of promotions being allocated in proportion to school district size. The corresponding figure for the variation across schools (see Figure A.2 in the appendix) shows that there is more variation within school districts.

⁹Formally, state grants are allocated according to this rule. A grant of SEK 85 000 (USD 9 600) is given per full-time career teacher. This also includes funding for employer contributions. School districts with fewer than 75 students apply for the grant from a common pool.

¹⁰The reform covered pre-school, compulsory school, upper-secondary school, Sami schools, schools for children with special needs and adult education from both public providers and voucher schools. In our main sample, we include municipal compulsory schools only.

¹¹We restrict our attention to municipal schools both because take-up of the program among voucher schools is substantially lower (Statskontoret 2015) and because we can only observe wages for a sub-sample of the voucher schools.



Figure 3.1. Allocation of promotions across school districts

Note: Each cross corresponds to one school district. The solid line is a linear prediction. The dashed line is the 45 degree line. For legibility, the figure excludes the four largest school districts. A figure with all school districts is included in Appendix Figure A.1.

The career teacher position

While the allocation of career teacher positions across schools rests with the school district, the promotion decisions are generally taken by school principals. Four minimum requirements need to be fulfilled to qualify for promotion. The teacher needs to be formally certified; have at least four years of experience with good testimonials from the principal; be able to demonstrate an ability to improve student outcomes and an interest to work with developing teaching; and be deemed particularly qualified by the school district in teaching and teaching-related tasks (Regeringen 2013a).

By 2016, around 14 percent of the compulsory school teachers in our sample have been promoted. The vast majority of promotions in our sample are internal – approximately 85–95 percent are working in the same school the year before they are promoted, depending on year. The career teacher positions are typically time limited but only around 2.5 percent of career teachers have a contract that lasts fewer than 12 months. A reason for having temporary contracts is to induce effort and to maintain flexibility. The proportion of permanent (i.e. not time limited) positions has increased over time and in 2016, over 45 percent were permanent. The same position is not transferable across schools or school districts.

According to the reform's regulation, teachers who become career teachers should receive a monthly wage increase of SEK 5 000 (USD 520). Ear-marked government funding is used to fund the wage increase. A main aim of the reform was to ensure that talented teachers keep teaching, as opposed to, for example, becoming principals or leaving the profession. The reform therefore stipulated that teaching and teaching-related tasks must constitute at least 50 percent of the career teachers' time. As a consequence, most career teachers have not reduced their teaching load in their new position (Statskontoret 2017).

Besides teaching, career teachers are given coaching and mentoring tasks and teaching development tasks aimed to, for example, improve teaching, train other teachers or work toward organizational change at their workplace. There may therefore be spillovers from the promoted teacher onto teacher peers in the same school (for evidence on teacher spillover effects, see Jackson and Bruegmann 2009).

Figure 3.2 describes these career teacher tasks in more detail. We extract this information from a survey to about 30 percent (92/290) of the school districts in our sample. The survey was conducted by the National Board of Education and the response rate is close to 100 percent.¹² The answers to the survey reveal that almost all school districts require that the career teachers are responsible for "Coaching other teachers" and for "Initiating pedagogical discussions". Besides that, there is a fair amount of variation in career teacher tasks across districts. We will return to this information in a heterogeneity analysis in Section 3.5.3 where we look at the differential response of the reform depending on the local design of the career teacher positions.

¹²Appendix Table A.3 compares school district characteristics depending on whether the district was included in the survey or not. Overall, the two groups appear highly similar although urban areas are somewhat underrepresented in the survey.



Figure 3.2. Career teacher tasks

Note: The figure shows the allocation of career teacher tasks by school districts. The data come from a survey conducted by the National Board of Education in 2015 covering around 30 percent (92/290) of school districts in our sample. The question was "What are the main tasks for your career teachers in addition/related to teaching?" and the respondents could answer yes or no to the nine tasks shown in the figure.

3.2.2 Data

To analyze the impact of the career teacher reform we combine administrative data from different Swedish registries held at Statistics Sweden. The underlying population for the analysis is the panel of Swedish schools for the years 2010 to 2016 and the teachers working at these schools, and is based on information from the Swedish teacher register (*Lärarregistret*).

The teacher register covers all school staff with educational duties employed at Swedish schools, and is collected as a part of the official statistics in the school area. Data is measured annually, in October each year, and for our purposes it contains information on person identifiers for teachers, information on where the teacher works, the teacher's experience and whether the teacher is certified.¹³ The teacher register can be linked to a school register that contains school level characteristics, such as number of students and

¹³To be precise, there is information on whether the teacher is qualified, i.e. has pedagogical higher education. Information on the formal occupational license introduced in 2011 is not available in the teacher register. The main requirement to receive the license is to hold proper credentials.

school district.¹⁴ Schools are defined using a combination of school name and municipality code.

Using person, school and year identifiers, we link the teacher register to a career teacher register that specifies whether the teacher is a career teacher. Using person and year identifiers, we also link the teacher register to demographic registers that include variables such as age, gender, level of education and field of specialization. We have data on teachers' 9th grade GPA from 1988 onward (cohorts born after 1972) which can be linked to our data using person identifiers.¹⁵ GPA is standardized in the full population by year of graduation to have mean 0 and standard derivation 1. In addition, we retrieve information on teachers' wages from the structural earnings statistics, which contains monthly full-time adjusted wages in SEK, measured in November each year. The wage data covers everyone working in the public sector (and about 50 percent of workers in the private sector). As we only include teachers working in municipal schools in our sample (see Section 3.2.2), we have complete wage data.

Our data on student performance are drawn from records of student test scores on standardized central tests in Math, English and Swedish (*Ämnesprovsregistret*). The tests are taken in grade 3 (Math and Swedish only), grade 6 and grade $9.^{16}$ In the first two years of our observation period (2010 and 2011), students took the national test in grade 5 instead of in grade 6. To use as much information as we can from the available data, we let the grade 5 test scores proxy for the performance in grade 6. Using the student-level data, the results of the exams are standardized by year to have mean 0 and standard deviation 1 in the full student population.¹⁷

¹⁴As we only include municipal schools in our sample, school district is proxied by municipality code.

¹⁵It follows that the share of teachers in our sample for which we observe GPA increases over time. In 2010, we have GPA for 26 percent of teachers. This increases by around three percentage points per year. In 2016, we have GPA for around 45 percent.

¹⁶The tests are typically graded by the students' own teachers based on centrally provided guidelines, and with a recommendation from the Swedish National Agency for Education that exams are co-evaluated by another teacher (Skolverket 2018b). In 2015, co-evaluation was standard in over 75 percent of public schools (Skolinspektionen 2018). The tests are taken during the spring semester. In our analysis, we associate a spring test score with remaining data the preceding fall semester. For example, we link test scores from spring 2013 to our data from fall 2012. Consequently, regressions in Section 3.5.5 that use student test scores only use data until 2015. ¹⁷The tests consist of different parts that are graded separately. How many parts a test has can vary by subject and grade, and at times year. Most parts generate a test score, but some parts are pass/fail (P/F) only. An overall test score is provided for the grade 6 tests (from spring 2013) and for the grade 9 tests (all years). Whenever an overall score is provided, we use that score. When an overall score is missing, we calculate a mean test score as the aggregate number of points divided by the number of parts of the test taken, for all parts of the test that are not P/F. The Grade 5 test in all three subjects in spring 2010 only consisted of P/F questions. For this test we calculate an overall score based on the proportion of parts that the student passed. Once each student has one test score per subject and grade, we standardize the test scores by year to

Main sample

We restrict our attention to teachers whose main occupation is teaching. In particular, we include only those individuals who receive their main source of income from teaching and who work at least 50 percent at their main school. If a teacher works at several schools in the same year, only the teacher's main school is included in the sample, defined as the school with the most extensive contract. Moreover, we only include municipal compulsory schools, thereby excluding teachers who work at different tiers of education (notably, in upper secondary education) or at voucher schools. Finally, we only include schools that exist for all seven years and employ at least three teachers per year between 2010 and 2016.

Separation measures

We consider two main separation measures. The first, denoted *separations*, measures the fraction of teachers who are no longer employed next year by the current school. The second, denoted *exits*, applies the additional restriction that the teacher is not working with teaching at any compulsory or upper secondary, public or voucher school in any capacity.

As explained above, we observe teachers at schools in October each year, while career teacher promotions can take place at any time during the school year (Statskontoret 2015). In terms of timing, we therefore relate the fraction who separate between October in year t-1 and October in year t to the presence of career teachers in the fall of year t. These career teachers are generally hired at some point during January to December in year t. Given data availability, the separation measures can be calculated from 2011 to 2016.

Number of observations

Table 3.1 shows the number of schools and teachers in our main sample. There are annually around 56,000 teachers working at just under 3,000 municipal compulsory schools. Participating schools shows the number of schools that have had at least one career teacher in year t or earlier. Promoted teachers considers teachers that have been promoted and is equal to one if the teacher has held a career teaching position in year t or earlier. The number of participating schools and promoted teachers are increasing over time from 2013 as the reform is rolled out.

Teacher pay determinants

Before turning to the analysis of the career teacher reform, it is useful to describe how teacher wages were determined prior to the reform. What discretion did school principals have in the wage-setting process? We examine this question in Appendix Table A.1, where we show the results from a standard Mincer regression of teacher characteristics on teacher pay. We focus on

have mean 0 and sd 1. A school is assigned a mean standardized test score by year, subject and grade based on the students that attend the school.

	2010	2011	2012	2013	2014	2015	2016
Schools	2,950	2,950	2,950	2,950	2,950	2,950	2,950
Teachers	56,150	56,080	55,785	55,864	56,787	56,711	58,583
Participating schools	0	0	0	1.039	2.277	2.532	2.636
Promoted	ů.	ů.	ů	1,007	_,	_,	_,000
teachers	0	0	0	1,420	6,317	7,457	8,157

 Table 3.1. Schools and teachers per year

the pre-reform period 2010 to 2012, to avoid that the relationships reflect the outcomes of the career teacher reform of interest in this paper. As expected, wages grow with education and experience, and are higher for certified teachers and those on permanent contracts. Perhaps surprisingly, there is a negative wage-tenure relationship, which is likely to reflect that new hires (who are included in the omitted tenure category) are able to bargain their wages in times of teacher shortages. In addition, it is useful to note that the R² is only 0.67, even conditioning on a wide set of observable characteristics. This implies that principals to some extent do use their discretion in wage-setting.

3.3 Allocation of career teacher positions across schools and teachers

3.3.1 Allocation across schools

We are interested in studying the effect of improved career opportunities for teachers on school quality. We first explore how school districts allocate career teacher positions across schools and teachers. Panel A of Figure 3.3 shows how many of the 2,950 schools that participate by year – i.e. the timing of treatment for the schools in our sample. Panel B instead considers treatment intensity by plotting the distribution of the share of promoted teachers at a school separately by year. On average, between 1.4 and 2.2 teachers are promoted at a school (see Appendix Table A.2, which includes school summary statistics). The number of teacher promotions increased by a factor of four between 2013 and 2014 followed by a more moderate increase between 2014 and 2016 (see also Table 3.1). As a result, we see that the fraction of schools without career teachers declined from 65 to under 25 percent between 2013 and 2014. In 2016, on average 15 percent of teachers were promoted at a school that employed a career teacher, while 16 percent of schools lacked a career teacher.

To descriptively consider how the career teacher positions were allocated across schools, Table 3.2 relates the presence of career teachers to lagged school characteristics. In particular, it presents the results of the regression:

$$CT_{sdt} = \phi_{dt} + \beta_t X_{sdt-1} + \varepsilon_{sdt} \tag{3.1}$$

where CT_{sdt} is a dummy equal to 1 if school *s* in school district *d* has a career teacher in year *t*, ϕ_{dt} are school district fixed effects and X_{sdt-1} are lagged school characteristics. In Appendix Table B.1 we estimate equation 3.1 instead using the fraction of of career teachers out of all teachers at the school in year *t* as the dependent variable. All variables are measured at the school level. We run separate regressions by year between 2013 and 2016. We focus on when the school first participates in the reform and thereby only include schools that have no (never or not yet) career teachers in t - 1.

The results suggest that the probability of having career teachers is increasing in school size: doubling the number of students increases the likelihood that the school has at least one career teacher (in any year) by between 10 to 20 percentage points. Besides this factor, there appear to be little systematic relation between having career teachers and observable school characteristics. Most surprising is perhaps that there is no systematic relationship between the allocation of career teacher positions across schools and the lagged separation rate. There is thus no indication that school districts allocated the career positions to schools with greater difficulties in retaining their teaching pool.¹⁸

3.3.2 Allocation across teachers

Turning to *who* was promoted, Table 3.3 presents pre-reform (2012) summary statistics for our sample, separately by whether the teacher is ever promoted, never promoted and the full sample. The selection of teachers for promotion officially rested with the school district but was in practice often taken by the school principal. In line with the eligibility requirements, 97 percent of promoted teachers were employed on a permanent contract and were certified in 2012. We also see that promoted teachers are slightly more likely to be female, have slightly less experience and are slightly younger than those who are not promoted. Considering their educational field of specialization, over half of ever career teachers are specialized in either Swedish and social sciences or math and natural sciences, which is higher than those who are not promoted.

¹⁸In Table B.2, we also include the average test score among third- and sixth-graders, which are available for the subset of schools that have students in those grades (86/68 percent of the schools have students in grade 3/6). Reassuringly, student performance does not predict the selection of schools with promotions.



(a) Participating schools





Figure 3.3. Variation in reform participation at school level

Note: Panel A shows the number of non-participating and participating schools. Participation is defined as having at least one career teacher. Panel B shows the fraction of schools at different shares of career teachers per year.

	At least one CT at school				
	2013	2014	2015	2016	
School characteristics $t - 1$:					
Log nr students	0.210***	0.290***	0.211***	0.162***	
	(0.019)	(0.015)	(0.045)	(0.048)	
Student-to-teacher ratio	-0.010***	-0.011***	-0.007	-0.003	
	(0.003)	(0.003)	(0.007)	(0.006)	
Separation rate	0.022	0.003	0.081	0.214	
	(0.079)	(0.091)	(0.192)	(0.192)	
Exit rate	-0.067	0.114	0.050	-0.287	
	(0.106)	(0.120)	(0.220)	(0.217)	
Certified (share)	0.014	0.252*	0.240	-0.199	
	(0.157)	(0.145)	(0.246)	(0.246)	
Female (share)	-0.066	0.170*	-0.139	0.098	
	(0.076)	(0.091)	(0.153)	(0.170)	
Mean age (years)	-0.010**	-0.006	-0.000	0.008	
	(0.005)	(0.005)	(0.007)	(0.007)	
Mean experience (years)	0.003	-0.001	0.000	-0.004	
	(0.004)	(0.005)	(0.008)	(0.007)	
Math/natural science (share)	0.042	-0.110	-0.203	-0.036	
	(0.079)	(0.110)	(0.146)	(0.171)	
Swedish/soc. science (share)	0.073	-0.077	-0.092	0.232	
	(0.063)	(0.089)	(0.133)	(0.144)	
District FE	Yes	Yes	Yes	Yes	
R^2	0.349	0.411	0.471	0.572	
Ν	2,941	1,882	611	346	
Mean dep. var.	.35	.65	.38	.25	
Nr school districts	290	277	207	169	

Table 3.2. Factors that predict selection of schools

Note: This presents the results of regressions of $CT_{sdt} = \phi_{dt} + \beta_t X_{sdt-1} + \varepsilon_{sdt}$ where CT_{sdt} is a dummy equal to 1 if the school participates in the reform in year *t*. Variables are measured at school level. Regressions are estimated separately by year and only include schools that have not (never or yet) participated as well as schools that participate for the first time. Standard errors are clustered at school district level and reported in parenthesis.

	Ever career		Never career		Full sample	
	Mean	sd	Mean	sd	Mean	sd
Female	0.83	(0.38)	0.78	(0.42)	0.78	(0.41)
Age (years)	43.27	(7.87)	47.02	(10.67)	46.47	(10.39)
Experience (years)	13.75	(7.77)	16.01	(11.27)	15.68	(10.86)
Permanent contract	0.97	(0.16)	0.90	(0.30)	0.91	(0.28)
Certified	0.97	(0.17)	0.90	(0.30)	0.91	(0.28)
Monthly wage (SEK)	27,943	(2,639)	27,426	(2,894)	27,501	(2,864)
Educ. specialization						
Math/natural science	0.25	(0.43)	0.14	(0.35)	0.16	(0.37)
Swedish/social science	0.28	(0.45)	0.19	(0.39)	0.20	(0.40)
Languages	0.08	(0.27)	0.05	(0.22)	0.06	(0.23)
Vocational	0.06	(0.24)	0.11	(0.31)	0.10	(0.30)
Other teaching	0.32	(0.47)	0.44	(0.50)	0.43	(0.49)
Non-teaching	0.01	(0.12)	0.06	(0.23)	0.05	(0.22)
Observations	8,160		47,625		55,785	

Table 3.3. Teacher summary statistics, 2012

To more formally assess who was promoted, we estimate linear models using OLS by regressing a dummy for ever being a career teacher during our sample period, $EverCT_{ist}$, on teacher characteristics X_{ist} (age, gender, teacher GPA, wage decile, educational level and specialization, type of contract, certification, tenure and experience):

$$EverCT_{ist} = \beta X_{ist} + \delta_s + \varepsilon_{ist}$$
(3.2)

School fixed effects are also included. Regressions are estimated using only data from the pre-reform year 2012. Figure 3.4 presents the results of this analysis. It shows linear predictions of promotion with 95% confidence intervals. Full regression results are included in Appendix Table B.3.

The estimates confirm that the likelihood of being promoted increases slightly with tenure while it decreases with age. Experience shows an inverted U-shape. The likelihood of being promoted also increases marginally with teachers' compulsory school GPA. Wage decile, which measures in which decile in the school wage distribution the teacher is in 2012, predicts promotion most strongly. The results indicate, for example, that someone at the highest wage decile in their school has over a 25% likelihood of being promoted. In addition, we find that women as well as those with a pedagogical educational specialization in math and natural sciences are slightly more likely to be promoted.



Figure 3.4. Predicted probabilities of ever being promoted, 2012

Note: This presents linear predictions from the regression $EverCT_{ist} = \beta X_{ist} + \delta_s + \varepsilon_{ist}$, using data only from the year before the reform is introduced (2012). Full results are in Appendix Table B.3. Only in Panel (e) do we include the teacher's standardized 9th grade GPA. $EverCT_{ist}$ is an indicator equal to 1 if the teacher is ever a career teacher during our sample period. Standard errors are clustered at school district level.

3.4 Pass-through of stipulated wage increase on promoted teachers' wages

A central component of the reform was to give promoted teachers a wage increase. Considering aggregate wage effects among teachers, Figure 3.5 shows the distribution of teacher wages in the years surrounding the reform. It clearly suggests that the policy had a meaningful effect on the level and distribution of wages: mean full-time wages were around SEK 27 000 (approx. USD 2 800) between 2010 and 2012, and nearly SEK 31 000 (approx. USD 3 200) in 2015, representing a 15 percent increase in mean wages after introducing the promotion program.¹⁹ In addition, the reform seems to have created a much more prominent right hand tail of the wage distribution, as expected to see in professional occupations where pay aligns with performance.



Figure 3.5. Kernel density of teacher wages (2010–2015)

Note: The figure shows the distribution of teacher monthly full-time wages for each year before and after the career teacher reform.

Figure 3.6 shows mean wages for those that never become career teachers during our observation period compared to those that become career teachers, by year of promotion (2013 to 2016). Prior to promotion, mean wages are very similar. While non-career teacher wages trend upwards slightly over time, mean wages for career teachers increase sharply upon promotion.

¹⁹Wages are expressed in nominal terms since this is a zero inflation period. From 2010 to 2016 CPI increased by 4.3 percent with an average inflation rate of 0.07 percent.



Figure 3.6. Wages and the timing of promotion

Note: The figure shows the mean teacher wage for teachers divided by the year of promotion to career teacher. The solid line shows the wage of never promoted teachers.

The figure supports that wages increased after the implementation of the reform. From a theoretical perspective it is not clear, however, that we should expect full pass-through of the stipulated wage increase onto promoted teachers wages. If, for example, job satisfaction depends on relative pay as shown by Card et al. (2012), school principals may have incentives to, at least partly, compensate non-promoted teachers in local wage negotiations. To more formally assess how wages for promoted teachers differ from those who are not promoted, we estimate regressions of the following form:

$$\ln(w_{ist}) = \alpha_i + \lambda_t + \theta C T_{ist} + \beta X_{ist} + \varepsilon_{ist}$$
(3.3)

where $\ln(w_{ist})$ are log monthly full-time-equivalent teacher wages and CT_{ist} is a dummy equal to one if the teacher is promoted.²⁰ We include year fixed effects λ_t to control for time effects common to all individuals and teacher fixed effects, α_i , to control for individual-specific heterogeneity in wages. We therefore rely on within-individual deviations to identify estimates of θ . To account for correlation in wages between teachers that work in the same school district, standard errors are clustered at the school district level.

²⁰As mentioned above, promotions are not necessarily permanent. However, once an individual has held a CT position, we consider the individual to be treated. The dummy is therefore equal to 1 from year *t* onward, if the teacher is promoted in *t*.

The results are shown by Panel A of Table 3.4. Column (1) uses the full sample of teachers while column (2) only includes non-promoted teachers as well as career teachers in their first year of promotion. They suggest that the wage increase associated with a promotion is approximately 15%.²¹ In Panel B, we use the monthly wage in Swedish crowns (SEK) as the outcome. These results confirm that the wage impact of a promotion is very close to the 5 000 SEK wage increase stipulated by the reform, particularly if we consider the wage increase associated with the first time a teacher is promoted, shown in column 2.²²

	(1)	(2)
Sample	Full	First time CT
Panel A: ln(wage)		
Promoted	0.149***	0.141***
	(0.002)	(0.002)
R^2	0.958	0.904
Ν	374,108	260,619
Panel B: Monthly wage (SEK)		
Promoted	5329.2***	4744.3***
	(65.4)	(58.6)
R^2	0.954	0.900
Ν	374,108	260,619
Year FE	Yes	
Individual FE	Yes	
Controls	Yes	Yes including lag wage

Table 3.4. Wage effects of promotion

Note: The table provides results of the regressions $y_{ist} = \alpha_i + \lambda_t + \theta CT_{ist} + \beta X_{ist} + \varepsilon_{ist}$. Controls are included for female, age, level of education, teacher certification, permanent contract, educational specialization, experience and tenure. Standard errors are clustered at school district level and reported in parenthesis. In specification (2) the sample is censored to only include the first year of becoming a career teacher.

Given that selection of career teachers is non-random, we may worry that those that are promoted are on a different wage trend than those who are not promoted. Indeed, the analysis above showed that career teachers are more often taken from higher wage deciles in the schools that they work. As a more formal complement to Figure 3.6, we consider whether there are differences in pre-treatment wage trends as well as the dynamics after promotion by esti-

²¹Appendix Table B.4 shows results from alternative specifications including models without teacher fixed effects.

²²The fact that the estimate in column (1) of Panel B is slightly larger probably reflects the outcome in subsequent wage negotiations, where pay raises are based on the current wage.

mating an event-time specification as follows:

$$\ln(w_{ist}) = \alpha_i + \lambda_t + \sum_{\tau \neq -1} \gamma_\tau D_i \mathbb{1}[\tau] + \beta X_{ist} + \varepsilon_{ist}$$
(3.4)

t is calendar time and τ is event-time. τ denotes the time relative to when the teacher is first promoted, which occurs when τ equals 0. Observations three or more event years before treatment ($\tau \leq -3$) or two or more event years after ($\tau \geq 2$) are grouped. D_i is a dummy variable indicating whether the teacher is ever promoted and $1[\tau]$ is an indicator function equal to 1 in τ . The year before treatment is omitted.

From Figure 3.7, which plots the estimates of the parameters γ_{τ} , we see a clear jump in wages the year the teacher becomes promoted relative to the year prior to promotion. The higher wage is persistent, but does not appear to grow, over time. The non-zero effect prior to promotion suggests that the wage trajectories for promoted and non-promoted individuals are nearly but not exactly parallel; those who are selected for promotion appear to be on a slightly higher wage trend.



Figure 3.7. Dynamic effects in teacher wage outcomes

Note: Figure 3.7 displays γ_{τ} -coefficients with 95%-confidence intervals from estimating equation 3.4. $\tau = -1$ is omitted. Controls included in X_{ist} are dummies for female, certification, permanent contract, age, level of education, educational specialization, experience and tenure. Standard errors are clustered at school district level.

3.5 Impact of promotions on teacher separations, composition and student performance

The results from section 3.4 suggest that the career teacher reform had a substantial impact on the wages of promoted teachers. In this section we examine its impact on teacher separations, teacher composition and student performance. To this end, we use data on outcomes aggregated to the school-level, and rely on variation in the *timing* of participation across schools, which we show below appears to be unrelated to observable school characteristics.

We first discuss the empirical model and the validity of the identifying assumptions in Section 3.5.1. Section 3.5.2 presents results on teacher turnover while Section 3.5.3 provides a heterogeneity analysis. Section 3.5.4 focuses on teacher composition and Section 3.5.5 finally looks at student performance.

3.5.1 Empirical strategy and identification

An empirical challenge we face is the lack of a natural control group: all school districts can potentially participate in the reform, and the extent to which they can potentially participate is determined by their share of students (see Section 3.2.1). Indeed, all but one school district in our sample participates and there is limited variation in the timing of participation.

We therefore estimate the impact of the new career opportunities using school-level variation, rather than school district-level variation, in the appointment of career teachers. More specifically, we employ a difference-indifferences strategy that compares outcomes such as the separation rate in schools that have at least one career teacher to schools that do not (never or yet) have career teachers. We estimate models of the following form:

$$y_{st} = \gamma C T_{st} + \delta_s + \lambda_t + \beta X_{st} + \varepsilon_{st}$$
(3.5)

 y_{st} is the outcome of interest in school *s* in year *t* and CT_{st} is a variable indicating if school *s* has at least one career teacher in year *t* (i.e. if it has participated in the reform in year *t*). We also report results when we use the fraction of career teachers at the school in year *t* relative to the total number of teachers in t - 1 as CT_{st} to exploit that treatment intensity may vary across schools, thus gaining more variation.²³ Furthermore, we control for the student to teacher ratio (defined as the number of students in year *t* as the share of the number of teachers in t - 1) and the log number of students in *t*, captured by X_{st} . δ_s and λ_t are school and year fixed effects respectively.

The empirical strategy will identify relative differences in outcomes across schools rather than aggregate effects on school quality. It relies on the assumption that, in absence of appointing a career teacher, the outcome variable would have evolved in parallel in participating and non-participating schools. For our empirical strategy to work, the timing of when the reform is implemented in specific schools must be uncorrelated with changes in other determinants of the outcome that we do not control for.

School districts decide how to allocate the career teaching positions across schools in their district – participation is not random. There is potential selection both with regards to which schools participate and when they partici-

²³We compute the share using lagged number of teachers as separations may be affected by the reform.

pate. Reassuringly, the results in Table 3.2 suggest that no (lagged) observable school characteristics besides school size systematically predict the probability of participating in the reform in a given year. To further assess the identifying assumption, we consider whether any factors predict the timing of participation, conditional on having at least one promoted teacher between 2013 and 2016. In particular, we are interested in whether pre-reform observed school characteristics are orthogonal to the year of first participation, conditional on sometime participating in the reform. To this end, we estimate the following regression separately by year for schools that do not yet participate:

$$Year_{sd} = \phi_{dt} + \beta_t X_{sdt-1} + \varepsilon_{sdt} \tag{3.6}$$

Year_{sd} is equal to the year that the school first participates (i.e. a year in the interval 2013 to 2016), ϕ_{dt} are school district fixed effects and X_{sdt-1} are lagged school characteristics.²⁴ The results, included in Appendix Table B.5 and B.6, suggest that schools with more students first participate earlier. No other factors appear to systematically influence the timing of participation. We control for the log number of students and the student-to-teacher ratio in the regressions below.

To further assess the validity of the identifying assumption, we also perform an event-study analysis to rule out pre-participation trend differences in wages and separation rates between teachers in promoting and non-promoting schools using a dynamic version of equation (3.5):

$$y_{st} = \sum_{\tau \neq -1} \gamma_{\tau} D_s \mathbf{1}[\tau] + \delta_s + \lambda_t + \beta X_{st} + \varepsilon_{st}$$
(3.7)

 D_s is a dummy variable indicating whether the school ever participates in the reform and τ denotes the time relative to when the school first participates, which occurs when τ equals 0. The year before treatment is omitted.

3.5.2 Results on teacher turnover

Table 3.5 shows the γ -coefficients obtained when estimating the model given by equation 3.5. We focus on four outcomes. First, in column (1) we again confirm that the implementation of the new career step translates into a wage difference between promoting and non-promoting schools; the effect is of similar size as in section 3.4.²⁵ In line with the intentions of the reform we also see increased wage dispersion in schools with career teachers (see column 2).

²⁴This analysis was inspired by Deshpande and Li (2019) who provide a similar analysis to assess the systematic factors predicting the timing of closings of Social Security Administration field offices. It has many parallels to the methodology in Jackson (2010) who uses variation in the time of adoption to analyze a program in Texas that pays students and teachers for passing Advanced Placement exams.

 $^{^{25}}$ We infer this from column (1) of Panel B.

Wage dispersion is defined as the variance in log wages at the school-level. In columns (3) and (4) we consider separations and exits from teaching. In column (3), we look at school separations in general. Our results suggest that schools with at least one career teacher have a one percentage point lower separation rate, which corresponds to roughly four percent. In column (4), we focus on the fraction of teachers exiting the teaching occupation. This effect is also negative, but smaller in magnitude and does not reach statistical significance at the 10%-level.

In Panel B, we show results when we take the treatment intensity into account by relating the outcomes of interest to the fraction of career teachers at the school-level. While this model gives us more variation and provides estimates that are easier to interpret, it also requires that the "share of promoted teachers" at the school level is exogenous. In Appendix Table B.1 we do not find that any (lagged) observable school characteristics systematically predict the fraction of career teachers in a given year. Bearing this in mind, the estimates suggest that increasing the share of career teachers by 10 percent at a school is associated with a reduction in the separation rate by around two percentage points, or nine percent, and a reduction in exits from the profession with 0.6 percentage points, or five percent. Hence, separations in general and from the occupation is reduced and the effects are significant in size. Moreover, a 10 percent increase in the share of career teachers increases the variance of log wages by 0.0024 units, or 25 percent.

In Figure 3.8 we plot estimates of γ_{τ} from equation 3.7 to further investigate whether the timing of exposure to the reform is exogenous. Reassuringly, outcomes evolve very similarly in promoting and non-promoting schools prior to the implementation of the new career step. This analysis also shows that the responses grow over time, which is likely to reflect that the number of career teachers increases after the first year of participation in the reform.

In summary, we find that schools that promote teachers to 'career teachers' have higher average wages, a wider wage distribution, and lower teacher turnover both in terms of general separations and exits from the teacher profession. The impact of the reform tends to grow over time, and there appear to be no pre-reform effects.

Appendix C presents a range of robustness checks to ensure that our results are robust to alterations of the empirical model. Table C.1 shows the results when the treatment is defined as hiring at least one career teacher, whereas Table C.2 considers treatment as the share of teachers at the school that have been promoted. Panel A of Tables C.1 and C.2 first report the baseline results from Table 3.5. Panels B to F include sensitivity analyses. In Panel B we omit the time-varying school controls (i.e. the number of students and the student/teacher ratio); in Panel C we restrict the comparison to outcomes among teachers who were eligible for the career-teacher promotions, i.e. to certified teachers with at least four years of teaching experience; in Panel D we weight the regressions with the number of teachers in t - 1; in Panel E we restrict the

	(1)	(2)	(3)	(4)
	Log wages	Wage disper- sion	Separations	Exits
Panel A:				
At least one CT	0.019***	0.004***	-0.010*	-0.004
	(0.001)	(0.000)	(0.005)	(0.003)
R^2	0.914	0.583	0.339	0.238
Ν	17,689	17,689	17,689	17,689
Panel B:				
Share CT	0.165***	0.024***	-0.207***	-0.057***
	(0.007)	(0.001)	(0.025)	(0.015)
R^2	0.921	0.596	0.345	0.239
Ν	17,689	17,689	17,689	17,689
Year FE:s	Yes	Yes	Yes	Yes
School FE:s	Yes	Yes	Yes	Yes
School controls	Yes	Yes	Yes	Yes
Control mean	10.231	0.009	0.222	0.114

Table 3.5. Wages and separations in participating vs. non-participating schools

Note: In the table, we relate the change in the presence of career teachers within a school in year *t* to the change in mean wages (col. 1), wage dispersion (col. 2), school separations (col 3) and exits (col 4) (see Section 3.2.2 for the exact definition of these variables). At least one CT is an indicator variable equal to one from the first year that the school has promoted a career teacher onward. Share CT at school is defined as the number of CT in *t* divided by the number of teachers in t - 1. School controls are the student-to-teacher ratio and the log number of students. Standard errors are clustered by school district and reported in parenthesis.



Note: The figure plots γ_{τ} -coefficients with 95%-confidence intervals from estimating eq. 3.7. It shows the evolution of log wages, wage dispersion, school separations and exits from the teaching professions within schools before and after promoting at least one career teacher. $\tau = -1$ is omitted.

comparison to schools that had at least one career teacher during 2013–2016; and in Panel F we control for whether the school participated in the Boost for Mathematics, the Boost for Reading, or the Teachers' Salary Boost, three other policy initiatives implemented in 2013, 2015 and 2016 respectively. We conclude that the results are very stable across different empirical models and samples for both treatment variables.

3.5.3 Heterogeneity analysis

By type of career teacher

To understand how the results differ for different types of career teachers, we perform two types of heterogeneity analysis. First, we assess how the impact of the career teachers varies with the local design of the positions. To this end, we use the school district survey information about career teacher tasks described in Section 3.2.1. To reduce the dimensionality of the data we use principal component analysis focusing on the simplest two-component

case in which there are two possible career teacher designs. Based on this exercise, we identify two career teacher designs, which we label "Coaching and mentoring" and "Teaching development".²⁶ Table A.3 in Appendix shows that districts adopting a "Coaching and mentoring" design are somewhat more common in urban areas and have more students and teachers compared to districts adopting a "Teaching development design". Reassuringly, wages and exit rates appear almost identical in the two groups. Figure 3.9 shows the impact on wages and separations separately in school districts with the two types of career teacher designs. As we would expect, wage responses are the same in the two groups. However, the separation responses, both in terms of school turnover and exits from the teaching profession, appear stronger in districts where the career teachers are instructed to engage more in teaching development (see Panel A in Table B.7 for the estimates obtained when running eq. 3.5 separately for the two groups). This heterogeneous impact suggests that the reduction in teacher turnover is not only due to the pay increase.

Second, we calculate the four outcomes (log wages, wage dispersion, share separate and share exit) separately for junior and senior teachers.²⁷ We then estimate equation 3.5 using the outcomes for senior and junior teachers respectively. The results, presented in Panel B of Appendix Table B.7, show that the effects of the reform are driven by the senior teachers exclusively. In particular, it is wages and wage dispersion for senior teachers that respond positively to the reform, and it is senior teachers who quit to a lower extent. This is in line with the reform's design, as one criteria for promotion was to be experienced (see Section 3.2.1).

By district type

The probability of leaving a school should be related to the outside options for teachers in the local area. Thus, we would expect the reform to have a greater impact in areas with better outside options both in and outside teaching. To assess this, we have estimated equation 3.5 separately for urban and rural school districts. The results of this analysis, presented in Appendix Table B.7, show that the wage effects are very similar across urban and rural school districts. Given that schools have closely followed the rules stipulated by the reform, this is not surprising. The results on separations, on the other hand, suggest that the reduction in separations is largely driven by schools located in urban areas. For schools in urban areas, participating in the reform reduced separations by approximately 1.7 percentage points, or 7 to 8 percent. For

²⁶Figure A.3 shows the scatter plot of the loadings on the two components. We note that pedagogical tasks such as "Subject didactics" (CTtask 7) and "Leading pedagogical discussions" (CTtask 2) load on both components, but that mentoring colleagues, interns and students (CTtask 1, 3 and 8) load on component 1 (Coaching and mentoring) only whereas teaching development tasks such as "Research engagement" (CTtask 5) load on component 2 only (Teaching development).

²⁷We define senior teachers as those with at least five years of teaching experience.



Figure 3.9. Dynamic school responses by local CT design

Note: The figure plots γ -coefficients with 95%-confidence intervals from estimating eq. 3.7 separately for districts using "Coaching and mentoring CTs" and districts using "Teaching development CTs". The survey information is available for 1/3 of the districts used in the main analysis.

schools in rural areas, the result is marginally negative but not statistically significant at the 10%-level.²⁸

The second heterogeneity analysis focuses on school districts that appear to follow an explicit decision rule. As explained in Section 3.2.1, career teaching positions were allocated across school districts nationally using the school district's share of students. Figure A.2 shows that there is much more variation within school districts – school districts have not systematically followed this rule to allocate positions across schools. Nevertheless, the extent to which this rule is used differs by school district. To identify school districts that seem to allocate positions across schools according to this rule, we correlate the share of career teachers with the share of students within school districts. The distribution of the correlation coefficient is displayed in Figure B.1. We then estimate equation 3.5 for schools that are above the 50th and 75th percentile of

 $^{^{28}}$ The results that consider treatment intensity – share of teachers promoted – find negative and statistically significant effects for both urban and rural areas, but the effects in urban areas are of a magnitude 1.5 times higher than those in rural areas. These results are available upon request.

correlation coefficients. The results, presented in Panel D of Appendix Table B.7, show that there is little heterogeneity irrespective of whether the school district allocated positions according to this rule or not.

Responses among non-promoted teachers

Finally, we consider the response among non-promoted teachers. While the career teacher promotions are expected to reduce turnover among promoted teachers it is, as previously discussed, not clear how they will impact turnover in the pool of non-promoted teachers. Since the reform entailed both higher pay and increased responsibilities for teaching development and coaching other teachers, it is possible that part of the overall impact on turnover rates reflects reduced separations among non-promoted teachers, due to, for example, positive effects on work environment and support. At the same time, promotions could also increase teacher turnover among teachers *not* selected for a career teacher promotion via the kind of "discouragement effects" documented by Card et al. (2012).

We explore this aspect by comparing non-promoted teachers in schools that have and have not introduced promotions. A concern is that the pool of nonpromoted teachers is likely different in schools with and without promotions (due to the selection of teachers for promotions documented in Section 3.3). To make the comparison as credible as possible, we therefore also account for the teacher's observable lagged characteristics with the purpose of comparing non-promoted teachers with similar chances of being promoted (based on observables). The analysis is based on the following model:

$$y_{ist}^{notCT} = \gamma CT_{st} + \alpha X_{ist-1} + \delta_s + \lambda_t + \beta X_{st} + \varepsilon_{ist}$$
(3.8)

where y_{ist}^{notCT} is a dummy variable equal to 1 if a non-promoted teacher separates or exist, and CT_{st} is either a dummy for having at least one career teacher or the fraction of career teachers at the school in year *t*. The observable teacher characteristics are included in X_{ist-1} . In addition to the teacher characteristics we, as before, account for the student to teacher ratio and the log number of students, as well as school and year fixed effects.

Results are presented in Table 3.6. Interestingly, these suggest that nonpromoted teachers also decrease their separation rate when promotions are introduced at the school. The magnitudes are smaller than the overall effects documented in Table 3.5, but suggest that the lower turnover rates in schools with promotions are partly driven by the pool of non-promoted teachers.²⁹

²⁹In Table B.8 we use the wage of non-promoted teachers as outcome when we estimate equation 3.8. The results suggest a slight tendency for non-promoted teachers in schools having implemented the reform to be compensated relative to non-promoted teachers in schools without career teachers. However, the magnitudes are small and only statistically significant when we use the share of career teachers as treatment.

	(1)	(2)	(3)	(4)
	Separations	Separations	Exits	Exits
Panel A:	-	-		
At least one CT at school	-0.008**	-0.008*	-0.002	-0.002
	(0.004)	(0.004)	(0.003)	(0.003)
R^2	0.033	0.075	0.016	0.061
Ν	322,011	321,279	322,011	321,279
Panel B:				
Share CT at school	-0.078***	-0.079***	-0.005	-0.004
	(0.021)	(0.021)	(0.015)	(0.014)
R^2	0.033	0.075	0.016	0.061
Ν	322,011	321,279	322,011	321,279
Year FE:s	Yes	Yes	Yes	Yes
School FE:s	Yes	Yes	Yes	Yes
School controls	Yes	Yes	Yes	Yes
Teacher controls		Yes		Yes
Control mean	0.223	0.223	0.118	0.118

Table 3.6. Separations for non-promoted teachers in participating vs.non-participating schools

Note: In the table, we relate the change in the presence of career teachers within a school in year t to an indicator for if the teacher leaves the school or occupation between t - 1 and t. School controls are the student to teacher ratio and log number of students. Lagged teacher controls are gender, age band, experience band, tenure band, level of education, educational specialization, wage decile and whether the teacher is certified.

3.5.4 Teacher composition

Since our results suggest a reduction in teacher turnover in response to the career teacher promotions, it is interesting to also consider compositional effects. Table 3.7 shows the γ -estimates from equation 3.5 for each of our four teacher composition outcomes: the fraction of certified teachers, the fraction experienced teachers, the median years of experience among teachers per school, and teacher average compulsory school grades.

The results suggest a positive effect on all four outcomes within schools that have career teachers, but the magnitudes are fairly small: a ten percent increase in the fraction of promoted teachers is associated with an increase in the fraction of certified teachers by 0.5 percentage points; an increase in the fraction of experienced teachers by 1 percentage point or about two years of experience; and a 1.5 percent of a standard deviation increase in average teacher grades. The corresponding event study results in Appendix Figure B.2 do not show significant pre-effects.

	(1)	(2)	(3)	(4)
	Certified	Experienced	Median experience	Teacher grades
Panel A:				
At least one CT at school R^2 N	0.005*** (0.002) 0.627 17,689	0.007* (0.004) 0.468 17,689	0.361*** (0.135) 0.620 17,689	0.027*** (0.010) 0.608 17,057
Panel B:				
Share CT at school	0.060*** (0.010)	0.101*** (0.021)	2.229*** (0.670)	0.149*** (0.054)
R ² N	0.628 17,689	0.470 17,689	0.621 17,689	0.608 17,057
Year FE:s School FE:s School controls Control mean	Yes Yes Yes 0.930	Yes Yes Yes 0.888	Yes Yes Yes 13.66	Yes Yes Yes 0.587

 Table 3.7. Teacher composition

Note: In the table, we relate the effect of introducing the reform in a school in year t to the change in the share of certified teachers (col. 1), the share of experienced teachers, defined as the share with at least four years of experience (col. 2), the median level of experience at the school (col. 3), and the average standardized grades among teachers (col. 4). Teacher grades are standardized by graduation year to have mean 0 and st.d. 1. School controls are the student to teacher ratio and log number of students.
3.5.5 Student performance

Finally, we look at the impact of promotions on student performance. The fact that the reform had a substantial impact on teacher wages and wage dispersion, a negative effect on teacher separations, and led to a small increase in the share of certified teachers, experienced teachers and teachers who themselves had higher average compulsory school grades implies that the career teacher reform may also have affected student outcomes, either through teacher sorting or effort. As career teachers were tasked with improving teaching practices, to be a mentor, and to lead pedagogical development projects at their schools, the general work environment may also have improved. The reduced separations among non-promoted teachers in participating schools indicates that the reform may have had a positive impact on the motivation of non-promoted teachers. This is in line with the findings by Jackson and Bruegmann (2009) and Papay et al. (2016) that the quality of teacher peers has a positive influence on student outcomes.

As described in Section 3.2.2, we measure student performance using standardized test scores on national exams in Math, English and Swedish at different grade levels.³⁰ While we can link teachers to students using the school identifier, unfortunately we cannot match teachers to classes. However, to make the analysis more precise, we use information about the teachers' subject and level of teaching in order to associate the standardized result from the subject-specific national exams in grade 3, 6 and 9 to subject and teaching level-specific career teacher reform variables.

First, we pool the data over level *l* (grade 1–3, grade 4–6, grade 7–9) and subject *b* (Math, English, Swedish). In Figure 3.10, we plot θ_{τ} from the following dynamic model:

$$y_{stbl} = \sum_{\tau \neq -1} \theta_{\tau} D_{sbl} \mathbf{1}[\tau] + \delta_{sbl} + \lambda_{tbl} + \beta X_{st} + \varepsilon_{stbl}$$
(3.9)

where y_{stbl} is the standardized national exam result at school *s* in year *t* in subject *b* at level *l*. D_{sbl} is a dummy variable indicating whether the school has ever promoted a teacher at a particular level in a particular subject. τ denotes the time relative to when a teacher at school *s* in subject *b* at level *l* is first promoted, which occurs when τ equals 0. Observations three or more event years before treatment ($\tau \leq -3$) are grouped.³¹ The effect of the career teacher, captured in θ_{τ} , is pooled over grades and subjects. Controls included in X_{st} are log number of students and student to teacher ratio (which both vary

³⁰Note that Chetty et al. (2014) show that teachers who improve test scores improve students' high school completion, college attendance, and earnings, which supports that teachers' impact on students' test scores is a relevant outcome in this case.

³¹Because the reform was introduced in 2013 and we only have student test scores until the school year 2015/2016, τ can at most take value 2.



Figure 3.10. Dynamic response: Student performance Note: The figure plots θ-coefficients with 95%-confidence intervals from estimating eq. 3.9.

by school and year). School by subject by level fixed effects, δ_{sbl} , as well as year by subject by level fixed effects, λ_{tbl} , are also included.³²

When a school first hires a career teacher in a specific subject and level, the test scores in that subject and level increases with almost 2 percent of a standard deviation in comparison to the reference point the year prior to the career step. The effect is stable over time but only statistically significant at the 5%level in the first year of participation ($\tau = 0$). We also see that the effect two years prior to participation ($\tau = -2$) is close to zero, which supports the notion that the timing of participation is not related to lagged school characteristics, including student performance. This is also in line with the results in Sections 3.3.1 and 3.5.1. The pre-effects three or more years before a school introduces the career step is negative but not statistically significant at the 5%-level.

Next we break out the results by level or by subject. In Table 3.8 we pool the estimates by either level *l* or subject *b* and report θ_q , for q = b, l, from the following equation:

$$y_{stbl} = \theta_q CT_{stbl} + \delta_{sbl} + \lambda_{tbl} + \beta X_{st} + \varepsilon_{stbl}$$
(3.10)

 CT_{stbl} is the treatment variable. This is either an indicator for having at least one career teacher in year *t* at school *s* in specific subject *b* and level *l*, or the share of career teachers in a specific subject and level in year *t* among all the teachers in the same subject and level in t - 1 at the same school.³³

³²See Appendix Figures B.3 and B.4 for the corresponding event graphs by subject and by level. ³³See Appendix B Table B.9 for the corresponding results for each subject and level separately.

In Appendix C Table C.4 we show estimates obtained when we relate student performance to

In Panel A we present the results by teaching level. We see that participating in the reform improves test scores in grades 3 and 6 but not grade 9. In particular, having a career teacher in a specific subject in grade 3 (6) improves test scores with 1.9 (2.5) percent of a standard deviation. When exploiting the treatment intensity, we find that a ten percentage point increase in the share of career teachers in either math or Swedish in the lowest level of education increases test scores with on average 0.97 percent of a standard deviation. The corresponding number for the middle tier is 0.75 percent of a standard deviation. In Panel B, we instead present results by subject. When pooling over levels, we find positive effects for all subjects. The point estimates suggest slightly larger effects in math and Swedish than in English. In Appendix C Table C.3 we find that these results are robust to including controls for the Boosts for Mathematics and Boost for Reading reforms.³⁴

A contributing factor to the lack of effect in the highest level of compulsory schooling (grade 7–9) could be that students have subject teachers rather than classroom teachers in the highest level of education, such that exposure to career teachers becomes more fragmented than at the low or middle levels. Pedagogical development projects may also become more specialized at higher tiers. Moreover, larger effects on student outcomes for younger children is consistent with that human capital interventions have higher returns in younger ages (Cunha and Heckman 2007; Heckman 2006).

To appreciate the size of these effects, consider a school with three parallel classes in the lowest level (grade 1–3) where one of the three class teachers (teaching both Swedish and Math) in each parallel class is promoted to career teacher. The results suggest that this would increase the school's grade 3 national exam scores in Swedish and Math with around 3 percent of a standard deviation.³⁵ As a comparison, it is instructive to note that Rockoff (2004) finds that raising teacher quality by one standard deviation translates into 0.10 standard deviation increase in student test scores; Fryer (2017) finds that 300 hours of principal training, including coaching and feedback to teachers, improves test scores by 0.10 standard deviations; and Fredriksson et al. (2012) find that a reduction in class size with 5 pupils increase test scores with 0.10 standard deviations.³⁶ In this respect, our results suggest a non-negligible impact of career teachers on student performance. At the same time, these effects are relatively small, for example, compared with that girls in the lowest level had

⁽i) the share and lead share of promoted teachers, (ii) the number of promoted teachers as a fraction of all teachers, or (iii) the presence of at least one career teacher, irrespective of subject and level.

³⁴Since we only have student test scores until the school year 2015/2016, the Teachers' Salary Boost is not relevant as it was only introduced in 2016.

³⁵In 2016, the share of career teachers teaching Swedish or Math in the lower tier at participating schools was 24 or 25 percent, respectively.

³⁶Our results also resonate with the finding of Jackson and Bruegmann (2009) that about 20 percent of the teacher effectiveness is due to the influence from teachers' peers during the previous three years.

	(1)	(2)	(3)
Panel A: Pool across subject	Grade 3	Grade 6	Grade 9
Treat	0.019*	0.025***	0.005
	(0.010)	(0.008)	(0.008)
R^2	0.474	0.590	0.702
Share	0.097***	0.075***	0.027
	(0.028)	(0.018)	(0.020)
R^2	0.474	0.591	0.702
Ν	25,207	31,702	15,800
Year \times subject FE:s	Yes	Yes	Yes
School \times subject FE:s	Yes	Yes	Yes
School controls	Yes	Yes	Yes
Panel B: Pool across level	Math	English	Swedish
Treat	0.018**	0.015*	0.021***
	(0.008)	(0.008)	(0.007)
R^2	0.532	0.627	0.561
Share	0.084***	0.039**	0.076***
	(0.017)	(0.018)	(0.017)
R^2	0.533	0.627	0.561
N	28,428	15,831	28,450
Year \times level FE:s	Yes	Yes	Yes
School \times level FE:s	Yes	Yes	Yes
School controls	Yes	Yes	Yes

 Table 3.8. Student performance

Note: The table presents the results of estimating eq. 3.10. In Panel A we pool the results across subjects. In Panel B we pool the results across levels. Treatment variables are "treat" (a dummy equal to one if there is at least one subject & level career teacher at the school) or "share" (the number of subject & level career teachers at the school in year *t* divided by the number subject & level teachers in the school in t - 1). School controls are log number of students and student to teacher ratio. The results of the national exams are standardized by year, subject and level to have mean 0 and st.d. 1. Each school obtains a mean standardized score. Standard errors are clustered at school district level and reported in parenthesis.

0.10 standard deviations higher test scores in 2015 than boys did. Appendix Table B.10 shows that, unlike the teacher turnover response, which appeared to be stronger in school districts where the career teachers have more teaching development responsibilities, the impact on student outcomes is about the same irrespective of the type of career teacher.

3.6 Conclusion

Despite the widespread interest in the determinants of student outcomes among policy-makers and researchers, evidence on how policies aimed at improving the teaching pool impacts teachers and students remains scarce. One likely reason is the rigidity of the teacher labor market in many countries, which often prevents large-scale interventions. We contribute to this gap in the literature by analyzing the impact of a Swedish reform which introduced a new career step for teachers starting in 2013. The reform allowed schools to promote talented teachers to a new title ("career teacher") with a substantial associated wage increase financed entirely through ear-marked state funding. The reform intended to reward talented teachers, to increase the attractiveness of the profession through higher wage dispersion, and to take advantage of teachers' professional competence. While the allocation of the number of promotions across school districts was based on the number of students, the school districts had discretion over the allocation of promotions across schools and teachers within each district.

The paper provides evidence on the response to this reform, both in terms of the teachers selected for promotion and the impact it had on the wage structure, teacher separations and student performance. Our estimates capture the overall impact of the promotion program, which entails both higher pay and increased responsibilities for planning the pedagogical work and coaching other teachers. We show that the allocation of teacher promotions across schools is related to school size but unrelated to other pre-determined school characteristics such as teacher turnover rates. Within schools, high-wage teachers were more likely to be promoted. Our interpretation is that principals complied with the intentions of the reform and promoted the most talented teachers.

The reform induced significant changes in teacher pay: the stipulated wage increase had full pass-trough onto promoted teachers' wages and led to an increase in wage dispersion both within and across schools. This wage response is substantially larger than the documented impacts of just introducing more discretion over teacher pay, both in the US (Biasi 2018) and in Sweden (Willén 2019). The ear-marked funding and increased responsibilities for career teachers (making differential pay seem more acceptable) are probably key components behind the minimal crowd-out in our setting, highlighting the importance of policy-design to get the desired impact on the wage dispersion. Compared to schools that did not introduce the career step, we find

that the promotion program led to a reduction in teacher separations and small changes in the teaching pool, despite very similar trajectories before the reform. These responses appear driven by school districts using their career teachers for teaching development responsibilities rather than coaching and mentoring tasks, suggesting that the pure salary increase cannot fully explain the turnover patterns.

It is important to highlight that our estimates capture relative differences in teacher turnover (and other outcomes) across schools, not aggregate effects on school quality. Nevertheless, the fact that we find a reduction in the fraction of teachers exiting the teaching profession in schools with career teachers suggests that promotions can incentivize teachers to stay in the profession. We also find non-negligible effects on student test scores in Math, English and Swedish in grades 3 and 6. Our results suggest that promoting one third of grade 1–3 teachers increases test scores at the grade 3 national tests by 3 percent of a standard deviation. As a comparison, previous studies suggest that these effects are about a third of the size of improving teacher quality by one standard deviation. Together, our results lend support to that performance-based promotions could be an important tool for raising school quality.

References

- Alva, C., M. Bobba, G. Leon, C. Neilson, and M. Nieddu (2017). "Teacher Wages, Student Achievement and the Recruitment of Talent in Rural Peru". Unpublished.
- Bacolod, M.P. (2007). "Do Alternative Opportunities Matter? The Role of Female Labor Markets in the Decline of Teacher Quality". *The Review of Economics and Statistics* 89.4, pp. 737–751.
- Biasi, B. (2018). *The Labor Market for Teachers Under Different Pay Schemes*. Tech. rep. 24813. National Bureau of Economic Research.
- Britton, J. and C. Popper (2016). "Teacher Pay and School Productivity: Exploiting Wage Regulation". *Journal of Public Economics* 133, pp. 75–89.
- Cabrera, J.M and D. Webbink (2018). "Do Higher Salaries Yield Better Teachers and Better Student Outcomes?" *MPRA Paper No.* 86972.
- Cantrell, S., J. Fullerton, T.J. Kane, and D.O. Staiger (2008). *National Board Certification and Teacher Effectiveness: Evidence from a Random Assignment Experiment*. Tech. rep. 14608.
- Card, D., A. Mas, E. Moretti, and E. Saez (2012). "Inequality at Work: The Effect of Peer Salaries on Job Satisfaction". *American Economic Review* 102.6, pp. 2981–3003.
- Chetty, R., J.N. Friedman, and J.E. Rockoff (2014). "Measuring the Impacts of Teachers II: Teacher Value-Added and Student Outcomes in Adulthood". *American Economic Review* 104.9, pp. 2633–2679.
- Clotfelter, C., E. Glennie, H. Ladd, and J. Vigdor (2008). "Would Higher Salaries Keep Teachers in High-Poverty Schools? Evidence From a Policy Intervention in North Carolina". *Journal of Public Economics* 92, pp. 1352– 1370.
- Corcoran, S.P., W.N. Evans, and R.M. Schwab (2004). "Changing Labor-Market Opportunities for Women and the Quality of Teachers, 1957-2000". American Economic Review 94.2, pp. 230–235.
- Cunha, F. and J. Heckman (May 2007). "The Technology of Skill Formation". *American Economic Review* 97.2, pp. 31–47.
- Dee, Thomas S. and James Wyckoff (2015). "Incentives, Selection, and Teacher Performance: Evidence from IMPACT". Journal of Policy Analysis and Management 34.2, pp. 267–297.
- Deshpande, M. and Y. Li (2019). "Who Is Screened Out? Application Costs and the Targeting of Disability Programs". *American Economics Journal: Economic Policy* 11.4, pp. 213–248.

- Dolton, P. and O.D. Marcenaro-Gutierrez (2011). "If You Pay Peanuts Do You Get Monkeys? A Cross-Country Analysis of Teacher Pay and Pupil Performance". *Economic Policy* 26.65, pp. 5–55.
- Dube, A., L. Giuliano, and J. Leonard (2019). "Fairness and Frictions: The Impact of Unequal Raises on Quit Behavior". *American Economic Review* 109.2, pp. 620–63.
- Falch, T. (2011). "Teacher Mobility Responses to Wage Changes: Evidence from a Quasi-natural Experiment". American Economic Review: Papers & Proceedings 101.3, pp. 460–465.
- Figlio, D.N. (1997). "Teacher Salaries and Teacher Quality". *Economics Letters* 55.2, pp. 267–271.
- Fredriksson, P. and B. Öckert (2007). "The Supply of Skills to the Teacher Profession". Unpublished manuscript.
- Fredriksson, P., B. Öckert, and H. Oosterbeek (2012). "Long-Term Effects of Class Size". *The Quarterly Journal of Economics* 128.1, pp. 249–285.
- Fryer, R.G. (2017). *Management and Student Achievement: Evidence from a Randomized Field Experiment*. Tech. rep. National Bureau of Economic Research.
- Gilpin, G.A. (2012). "Teacher Salaries and Teacher Aptitude: An Analysis Using Quantile Regressions". *Economics of Education Review* 31.3, pp. 15–29.
- Grönqvist, E., L. Hensvik, and A. Thoresson (2020). *Teacher Career Opportunities and School Quality*. Tech. rep. 2020:2. IFAU.
- Grönqvist, E. and J. Vlachos (2016). "One Size Fits All? The Effects of Teacher Cognitive and Non-cognitive Abilities on Student Achievement". *Labour Economics* 42, pp. 138–150.
- Hanushek, E.A. and S.G. Rivkin (2006). "Teacher Quality". *Handbook of the Economics of Education* 2, pp. 1051–1078.
- Heckman, J.J. (2006). "Skill Formation and the Economics of Investing in Disadvantaged Children". *Science* 312.5782, pp. 1900–1902.
- Hendricks, M. (2014). "Does It Pay to Pay Teachers More? Evidence from Texas". *Journal of Public Economics* 109, pp. 50–63.
- Jackson, C.K and E. Bruegmann (2009). "Teaching Students and Teaching Each Other: The Importance of Peer Learning for Teachers". *American Economic Journal: Applied Economics* 1.4, pp. 85–108.
- Jackson, C.K. (2010). "A Little Now for a Lot Later A Look at a Texas Advanced Placement Incentive Program". *Journal of Human Resources* 45.3, pp. 591–639.
- (2012). "Recruiting, Retaining and Creating Quality Teachers". Nordic Economic Policy Review 3.1, pp. 61–103.
- Jackson, C.K., J. Rockoff, and D. Staiger (2014). "Teacher Effects and Teacher-Related Policies". *Annual Review of Economics* 6.1, pp. 801–825.

- Jacob, B.A., J.E. Rockoff, E.S. Taylor, B. Lindy, and R. Rosen (2018). "Teacher Applicant Hiring and Teacher Performance: Evidence from DC public schools". *Journal of Public Economics* 166, pp. 81–97.
- Lazear, E.P and K.L. Shaw (2007). "Personnel Economics: The Economist's View of Human Resources". *Journal of Economic Perspectives* 21.4, pp. 91–114.
- Lazear, E.P. and S. Rosen (1981). "Rank-Order Tournaments as Optimum Labor Contracts". *Journal of Political Economy* 89.5, pp. 841–864.
- Leigh, A. (2010). "Estimating Teacher Effectiveness from Two-Year Changes in Students' Test Scores". *Economics of Education Review* 29.3, pp. 480– 488.
- (2012). "Teacher Pay and Teacher Aptitude". *Economics of Education Review* 31.3, pp. 41–53.
- Leigh, A. and C.A. Ryan (2008). "How and Why Has Teacher Quality Changed in Australia?" *The Australian Economic Review* 41.2, pp. 141–159.
- Li, X. (2019). "The Costs of Workplace Favoritism: Evidence from Promotions in Chinese High Schools". Unpublished manuscript.
- Loeb, S. and M.E. Page (2000). "Examining the Link Between Teacher Wages and Student Outcomes: The Importance of Alternative Labor Market Opportunities and Non-Pecuniary Variation". *The Review of Economics and Statistics* 82.3, pp. 393–408.
- Neal, D. (2011). "Chapter 6 The Design of Performance Pay in Education". *Handbook of the Economics of Education*. Ed. by E.A. Hanushek, S. Machin, and L. Woessmann. Vol. 4. Handbook of The Economics of Education. Elsevier, pp. 495–550.
- Oyer, P and S. Schaefer (2011). "Personnel Economics: Hiring and Incentives". *Handbook of Labor Economics* 4.B, pp. 1769–1823.
- Papay, J.P., E.S. Taylor, J.H. Tyler, and M. Laski (2016). Learning Job Skills from Colleagues at Work: Evidence from a Field Experiment Using Teacher Performance Data. Tech. rep. 21986. National Bureau of Economic Research.
- Ree, J. de, M. Muralidharan, M. Pradhan, and H. Rogers (2018). "Double for Nothing? Experimental Evidence on an Unconditional Teacher Salary Increase in Indonesia". *The Quarterly Journal of Economics* 133.2, pp. 993– 1039.
- Regeringen (2013a). Förordning (2013:70) om statsbidrag till skolhuvudmän som inrättar karriärsteg för lärare (Ordinance (2013: 70) on state grants for school districts who establish career steps for teachers).
- (2013b). Karriärvägar för lärare i skolväsendet m.m. (Career paths for teachers in the school system, etc.) Regeringens proposition 2012/13:136.
- (2013c). Karriärvägar m.m. i fråga om lärare i skolväsendet (Career paths etc. in the case of teachers in the school system). Utbildningsdepartementets promemoria U2012/4904/S.

- Rivkin, S.G., E.A. Hanushek, and J.F. Kain (2005). "Teachers, Schools, and Academic Achievement". *Econometrica* 73.2, pp. 417–458.
- Rockoff, J.E. (2004). "The Impact of Individual Teachers on Student Achievement: Evidence from Panel Data". *American Economic Review* 94.2, pp. 247– 252.
- Rockoff, J.E., B.A. Jacob, T.J. Kane, and D.O. Staiger (2011). "Can You Recognize an Effective Teacher When You Recruit One?" *Education Finance and Policy* 6.1, pp. 43–74.
- Rockoff, J.E. and C. Speroni (2011). "Subjective and Objective Evaluations of Teacher Effectiveness: Evidence from New York City". *Labour Economics* 18.5, pp. 687–696.
- Skolinspektionen (2018). Samverkan för en likvärdig bedömning: Ombedömning av nationella prov 2015 (Collaboration for an equivalent assessment: Reassessment of national tests 2015).
- Skolverket (2014). Skolverkets årsredovisning 2014 (The Swedish National Agency for Education Annual Report 2014).
- (2016). Skolverkets årsredovisning 2016 (The Swedish National Agency for Education Annual Report 2016).
- (2018a). Antal elever och skolenheter inkl fritidshem (Number of students and school units including youth centers). URL: https://www.skolverket.
 se / skolutveckling / statistik / sok - statistik - om - forskola skola-och-vuxenutbildning (visited on 07/12/2018).
- (2018b). Skolverkets allmänna råd med bedömning: Betyg och betygssätting (The Swedish National Agency for Education's general advice with assessment: Grades and grading). URL: https://www.skolverket.se/publikationsserier/allmanna-rad/2018/allmanna-rad-ombetyg-och-betygssattning (visited on 06/10/2020).
- Söderström, M. (2010). "Wage Scales and Centralized Bargaining A Binding Constraint on the Wage-Setting?" *Applied Economics Letters* 17.3, pp. 247– 250.
- Sojourner, A.J., E. Mykerezi, and K.L. West (2014). "Teacher Pay Reform and Productivity Panel Data Evidence from Adoptions of Q-Comp in Minnesota". *Journal of Human Resources* 49.4, pp. 945–981.
- Statskontoret (2015). Uppföljning av karriärstegsreformen. Delrapport 1 (Followup of the career step reform. Interim report 1).
- (2017). Uppföljning av karriärstegsreformen. Delrapport 3 (Follow-up of the career step reform. Interim report 3).
- Willén, A.L.P. (2019). *Decentralization of Wage Determination*. Tech. rep. 2019:17. IFAU.

Appendix A: Supplementary description



Figure A.1. Allocation of promotions across school districts - all school districts

Note: Each cross corresponds to one school district. The solid line is a linear prediction. The dashed line is the 45 degree line.



Figure A.2. Allocation of promotions within school districts

Note: Each cross corresponds to one school. The solid line is a linear prediction. The dashed line is the 45 degree line.



Figure A.3. Career teacher tasks: component loadings

Note: The figure shows the loadings of the different career teacher tasks displayed in Figure 3.2 on the two components. Career teacher tasks are defined as follows: CTtask1="Coaching other teachers colleagues", CTtask2="Initiate pedagogical discussions", CTtask3="Mentor teacher students", CTtask4="Support teaching methods", CTtask5="Engage in research", CTtask6="Subject development", CTtask7="Subject didactic", CTtask8="Mentor new hires", CTtask9="Select teaching materials". Component 1 is "Coaching and mentoring" while component 2 is "Teaching development".

	• • • •
	(1)
	Log wages
Female	-0.007***
	(0.001)
Age:	
30-39	0.032***
	(0.001)
40-49	0.049***
	(0.002)
50-59	0.064***
	(0.002)
60 and over	0.079***
	(0.003)
Years of experience:	
4-9	0.042***
	(0.001)
10-14	0.081***
	(0.002)
15-24	0.116***
	(0.002)
25 and over	0.167***
	(0.003)
Years of tenure:	
2-5	-0.006***
	(0.001)
5-10	-0.004***
	(0.001)
10-15	0.001
	(0.002)
Level of education:	
Upper secondary	0.025***
	(0.009)
Post-secondary	0.073***
	(0.009)
Doctoral	0.096***
	(0.010)
Certification:	
Certified	0.026***
	(0.002)
Type of contract:	
Permanent	0.033***
	(0.001)

 Table A.1. Determinants of teacher pay

	(1)
	Log wages
Educational specialization:	
Swedish & social sciences	-0.009***
	(0.001)
Languages	0.017***
	(0.001)
Vocational	-0.006***
	(0.002)
Other teaching	0.009***
	(0.002)
Non-teaching	-0.005**
	(0.002)
Constant	9.932***
	(0.009)
R^2	0.676
Ν	167,604
Year FE:s	Yes
School District FE:s	Yes
Control mean	10.22

 Table A.1. Determinants of teacher pay – continued from previous page

Note: The table shows the correlation between teacher observable characteristics and the monthly full-time wage. The period is restricted to the pre-reform period, 2010 to 2012. Omitted categories are "Under 30" (Age), "Under 4" (Experience), "Under 2 (Tenure), "Compulsory" (Level of education) and "Math and natural sciences" (Educational specialization). Educational specialization is inferred from the field of education according to the Swedish SUN classification. "Non-teaching" includes everyone without pedagogical specialization.

	١	No CT	At le	ast one CT
	Mean	sd	Mean	sd
Year: 2013				
Nr career teachers	0.00	(0.00)	1.37	(0.68)
Nr teachers t-1	15.76	(11.26)	24.70	(13.03)
Nr students t-1	204.92	(140.32)	323.39	(169.24)
Student teacher ratio t-1	13.65	(3.27)	13.56	(3.04)
Share separate t-1	0.22	(0.15)	0.22	(0.12)
Share exit t-1	0.11	(0.10)	0.11	(0.08)
Ν	1,911		1,039	
Year: 2014				
Nr career teachers	0.00	(0.00)	2.23	(1.49)
Nr teachers t-1	10.64	(8.35)	18.31	(11.45)
Nr students t-1	145.42	(111.07)	243.70	(147.15)
Student teacher ratio t-1	14.36	(4.41)	13.95	(3.80)
Share separate t-1	0.23	(0.17)	0.23	(0.15)
Share exit t-1	0.11	(0.12)	0.11	(0.10)
Ν	673		1,238	× /
Year: 2015				
Nr career teachers	0.00	(0.00)	1.56	(0.88)
Nr teachers t-1	8.89	(7.04)	13.48	(9.01)
Nr students t-1	125.29	(104.26)	187.20	(121.69)
Student teacher ratio t-1	14.64	(4.43)	14.53	(3.77)
Share separate t-1	0.24	(0.18)	0.23	(0.15)
Share exit t-1	0.11	(0.13)	0.11	(0.11)
Ν	418		255	
Year: 2016				
Nr career teachers	0.00	(0.00)	1.40	(0.70)
Nr teachers t-1	7.96	(6.36)	11.41	(7.17)
Nr students t-1	117.74	(109.16)	161.63	(95.47)
Student teacher ratio t-1	14.89	(4.37)	14.81	(4.04)
Share separate t-1	0.26	(0.20)	0.24	(0.17)
Share exit t-1	0.12	(0.14)	0.11	(0.11)
Ν	314		104	

 Table A.2. School summary statistics

Note: In each year, we only include schools that have not (never or yet) participated as well as schools that participate for the first time.

	In surve	ey or not	By caree	er teacher design
	Not in survey	In sur- vey	Coachin and men- toring	g Teaching devel- op- ment
Mean teacher wage	10.28	10.28	10.29	10.28
Separation rate	0.23	0.24	0.24	0.23
Exit rate	0.11	0.12	0.11	0.11
Certified	0.92	0.91	0.91	0.91
Experienced	0.88	0.87	0.87	0.87
Teacher 9th grade GPA	0.56	0.56	0.56	0.57
Urban	0.40	0.34	0.38	0.27
Nr students per school	232	235	252	230
Nr teachers per school	18.24	18.17	19.04	17.99
Student-to-teacher ratio per school	13.55	13.61	14.04	13.62
Nr career teachers per school	1.17	1.17	1.29	1.11
N	198	92	48	49

Table A.3. School district summary statistics

Note: The table shows district summary statistics. Column 1 and 2 show districts broken down by whether they are included in the survey sample or not. Column 3 and 4 show summary statistics separately for districts with a coaching and mentoring CT design and a teaching development design. The construction of these groups is described in Section 3.5.3. Experienced is the fraction of teachers with at least four years of teaching experience.

Appendix B: Additional results



Figure B.1. Distribution of correlation coefficients between share career teachers and share students

Note: The figure plots the distribution of correlation coefficients between the share of career teachers and share of students within school districts. The two lines mark the 50th and 75th percentiles. A correlation coefficient closer to 1 identifies school districts that appear to allocate career teaching positions according to the same rule used to allocate positions across school districts on the national level. Only school districts that at some point participate in the reform and that have more than one school are included. The figure is based on 280 school districts.



Figure B.2. Dynamic responses: teacher composition

Note: The figure plots γ -coefficients with 95%-confidence intervals from estimating eq. 3.7. It shows the evolution of the share of certified teachers, the share of experienced teachers, the median level of experience at the school, and the average grades among teachers within schools before and after promoting at least one career teacher. $\tau = -1$ is omitted.



Figure B.3. Dynamic responses: Student performance – pool across subject

Note: The figure plots the θ -coefficients with 95%-confidence intervals from estimating eq. 3.9, where we only pool across subject.



Figure B.4. Dynamic responses: Student performance – pool across level

Note: The figure plots the θ -coefficients with 95%-confidence intervals from estimating eq. 3.9, where we only pool across level.

		Share C	Γ at school	
	2013	2014	2015	2016
School characteristics $t - 1$:				
Log nr students	0.001	0.008**	0.000	-0.005
	(0.002)	(0.004)	(0.006)	(0.009)
Student-to-teacher ratio	0.000	-0.000	0.000	0.001
	(0.001)	(0.001)	(0.001)	(0.001)
Separation rate	-0.000	-0.022	0.014	0.009
	(0.010)	(0.022)	(0.037)	(0.048)
Exit rate	-0.003	0.039	0.021	-0.046
	(0.015)	(0.027)	(0.045)	(0.045)
Certified (share)	0.014	0.036	0.078*	-0.029
	(0.015)	(0.034)	(0.044)	(0.041)
Female (share)	0.006	0.022	-0.030	-0.006
	(0.008)	(0.023)	(0.034)	(0.034)
Mean age (years)	-0.001*	-0.001	-0.001	0.001
	(0.001)	(0.001)	(0.001)	(0.002)
Mean experience (years)	0.000	0.000	0.001	-0.001
	(0.001)	(0.001)	(0.002)	(0.001)
Math/natural science (share)	-0.008	-0.003	-0.056*	0.018
	(0.010)	(0.027)	(0.030)	(0.034)
Swedish/social science (share)	0.004	0.024	-0.010	0.068**
	(0.009)	(0.020)	(0.032)	(0.029)
District FE	Yes	Yes	Yes	Yes
R^2	0.183	0.253	0.404	0.458
Ν	2,941	1,882	611	346
Mean dep. var.	.02	.09	.05	.04

 Table B.1. Factors that predict selection of schools – Share CT

Note: This presents the results of regressions of $CT_{sdt} = \phi_{dt} + \beta_t X_{sdt-1} + \varepsilon_{sdt}$ where CT_{sdt} is a variable equal to the share of career teachers at the school in year *t*. Variables are measured at school level. Regressions are estimated separately by year and only include schools that have not (never or yet) participated as well as schools that participate for the first time. Standard errors are clustered at school district level and reported in parenthesis.

7	¢							
		At least one	CT at school			Share C	T at school	
	2013	2014	2015	2016	2013	2014	2015	2016
Panel A: 3rd grade test results								
Log nr students	0.228^{***}	0.302^{***}	0.234^{***}	0.144^{***}	0.001	0.008*	0.002	-0.009
1	(0.021)	(0.018)	(0.046)	(0.046)	(0.002)	(0.004)	(0.007)	(0.00)
Student-to-teacher ratio	-0.009***	-0.010^{**}	0.000	-0.005	0.000	-0.000	0.001	0.001
	(0.003)	(0.003)	(0.007)	(0.006)	(0.001)	(0.001)	(0.001)	(0.001)
Separation rate	0.00	-0.068	0.003	0.252	0.004	-0.024	0.005	0.022
	(0.088)	(0.095)	(0.217)	(0.222)	(0.010)	(0.023)	(0.044)	(0.054)
Exit rate	-0.084	0.213	0.051	-0.312	-0.008	0.054^{*}	0.026	-0.059
	(0.110)	(0.133)	(0.232)	(0.230)	(0.015)	(0.029)	(0.048)	(0.052)
Certified (share)	-0.109	0.192	0.134	-0.308	0.003	0.019	0.076^{*}	-0.039
	(0.172)	(0.159)	(0.249)	(0.214)	(0.017)	(0.041)	(0.046)	(0.042)
Female (share)	-0.029	0.270^{**}	-0.129	0.047	0.009	0.047*	-0.028	-0.027
	(0.087)	(0.106)	(0.172)	(0.186)	(0.00)	(0.025)	(0.039)	(0.036)
Mean age (years)	-0.007	-0.006	-0.002	0.011	-0.001	-0.001	-0.001	0.001
	(0.005)	(0.005)	(0.007)	(0.008)	(0.001)	(0.001)	(0.001)	(0.002)
Mean experience (years)	0.003	0.001	0.002	-0.010	0.000	0.000	0.001	-0.002
	(0.005)	(0.005)	(0.008)	(0.008)	(0.001)	(0.001)	(0.002)	(0.002)
Math/natural science (share)	0.068	-0.072	-0.244	-0.083	-0.007	0.003	-0.068**	0.003
	(0.081)	(0.115)	(0.153)	(0.170)	(0.011)	(0.029)	(0.033)	(0.034)
Swedish/soc. science (share)	0.133^{**}	-0.009	-0.109	0.205	0.00	0.041^{*}	-0.015	0.063^{*}
	(0.066)	(0.091)	(0.145)	(0.161)	(0.010)	(0.021)	(0.036)	(0.032)

 Table B.2. Factors that predict selection of schools – with student test scores

Table B.2. Factors that predict s	election of sc	hools – with st	udent test sco	res – continu	ed from previ	ous page		
		At least one	CT at school			Share C	T at school	
	2013	2014	2015	2016	2013	2014	2015	2016
3rd grade math score	-0.008	-0.047	0.020	0.081	-0.000	-0.002	0.012	0.009
	(0.021)	(0.029)	(0.043)	(0.069)	(0.003)	(0.006)	(0.00)	(0.014)
3rd grade Swedish score	0.013	0.045	-0.039	-0.029	0.006	0.012^{*}	-0.017	0.013
	(0.031)	(0.029)	(0.060)	(0.076)	(0.004)	(0.006)	(0.013)	(0.015)
District FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
R^2	0.360	0.415	0.481	0.586	0.194	0.272	0.417	0.489
N	2,509	1,635	543	307	2,509	1,635	543	307
Mean dep. var.	.34	.63	.37	.24	.02	60.	.05	.04
Panel B: 6th grade test results								
Log nr students	0.210^{***}	0.298^{***}	0.255^{***}	0.103*	-0.001	0.010^{**}	0.007	-0.007
	(0.022)	(0.019)	(0.051)	(0.052)	(0.002)	(0.005)	(0.008)	(0.012)
Student-to-teacher ratio	-0.009**	-0.008***	-0.002	-0.008	0.000	0.000	0.000	0.000
	(0.004)	(0.003)	(0.008)	(0.008)	(0.001)	(0.001)	(0.001)	(0.001)
Separation rate	0.072	-0.197	0.329	0.551^{**}	0.001	-0.044	0.070	0.066
	(0.081)	(0.120)	(0.241)	(0.233)	(0.011)	(0.029)	(0.046)	(0.052)
Exit rate	-0.107	0.366^{**}	-0.484	-0.615**	-0.010	0.074^{**}	-0.091	-0.119***
	(0.125)	(0.156)	(0.298)	(0.254)	(0.016)	(0.034)	(0.056)	(0.044)
Certified (share)	-0.031	0.181	0.020	-0.452*	0.003	0.030	0.034	-0.081
	(0.162)	(0.166)	(0.292)	(0.248)	(0.016)	(0.041)	(0.058)	(0.053)

		At least on	e CT at school			Share C	T at school	
	2013	2014	2015	2016	2013	2014	2015	2016
Female (share)	-0.024	0.127	-0.207	0.093	0.002	0.042	-0.042	-0.007
	(0.087)	(0.120)	(0.189)	(0.202)	(0.010)	(0.028)	(0.040)	(0.037)
Mean age (years)	-0.011^{**}	-0.007	0.005	0.018^{*}	-0.001^{**}	-0.002*	0.001	0.004^{**}
	(0.005)	(0.006)	(0.008)	(0.00)	(0.001)	(0.001)	(0.002)	(0.002)
Mean experience (years)	0.004	0.001	0.009	-0.007	0.000	0.001	0.002	-0.002
	(0.005)	(0.006)	(0.009)	(0.010)	(0.001)	(0.001)	(0.002)	(0.002)
Math/natural science (share)	-0.091	-0.142	-0.247	-0.182	-0.020	-0.023	-0.047	-0.019
	(0.101)	(0.132)	(0.197)	(0.260)	(0.015)	(0.034)	(0.040)	(0.046)
Swedish/soc. science (share)	0.140^{*}	-0.055	0.333*	-0.054	0.006	0.029	0.096^{**}	0.036
	(0.078)	(0.119)	(0.177)	(0.216)	(0.010)	(0.026)	(0.041)	(0.042)
6th grade math score	-0.035	-0.001	0.122	-0.025	-0.005	0.009	0.011	0.002
	(0.033)	(0.047)	(0.074)	(0.068)	(0.004)	(0.010)	(0.013)	(0.015)
6th grade English score	0.007	0.006	-0.072	0.001	-0.002	-0.004	-0.008	-0.003
	(0.036)	(0.046)	(0.079)	(0.074)	(0.005)	(0.010)	(0.015)	(0.017)
6th grade Swedish score	0.015	-0.028	-0.117	0.046	0.006	-0.009	-0.006	0.001
	(0.029)	(0.047)	(0.084)	(0.085)	(0.004)	(0.010)	(0.015)	(0.019)
District FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
R^2	0.397	0.450	0.585	0.629	0.217	0.287	0.504	0.555
Z	2,041	1,280	390	206	2,041	1,280	390	206
Mean dep. var.	.38	.65	.40	.23	.03	60.	.05	.03
Note: This presents results of estim	nating $CT_{sdt} =$	$\phi_{dt} + eta_t X_{sdt-1}$	+ ε_{sdt} where C7	sdt is equal to 1	l if the school pa	urticipates in the	e reform in yea	<i>t</i> (columns 1–4),
or the share of career teachers at the	ne school in ye: (rever or vet) r	ar t (columns 5	-8). Variables :	are measured at	t school level. R	egressions are	estimated separ	ately by year and
OILLY INCLUDE SCHOOLS UTAL ITAVE HOLY	(liever ut yet)	varucipateu anu	i sciiouis uiat pa	nucipate tot ut	c IIISt UIIIC. Stall	ualu cituis alc	ciusicica uy sci	noor district.

	(1)	(2)	(3)
Female	0.036***	0.024***	0.010
	(0.004)	(0.004)	(0.008)
Permanent contract		0.024***	0.035***
		(0.005)	(0.008)
Certified		0.041***	0.058***
		(0.006)	(0.011)
Age			
30-39	0.029***	-0.003	-0.017*
	(0.008)	(0.008)	(0.010)
40-49	-0.003	-0.048***	-0.046***
	(0.008)	(0.009)	(0.015)
50-59	-0.109***	-0.156***	
	(0.009)	(0.009)	
60+	-0.208***	-0.264***	
	(0.009)	(0.010)	
Experience			
4-9	0.105***	0.033***	0.011
	(0.006)	(0.006)	(0.009)
10-14	0.174***	0.038***	-0.000
	(0.006)	(0.007)	(0.014)
15-24	0.178***	0.010	-0.036
	(0.007)	(0.010)	(0.024)
25+	0.181***	-0.030***	
	(0.007)	(0.010)	
Tenure			
2-5		0.026***	0.039***
		(0.005)	(0.010)
5-10		0.048***	0.071***
		(0.006)	(0.011)
10-15		0.054***	0.090***
		(0.006)	(0.012)
Level of education			
Upper secondary		-0.051***	-0.065
		(0.017)	(0.045)
Post-secondary		-0.079***	-0.082*
,		(0.018)	(0.048)
Doctoral		-0.125***	-0.018
		(0.030)	(0.125)
Educational specialization			
Swedish and social sciences		-0.022***	-0.015
		(0.007)	(0.010)
Languages		-0.039***	-0.042***
		(0.008)	(0.015)
Vocational		-0.103***	-0.106***
		(0.006)	(0.014)
Other teaching		-0.070***	-0.050***
e		(0.005)	(0.009) 14
			10

 Table B.3. Selection of teachers for promotion

	(1)	(2)	(3)
Non-teaching		-0.072***	-0.062***
e		(0.008)	(0.015)
School wage decile			. ,
2nd wage decile		0.010*	0.014*
		(0.005)	(0.009)
3rd wage decile		0.039***	0.049***
·		(0.007)	(0.011)
4th wage decile		0.070***	0.083***
-		(0.007)	(0.012)
5th wage decile		0.102***	0.125***
		(0.007)	(0.013)
6th wage decile		0.137***	0.161***
		(0.008)	(0.015)
7th wage decile		0.171***	0.220***
		(0.010)	(0.021)
8th wage decile		0.207***	0.291***
		(0.009)	(0.024)
9th wage decile		0.230***	0.289***
		(0.011)	(0.027)
10th wage decile		0.264***	0.374***
		(0.010)	(0.036)
Teacher 9th grade GPA			
-1 to -0.5			-0.015
			(0.028)
-0.5 to 0			-0.021
			(0.026)
0 to 0.5			-0.005
			(0.026)
0.5 to 1			0.001
			(0.025)
1+			0.013
			(0.024)
Constant	0.026***	0.108***	0.082*
	(0.007)	(0.021)	(0.044)
School FE	Yes	Yes	Yes
R^2	0.101	0.142	0.240
Ν	55,785	55,658	17,881

Table B.3. Selection of teachers for promotion – continued from previous page

Note: This shows the results of estimating $EverCT_{ist} = \beta X_{ist} + \delta_s + \varepsilon_{ist}$, using data only from 2012. Omitted categories are "Under 30" (Age), "Under 4" (Experience), "Under 2 (Tenure), "Compulsory" (Level of education), "Math and natural sciences" (Educational specialization), "1" (Wage decile), and "Less than -1" (Teacher GPA). "Non-teaching" includes everyone without pedagogical specialization. $EverCT_{ist}$ is an indicator equal to 1 if the teacher is ever a career teacher. As we only have data on teacher's 9th grade GPA from 1988 onward, the sample size that includes this variable is smaller than in the other specifications. Standard errors are clustered at school district level and included in parenthesis.

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Sample	Full	Full	Full	Full	Full	Full	First time	First time
							CI	CI
Panel A: ln(wage)								
Promoted	0.212^{***}	0.210^{***}	0.179^{***}	0.179^{***}	0.179^{***}	0.149^{***}	0.141^{***}	
	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	(0.002)	
2012 mean wage	10.22							
R^{2}	0.404	0.512	0.800	0.796	0.807	966.0	0.904	
Panel B: Monthly wage (SEK)								
Promoted	7176.4^{***}	7104.4^{***}	6253.5***	6231.7^{***}	6243.3***	5329.2***	4744.3***	5315.1^{***}
	(90.5)	(84.9)	(73.2)	(74.4)	(73.9)	(65.4)	(58.6)	(56.6)
2012 mean wage	27501.3							
R^2	0.438	0.542	0.799	0.795	0.808	0.954	0.900	0.949
Year FE	Yes	Yes	Yes		Yes	Yes		Yes
School FE		Yes	Yes	Yes	Yes			
Individual FE						Yes		Yes
Year \times district FE					Yes			
Linear & quadratic time trends				Yes				
Controls			Yes	Yes	Yes	Yes	Yes incl.	Yes
							lag wage	
Z	395,960	395,960	395,083	395,083	395,083	374,108	260,619	360,260
<i>Note:</i> This provides the results of e contract, age (in five age bands), lev tenure (in five bands). Standard erro	stimating $y_{ist} = y_{el}$ of education ors are clustered	: $\alpha_i + \delta_s + \lambda_r + (in four categories at school distribution)$	$\theta CT_{ist} + \beta X_{ist}$ pries), education (ct level and inc	$+ \varepsilon_{ixt}$. Contro nal specializati sluded in paren	ls are dummies on (in one of si thesis. In speci	s for female, te ix categories), (fication (7) and	acher certificat experience (in 1 1 (8) the sample	ion, permanent ive bands) and is censored to
only include the first year of becomi	ng a career teach	her as well as th	lose that are not	(yet) promoted				

*	0 0 1 1		
	(1)	(2)	(3)
	2013	2014	2015
School characteristics $t - 1$:			
Log nr students	-0.358***	-0.216***	-0.078
	(0.032)	(0.025)	(0.051)
Student-to-teacher ratio	0.016***	0.012**	0.003
	(0.006)	(0.005)	(0.009)
Separation rate	0.008	-0.079	0.015
	(0.158)	(0.140)	(0.301)
Exit rate	0.009	-0.107	-0.188
	(0.198)	(0.180)	(0.324)
Certified (share)	-0.020	-0.367	-0.225
	(0.249)	(0.243)	(0.414)
Female (share)	0.036	-0.164	0.045
	(0.139)	(0.142)	(0.216)
Mean age (years)	0.022**	0.011	-0.002
	(0.009)	(0.007)	(0.012)
Mean experience (years)	-0.009	0.000	0.000
	(0.008)	(0.007)	(0.011)
Math/natural science (share)	0.002	0.037	0.166
	(0.168)	(0.163)	(0.216)
Swedish/social science (share)	0.082	0.205*	0.118
	(0.118)	(0.111)	(0.253)
District FE	Yes	Yes	Yes
R^2	0.362	0.366	0.536
Ν	2,620	1,562	297

 Table B.5. Factors that predict timing of participation

Note: This presents results of the regressions $Year_{sd} = \phi_{dt} + \beta_t X_{sdt-1} + \varepsilon_{sdt}$ where $Year_{sd}$ is equal to the year that the school first has promoted a career teacher. Regressions are estimated separately by year, as indicated by the column headings. Regressions only include schools that at some point participate in the reform between 2013 and 2016, and that at the earliest participate in the year indicated by the column heading. Columns (2) and (3) therefore includes schools that had not yet promoted teachers earlier than 2014 (col. 2) or 2015 (col. 3). Standard errors are clustered at school district level.

	(1)	(2)	(3)	(4)	(5)	(6)
	2013	2013	2014	2014	2015	2015
Log nr students	_	-	-	-	-0.094	-
-	0.383***	0.360***	0.244***	0.239***		0.163***
	(0.033)	(0.039)	(0.030)	(0.035)	(0.057)	(0.060)
Student/teacher	0.016**	0.012	0.010**	0.005	0.000	0.009
	(0.006)	(0.008)	(0.005)	(0.005)	(0.009)	(0.013)
Separation rate	0.063	-0.080	-0.013	0.121	-0.072	-0.680*
	(0.172)	(0.172)	(0.151)	(0.197)	(0.327)	(0.373)
Exit rate	-0.017	0.074	-0.231	-0.348	-0.026	0.570
	(0.214)	(0.238)	(0.207)	(0.229)	(0.330)	(0.463)
Certified	0.234	-0.009	-0.309	-0.168	-0.082	-0.249
	(0.262)	(0.276)	(0.249)	(0.263)	(0.506)	(0.568)
Female	-0.101	0.114	-0.225	-0.039	-0.079	0.396
	(0.178)	(0.179)	(0.175)	(0.185)	(0.274)	(0.243)
Mean age	0.021**	0.031***	0.014*	0.018*	0.005	0.019
	(0.009)	(0.009)	(0.008)	(0.010)	(0.012)	(0.013)
Mean exp.	-0.012	-0.017*	-0.003	-0.009	-0.006	-0.025*
	(0.009)	(0.009)	(0.008)	(0.010)	(0.011)	(0.014)
Math/nat. sci.	-0.081	0.213	0.017	0.182	0.229	0.710**
	(0.175)	(0.215)	(0.172)	(0.204)	(0.239)	(0.273)
Swe./soc. sci.	-0.044	0.031	0.093	0.174	0.063	-0.238
	(0.128)	(0.165)	(0.118)	(0.160)	(0.271)	(0.232)
3rd grade math	0.006		0.096**		-0.029	
-	(0.043)		(0.040)		(0.067)	
3rd grade Swe.	0.017		-0.062		0.073	
-	(0.057)		(0.043)		(0.091)	
6th grade math		0.140**		-0.001		-0.182*
		(0.062)		(0.067)		(0.100)
6th grade Eng.		-0.039		-0.029		0.144*
		(0.066)		(0.067)		(0.083)
6th grade Swe.		-0.080		0.005		0.062
		(0.054)		(0.067)		(0.116)
District FE	Yes	Yes	Yes	Yes	Yes	Yes
R^2	0.381	0.420	0.371	0.418	0.567	0.669
Ν	2,205	1,828	1,332	1,060	254	185

 Table B.6. Factors that predict timing of participation – with student test scores

Note: This presents results of the regressions $Year_{sd} = \phi_{dt} + \beta_t X_{sdt-1} + \varepsilon_{sdt}$ where $Year_{sd}$ is equal to the year that the school first has promoted a career teacher. Regressions are estimated separately by year, as indicated by the column headings. Regressions only include schools that at some point participate in the reform between 2013 and 2016, and that at the earliest participate in the year indicated by the column heading. Standard errors are clustered at school district level and included in parenthesis.

	(1)	(2)	(3)	(4)
	Log wages	Wage dis- persion	Separations	Exits
Panel A: By CT type				
Coaching and mentoring	0.020***	0.003***	-0.004	0.003
R^2	0.913	0.578	0.369	0.261
N	3,498	3,498	3,498	3,498
Teaching development	0.021***	0.004***	-0.020	-0.018**
R^2	0.897	0.538	0.357	0.234
N	2,442	2,442	2,442	2,442
Panel B: By teacher seniority				
Senior teachers	0.022*** (0.001)	0.004***	-0.009* (0.005)	-0.004
R^2	0.929	0.552	0.301	0.206
N	17,682	17,622	17,677	17,677
Junior teachers	0.000	0.000	-0.001	0.003
_	(0.002)	(0.000)	(0.013)	(0.011)
R^2	0.798	0.426	0.265	0.240
N	14,709	10,905	12,970	12,970
Panel C: By district location				
Urban areas	0.019***	0.004***	-0.017**	-0.007
D)	(0.002)	(0.000)	(0.007)	(0.004)
<i>K</i> ²	0.926	0.589	0.354	0.240
1N	10,874	10,874	10,874	10,874
Rural areas	0.020*** (0.002)	0.004*** (0.000)	-0.001 (0.007)	-0.001 (0.005)
R^2	0.890	0.572	0.318	0.235
N	6,815	6,815	6,815	6,815
Panel D: Districts following decision rule				
50th percentile and above	0.023***	0.004***	-0.014	-0.006
R^2	0.904	0.572	0.341	0.238
Ν	6,655	6.655	6.655	6,655

 Table B.7. Heterogeneous effects w.r.t teacher and district characteristics

	(1)	(2)	(3)	(4)
	Log wages	Wage dis- persion	Separations	Exits
75th percentile and above	0.022*** (0.003)	0.004*** (0.001)	-0.014 (0.012)	-0.012 (0.009)
R ² N	0.880 2,237	0.569 2,237	0.320 2,237	0.239 2,237

Table B.7. Heterogeneous effects w.r.t teacher and district characteristics- continued from previous page

Note: In the table, we relate the change in the presence of career teachers in a school in year t to the change in mean wages (col. 1), wage dispersion (col. 2), school separations (col 3) and exits (col 4). School controls are the student to teacher ratio and log number of students. We include year and school FE in all regressions. In Panel A we divide by the school district design of the career teachers. Career teacher types are derived from the career teacher tasks displayed in Figure 3.2 using principal components analysis resulting in 48 school districts adopting the Coaching and mentoring type and 49 school districts adopting the Teaching development type. Figure A.3 shows the component loadings. Panel B displays separate regressions by teacher seniority status where senior teachers are those with at least five years of teaching experience. In Panel C, urbanization is defined using Eurostat's degree of urbanization (degurba) variable. School districts that are in cities (code 1) or towns and suburbs (code 2) are treated as urban, while school districts that are in rural (code 3) areas are treated as rural. There are 111 urban school districts and 179 rural school districts. In Panel D, we run regressions separately by whether school districts follow a rule to allocate promotions. The rule considered is whether school districts allocate career teaching positions in proportion to the share of students at the school within the school district (i.e. whether they apply the national rule in the school district). The 50th percentile correlation coefficient is 0.683 (140 school districts) and the 75th percentile is 0.829 (70 school districts).

	(1)	(2)
Panel A:		
At least one CT at school	-0.002*	0.001
	(0.001)	(0.001)
R^2	0.366	0.869
Ν	316,290	252,083
Panel B:		
Share CT at school	-0.001	0.044***
	(0.008)	(0.007)
R^2	0.366	0.869
Ν	316,290	252,083
Year FE:s	Yes	Yes
School FE:s	Yes	Yes
School controls	Yes	Yes
Teacher controls		Yes
Control mean	10.231	10.243

Table B.8. Log wages among non-promoted teachers in participating vs. non-participating schools

Note: The table shows estimates from equation 3.8 where the dependent variable is the log monthly full-time non-CT teaching wage. That is, we relate the change in the presence of career teachers within a school in year t to the log wage in year t in the sample of non-CT teachers. School controls included are the student to teacher ratio and log number of students. Lagged teacher controls are gender, age band, experience band, tenure band, level of education, educational specialization, wage decile, and whether the teacher is certified. Standard errors are clustered at school district level and reported in parenthesis.

	č	ade 3		Grade 6			Grade 0	
	Math	Swedish	Math	English	Swedish	Math	English	Swedish
Panel A								
Treat	0.015	0.023^{**}	0.035^{***}	0.018*	0.022*	-0.00	0.00	0.015
	(0.014)	(0.010)	(0.012)	(0.010)	(0.012)	(0.013)	(0.011)	(0.012)
R^2	0.465	0.489	0.596	0.585	0.583	0.681	0.727	0.697
N	12,599	12,608	10,566	10,564	10,572	5,263	5,267	5,270
Panel B								
Share	0.097***	0.097***	0.109^{***}	0.044^{**}	0.075***	0.005	0.025	0.051*
	(0.037)	(0.028)	(0.027)	(0.022)	(0.027)	(0.028)	(0.032)	(0.028)
R^2	0.465	0.489	0.597	0.585	0.583	0.681	0.727	0.697
Z	12,599	12,608	10,566	10,564	10,572	5,263	5,267	5,270
Year FE:s	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
School FE:s	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
School controls	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
<i>Note:</i> In the table we (a dummy equal to c	e regress the star one if there is at	ndardized national t least one subject	l exam result on s t & level career to	ubject & level-sp eacher at the sch	ecific reform vari ool). In Panel B	ables. In Panel A we use the treat	we use the treatr ment variable "sh	nent variable "treat" are" (the number of
subject & level care	er teachers at th	e school in year t	t divided by the n	umber subject &	level teachers in	the school in t -	- 1). School cont	rols are log number

 Table B.9. Student performance – detailed

175

of students and student to teacher ratio. The results of the national exams are standardized by year, subject and level to have mean 0 and st.d. 1. Each school

obtains a mean standardized score. Standard errors are clustered by school district.

	(1)	(2)	(3)
Panel A: Pool across subject	Grade 3	Grade 6	Grade 9
Coaching and mentoring CT	0.034	0.028*	0.012
	(0.021)	(0.016)	(0.017)
R^2	0.490	0.632	0.729
Ν	4,936	6,262	3,029
Development work CT	0.026	0.028	0.009
	(0.025)	(0.025)	(0.023)
R^2	0.424	0.497	0.696
Ν	3,343	4,282	2,199
Year \times subject FE:s	Yes	Yes	Yes
School \times subject FE:s	Yes	Yes	Yes
School controls	Yes	Yes	Yes
Panel B: Pool across level	Math	English	Swedish
Coaching and mentoring CT	0.032*	0.009	0.030**
	(0.017)	(0.017)	(0.012)
R^2	0.535	0.687	0.623
Ν	5,564	3,096	5,567
Development work CT	0.025	0.015	0.025
	(0.022)	(0.021)	(0.016)
R^2	0.472	0.560	0.497
Ν	3,831	2,160	3,833
Year \times level FE:s	Yes	Yes	Yes
School \times level FE:s	Yes	Yes	Yes
School controls	Yes	Yes	Yes

Table B.10. Student performance by career teacher type

Note: The table presents the results of estimating eq. 3.10. In Panel A we pool the results across subjects. In Panel B we pool the results across levels. Treatment variables are "treat" (a dummy equal to one if there is at least one subject & level career teacher at the school), and the results are run separately for Coaching and mentoring CTs and Development work CTs. School controls are log number of students and student to teacher ratio. Standard errors are clustered at school district level and reported in parenthesis.

Appendix C: Robustness checks

	(1)	(2)	(3)	(4)
	Log wages	Wage disp.	Separations	Exits
Panel A: Baseline results				
At least one CT	0.019***	0.004***	-0.010*	-0.004
	(0.001)	(0.000)	(0.005)	(0.003)
R ²	0.914	0.583	0.339	0.238
N	17,689	17,689	17,689	17,689
Panel B: No school controls				
At least one CT	0.019***	0.004***	-0.010**	-0.003
	(0.001)	(0.000)	(0.005)	(0.003)
	0.914	0.582	0.311	0.230
N	17,700	17,700	17,700	17,700
Panel C: Eligible teachers				
At least one CT	0.022***	0.004***	-0.011**	-0.004
	(0.001)	(0.000)	(0.005)	(0.003)
R ²	0.930	0.560	0.304	0.212
N	17,684	17,628	17,684	17,684
Panel D: Weighted				
At least one CT	0.013***	0.003***	-0.009**	-0.003
	(0.001)	(0.000)	(0.004)	(0.003)
R ²	0.936	0.641	0.377	0.249
N	17,689	17,689	17,689	17,689
Panel E: Excl. never participating schools				
At least one CT	0.015***	0.003***	-0.009*	-0.002
	(0.001)	(0.000)	(0.005)	(0.004)
R ²	0.921	0.588	0.342	0.241
N	15,808	15,808	15,808	15,808
Panel F: With boost controls				
At least one CT	0.019***	0.004***	-0.009*	-0.004
	(0.001)	(0.000)	(0.005)	(0.003)
R ²	0.915	0.583	0.340	0.239
N	17,689	17,689	17,689	17,689

Table C.1. Sensitivity checks – Participating schools

Note: We relate the change in the presence of career teachers in a school to the change in mean wages, wage dispersion, school separations and exits. The regressions (apart from Panel B) control for student to teacher ratio and log number of students. In Panel F we also include controls for the boost reforms (math, reading and teacher wage boost). Year FE and school FE are included in all regressions. Standard errors are clustered by school district.

	(1)	(2)	(3)	(4)
	Log wages	Wage disp.	Separations	Exits
Panel A: Baseline results	-	•		
Share CT	0.165*** (0.007)	0.024*** (0.001)	-0.207*** (0.025)	-0.057*** (0.015)
R^2	0.921	0.596	0.345	0.239
Ν	17,689	17,689	17,689	17,689
Panel B: No school controls				
Share CT	0.161*** (0.006)	0.024*** (0.001)	-0.264*** (0.024)	-0.078*** (0.015)
R^2	0.921	0.596	0.321	0.231
Ν	17,700	17,700	17,700	17,700
Panel C: Eligible teachers				
Share CT	0.175*** (0.007)	0.022*** (0.001)	-0.240*** (0.026)	-0.078*** (0.017)
R^2	0.936	0.577	0.311	0.214
Ν	17,684	17,628	17,684	17,684
Panel D: Weighted				
Share CT	0.174*** (0.007)	0.022*** (0.001)	-0.230*** (0.021)	-0.067*** (0.014)
R^2	0.942	0.653	0.383	0.250
Ν	17,689	17,689	17,689	17,689
Panel E: Excl. never participating schools				
Share CT	0.161*** (0.006)	0.022*** (0.002)	-0.234*** (0.025)	-0.059*** (0.016)
R^2	0.929	0.603	0.350	0.242
Ν	15,808	15,808	15,808	15,808
Panel F: With boost controls				
Share CT	0.163*** (0.007)	0.024*** (0.001)	-0.203*** (0.025)	-0.054*** (0.015)
R^2	0.922	0.596	0.346	0.240
Ν	17,689	17,689	17,689	17,689

 Table C.2. Sensitivity checks – Share of career teachers

Note: We relate the change in the presence of career teachers in a school to the change in mean wages, wage dispersion, school separations and exits. The regressions (apart from Panel B) control for student to teacher ratio and log number of students. In Panel F we also include controls for the boost reforms (math, reading and teacher wage boost). Year FE and school FE are included in all regressions. Standard errors are clustered by school district.
	(1)	(2)	(3)
Panel A: Pool across subject	Grade 3	Grade 6	Grade 9
Treat	0.018*	0.025***	0.005
	(0.010)	(0.008)	(0.008)
R^2	0.474	0.590	0.702
Share	0.095***	0.074***	0.025
	(0.028)	(0.018)	(0.020)
R^2	0.474	0.591	0.702
Ν	25,207	31,702	15,800
Year \times subject FE:s	Yes	Yes	Yes
School \times subject FE:s	Yes	Yes	Yes
School controls	Yes incl boost	Yes incl boost	Yes incl boost
Panel B: Pool across level	Maths	English	Swedish
Treat	0.018**	0.015*	0.020***
	(0.008)	(0.008)	(0.007)
R^2	0.533	0.627	0.561
Share	0.083***	0.039**	0.075***
	(0.017)	(0.018)	(0.017)
R^2	0.533	0.627	0.561
N	28,428	15,831	28,450
Year \times level FE:s	Yes	Yes	Yes
School \times level FE:s	Yes	Yes	Yes
School controls	Yes incl boost	Yes incl boost	Yes incl boost

 Table C.3. Student performance – with boost controls

Note: The table presents the results of estimating eq. 3.10, with additional controls for the Math and Reading Boost reforms. In Panel A we pool the results across subjects. In Panel B we pool the results across levels. Treatment variables are "treat" (a dummy equal to one if there is at least one subject & level career teacher at the school) or "share" (the number of subject & level career teachers at the school in year *t* divided by the number subject & level teachers in the school in t - 1). School controls are log number of students, student to teacher ratio, and dummies for participating in the boost reforms. The results of the national exams are standardized by year, subject and level to have mean 0 and st.d. 1. Each school obtains a mean standardized score. Standard errors are clustered at school district level.

	Gra	ide 3		Grade 6			Grade 9	
	Math	Swedish	Math	English	Swedish	Math	English	Swedish
Panel A: Lead subject CT								
Share	0.095^{**}	0.089^{***}	0.108^{***}	0.044^{*}	0.073^{***}	0.006	0.020	0.045
	(0.038)	(0.028)	(0.027)	(0.022)	(0.027)	(0.028)	(0.031)	(0.028)
Lead share	0.019	0.047	0.014	0.009	0.022	-0.012	0.042^{**}	0.051^{**}
	(0.044)	(0.033)	(0.022)	(0.017)	(0.028)	(0.025)	(0.019)	(0.023)
R^2	0.465	0.489	0.597	0.585	0.583	0.681	0.727	0.697
N	12,599	12,608	10,566	10,564	10,572	5,263	5,267	5,270
Panel B: Share all teachers								
Subject/level CT as share of all teachers	0.166^{*}	0.165^{***}	0.422^{***}	0.296^{**}	0.234*	0.040	0.188	0.326^{*}
	(0.092)	(0.061)	(0.115)	(0.122)	(0.120)	(0.204)	(0.186)	(0.195)
R^2	0.465	0.489	0.597	0.585	0.583	0.681	0.727	0.697
N	12,599	12,608	10,566	10,564	10,572	5,263	5,267	5,270
Panel C: Participating schools								
At least one CT at school	0.005	0.002	0.028^{**}	0.010	0.025^{**}	-0.010	-0.007	0.026
	(0.017)	(0.010)	(0.014)	(0.011)	(0.012)	(0.014)	(0.017)	(0.016)
R^2	0.465	0.489	0.596	0.585	0.583	0.681	0.727	0.697
Z	12,599	12,608	10,566	10,564	10,572	5,263	5,267	5,270
<i>Note:</i> We regress standardized test scores on s career teachers in year <i>t</i> over the number of sub teaching-level career teachers in year <i>t</i> over all	subject & teac bject and teac l teachers in t	thing level-spe hing level teac - 1. In Panel	scific reform v hers in $t - 1$, a C we use the	ariables. In and the lead of a presence of a	Panel A we us of that share. Ir at least one car	e the number 1 Panel B we teer teacher a	r of subject & use the numbe t the school.	teaching level er of subject & All regressions

include year FE, school FE, log number of students and student to teacher ratio. Standard errors are clustered by school district.

Table C.4. Sensitivity checks – Student performance

4. Spousal Earnings and Household Dynamics: Evidence From a Promotion Reform

Co-authored with Erik Grönqvist and Lena Hensvik

Acknowledgments: We thank Erica Lindahl, Rune Vejlin as well as participants in Uppsala University's Equality of Opportunity group for helpful comments on this paper.

4.1 Introduction

Compared to men, women spend considerably more time caring for children and the household in all OECD countries (OECD 2016). On an average day, American women spend 0.77 hours more on household activities compared to men and 0.62 fewer hours on paid work.¹ Even in Sweden, known as a progressive country with equal labor force participation of men and women and with generous childcare policies for mothers and fathers, women spend about an hour more on household chores, and work one hour less, than men each day.² Swedish mothers also take 70 percent of the paid parental leave to care for a children when they are small (Försäkringskassan 2020).

A recent literature suggests that these differences in time allocation across mothers and fathers are crucial to understanding the remaining gender pay gap (Goldin 2014; Cortés and Pan 2020).³ Yet, the underlying causes behind the uneven distribution of household and family responsibilities are still not well-understood. Gender pay inequality may contribute to the persistence by enabling division of labor to exploit comparative advantages (Becker et al. 1977) or limit women's bargaining power (Lundberg and Pollak 1993). At the same time, prior literature also highlights the role of preferences or gender-specific norms as obstacles towards a more equal division of household and family responsibilities (Akerlof and Kranton 2000; Bertrand et al. 2015; Kleven et al. 2020b).

In this paper we study how a sudden and substantial improvement in female labor market outcomes affects the spousal division of time to care for children. More specifically, we leverage a Swedish promotion reform in a traditionally female-dominated occupation with relatively low wages, namely teaching. Teaching is the second most common occupation among Swedish mothers and the majority of teachers are secondary earners of the household.⁴ In 2013, Sweden introduced a new career step in the teaching profession, aimed at increasing the attractiveness of the occupation. The new career step entailed a substantial monthly wage increase corresponding to nearly 20 percent of mean pre-reform wages. Consequently, the reform led to a sudden and substantial shift in affected couples' relative wages. By disproportionately af-

¹Based on 2019 American Time Use Survey data.

²Based on the 2010/11 Swedish Time Use Survey, which is the latest survey conducted.

³For example, Duchini and Van Effenterre (2020) find that reduced time commitments to child care for mothers raised their average wages in France; Cortés and Pan (2019) show that low-skilled immigration inflows to the U.S have decreased the gender pay gap in occupations that disproportionately reward long work hours; and Denning et al. (2019) show that differences in hours worked have slowed women's labor market progress.

⁴47 percent of all working mothers with children under the age of 11 work in the public sector compared to 17 percent of fathers. 15 percent are found within the three detailed 4-digit occupations nursing, lower secondary teaching and childcare. Prior literature suggest that women's preferences for these public sector jobs is highly linked to motherhood (Hotz et al. 2017; Pertold-Gebicka et al. 2016).

fecting female spouses, the reform also led to an average increase in women's contribution to household earnings.

We focus on dual-earner couples with children, and quantify the impact of this positive labor market shock on the reallocation of time spent to care for children. We present results both for all promoted persons and for promoted women only. Information on childcare is inferred from the relative use of Temporary Parental Leave (TPL), a benefit available to either spouse to care for an ill child during working hours. In Sweden, mothers and fathers of children aged between eight months and twelve years can claim up to 120 days of TPL annually to care for sick children, reimbursed at 80 per cent of current earnings (up to a cap). However, the families do not divide the days equally; the division is skewed towards women (60 percent) and is a good indicator of the gender division of household work more generally (Eriksson and Nermo 2010; Ichino et al. 2019).⁵

Our empirical strategy uses an event study design relying on the *timing of promotion*. The promotion reform was implemented gradually due to budgetary constraints, and we use Swedish register data to estimate within-couple differences in wages and use of TPL around the time of promotion in households where one of the spouses was promoted between 2013 and 2015. We focus on ever promoted teachers as teachers that receive promotions may differ from teachers that do not get promoted. We follow couples over time, even if they separate, and show that within-couple outcomes evolve very similarly in couples with a promoted partner and not-yet promoted partner, supporting the validity of the empirical design.

Reflecting the female nature of the teaching profession, the vast majority (almost 80 percent) of promoted teachers are women. As a consequence, we show that the within-household gender pay gap declines by 6.4 percentage points (32 percent) in couples affected by the reform. The probability that the woman is the primary earner furthermore increases by 9.1 percentage points (30 percent). Promoted spouses respond to these new economic positions by reducing their use of temporary parental leave relative to their partners. The within-couple difference in TPL benefits (as share of monthly wages) declines by three percentage points, or approximately 0.6 days. We find a similar response when restricting the sample to couples where the female spouse is promoted, suggesting a non-trivial narrowing of more than half of the pre-promotion gender imbalance in childcare time. Estimating the impact of TPL use separately for both the promoted woman and her spouse, the results suggest that promotions induce a reallocation within the couple: the promoted woman reduces the number of days taken, while the male spouse

⁵Unequal division of care for sick children may directly increase the gender wage gap. For example, Azmat et al. (2020) show that TPL is negatively related to wages in jobs with few employee substitutes, suggesting that unpredicted absence is problematic for firms.

increases the number of days taken.⁶ The impact on partner TPL use is partly driven by a change on the extensive margin, increasing their propensity to use TPL by 2.5 percentage points (or 5 percent).

While the impact on TPL is highly interesting in its own right, prior literature suggests that the division of TPL is also linked to couples' overall allocation of time spent on household chores (Ichino et al. 2019). A back-of-the-envelope calculation based on our estimates for TPL use and the association between TPL days and household work in prior literature suggests that promotion of the mother increases the father's share of total home production by 6 percentage points, or 13 percent.⁷

The response in TPL use is considerably larger in couples with pre-school children, where the overall childcare demand is higher. Furthermore, the promoted person responds 50% stronger if the promotion makes them the primary earner. This suggests that the bargaining positions within the couple have a non-linear relationship to relative wage. Finally, we consider how the effect varies depending on spousal job characteristics. There is no differential effect on the within-couple difference in TPL when the spouse has a high ability to work from home. However, for promoted women the effect is over twice as large if the non-promoted spouse is also a teacher, possibly suggesting a larger shift in comparative advantage or relative bargaining power when the couple is on the same labor market.

Our paper contributes to the literature on how women's relative earnings relate to the household allocation of time between market work and home production by exploiting a policy reform that led to a substantial wage increase for highly educated mothers. While Bertrand et al. (2015) show that higher female relative income is associated with less time spent on home production until the woman becomes the primary earner, at which point she tends to endogenously adopt a more traditional gender role, we find no indication that promoted mothers compensate by increasing their share of family responsibilities once they become the primary earner. Our paper is also closely related to Ichino et al. (2019) who focus on the relationship between taxation and home production, finding that tax cuts generated by the Swedish EITC induced couples to reallocate time from childcare to the market. Whereas the response to tax cuts is symmetric for husbands and wives in the average couple, seemingly traditional couples respond more strongly to husband tax cuts.⁸

⁶As the reform targeted a female-dominated occupation, we do not have enough power to separately estimate the TPL-response for male promoted spouses.

⁷Two years prior to the promotion reform, in 2011, fathers' fraction of the weekly total amount of non-paid work in couples living together with small children was 45 percent (39.55 hours for women and 32.20 hours for men), according to the Swedish Time Use Survey (see Stanfors (2018) for a detailed description of how non-paid work differs between men and women depending on couple characteristics).

⁸Gelber and Mitchell (2012) find that single women in the US respond to stronger economic incentives by substituting housework for market work. However, due to potential adjustment

In contrast to these studies, we focus on an unusually large wage shift in a predominantly female segment of the labor market, enabling us to credibly assert the impact of women's labor market position. An important difference between the aforementioned studies and our paper is that we study the role of promotions in contrast to pure wage effects.⁹ While wages and promotions are highly linked, the consequences of a promotion may be broader than simply monetary. For instance, promotions may represent a shift in status and power within the workplace that spills over to the bargaining power in the family. It is important to note that the reform we study stipulated that teachers should continue to teach and that the workload of the promoted teachers appear to have remained fairly constant.¹⁰ Even if there is a shift in job responsibilities, we believe that the main influence of the new career step on the family was the sudden and substantial change in relative wage and changed status.

Our paper also contributes to the literature and discussion on the effectiveness of policy in addressing the unequal division of household and family responsibilities. Sweden and other countries have implemented a series of family policy reforms to facilitate the combination of parenthood and careers. Such reforms, however, appear to have little impact on fathers' involvement in the care of children (Ekberg et al. 2013) or on the overall child penalty (Kleven et al. 2020a).¹¹

By showing that couples respond to promotions by reallocating childcare time to the non-promoted spouse, our results highlight that reforms improving the careers of mothers can be effective in promoting gender equality in the domestic sphere. As such, labor market policies directly targeting gender pay inequalities and career opportunities can be a powerful tool in promoting a more equal division of child responsibilities between mothers and fathers. Our results suggest that more traditional economic explanations manifested in the labor market should not be overlooked, thus partly contrasting the recent literature highlighting the role of norms in explaining the persistence of traditional gender roles in the family.

frictions or norms it is not clear that these substitution effects are transferable to women in couples.

⁹Folke and Rickne (2020) consider the effect of promotions in a different context. They study how promotions to top political jobs affect marriage stability, finding that promotions increase the likelihood of divorce for promoted women but not promoted men.

¹⁰Statskontoret (2015) concludes that total working time did not increase due to a promotion. See Section 4.2.1 for a more detailed description of the reform.

¹¹Ekberg et al. (2013) study how the introduction of mandated paternity leave in Sweden affects fathers' share of TPL. While the daddy-month reform was supposed to promote gender equality in the household, there is no evidence that the reform had an impact on fathers' involvement in household work in addition to a positive effect on their paternity leave. In addition, Avdic and Karimi (2018) show that marriage stability was negatively affected by the reform. Studies from Germany, Canada and Spain do, however, provide more encouraging results regarding the relationship between paternity leave policies and fathers' involvement in childcare (see Schober 2014; Patnaik 2019; and Farré and González 2019).

The paper proceeds as follows. It begins by describing the institutional context in Section 4.2. In Section 4.3 we describe our empirical strategy and the data used. Section 4.4 presents summary statistics for teachers and couples before the reform and describes how the reform affected household relative wage. Section 4.5 presents the main results on the impact of promotions on the household division of TPL. Section 4.6 concludes.

4.2 Institutional setting

The labor market for secondary school teachers is dominated by women. In the year prior to implementing the promotion reform, the fraction of female teachers was 78 percent (Grönqvist et al. 2020). As in many other countries, the teacher labor market is also characterized by low and compressed wages compared to other occupations with similar education requirements (four years of higher education). As a consequence the majority of teachers are secondary wage earners in the household.

4.2.1 The promotion reform

To make the teaching profession more attractive by aligning teacher pay more closely to teacher skills, the Swedish government introduced a new career step for teachers in 2013 called 'career teachers'.¹² This promotion program was launched as a response to deteriorating student achievement and aimed to improve student outcomes by keeping and encouraging high quality teachers (Regeringen 2013b).

Promoted teachers receive a significant monthly wage increase of 5,000 SEK (520 USD), which is fully funded by the state and corresponds to nearly 20 percent of mean pre-reform wages. Career teachers primarily continue to teach but also engage in tasks like coaching their colleagues. Total working time did not increase in general due to a promotion, and it was delegated to the school districts (and principals) to free up time for the new responsibilities while maintaining at least a 50 percent teaching load. Many promoted teachers report that the time set aside was not sufficient, which infringed on their capacity to carry out the new tasks and for principals to make proper use of promoted teachers. As promotions reduced the time in the classroom it also opened up opportunities to more flexibly allocate working time and working location to adapt to domestic needs (Statskontoret 2017).

¹²This section draws on Grönqvist et al. (2020). In the companion paper Grönqvist et al. (ibid.) we study the impact of the career teacher reform on teacher and school outcomes. The reform had full pass-through onto wages and schools that implement career teacher promotions have lower teacher turnover, a more qualified teaching pool and higher student test scores in Math, English and Swedish in grades 3 and 6. In this paper, we instead turn attention to the impact of the reform-induced promotions on household allocation of childcare vs. market work.

The reform was rolled out over four years, and the number of positions increased year-on-year in line with the funding provided by the state. Career teacher positions were allocated to school districts in proportion to their total student population. School districts, in turn, allocated career teacher positions to individual schools at their discretion. In practice, school principals at individual schools often recruited teachers to become career teachers. According to the reform's regulation (Regeringen 2013a), the teacher needed to fulfill certain criteria to qualify for promotion. They needed to be certified; have at least four years of teaching experience; demonstrate an ability to improve student outcomes and a keen interest to develop teaching; and be deemed particularly qualified as a teacher. Because eventually promoted teachers differ in observable ways from teachers that are not promoted (see Grönqvist et al. 2020), we focus on promoted teachers only in our empirical strategy, and utilize the fact that promotions happened at different points in time.

4.2.2 Temporary parental leave (TPL) system

The Swedish parental leave system is very flexible, allowing parents up to 120 days of leave annually to care for sick children during work hours. In 2019, 877,000 parents received TPL to care for 801,000 children (Försäkringskassan 2020). TPL is normally used to care for children with minor illnesses such as common colds, and take-out therefore tends to follow seasonal patterns. It can be taken to care for children aged 8 months until the child turns 12, though parents cannot generally take leave at the same time. It is reimbursed at almost 80 per cent of current earnings (up to a cap), and can be taken out as full days, three quarters of a day, half days, one quarter of a day or one eight of a day.¹³

This generous system, which facilitates parental reconciliation of work and family duties, is not utilized to the same extent by mothers and fathers. While it is more equally shared than the use of standard parental leave, about 60 percent of TPL is taken by mothers (ibid.). While interesting to study in its own right, TPL has also been found to proxy well for the general gender division of household chores, and is, unlike other time-use data, available in registry data (Eriksson and Nermo 2010). In fact, Ichino et al. (2019) provide evidence of a positive relation between fathers' share of TPL and their share of total home production for dual earner couples with children aged 3 and above (conditioning on total spouse home production time and their human capital levels), suggesting that TPL is an informative proxy for the male contribution to household work.

¹³If earnings are above a cap, TPL is instead reimbursed at almost 80% of the cap. The cap is set at 7.5 so-called price basis amounts. In 2020, the cap was 29,563 SEK (3,400 USD) per month. A price basis amount is used in Sweden for the calculation of various benefits and fees, and is adjusted annually for inflation.

4.3 Empirical strategy and data

We are interested in how promotions affect household behavior. To estimate the impact of promotions on the relative take-up of temporary parental leave and relative wage changes within couples, we consider the following specification for households:

$$\tilde{y}_{hct} = \alpha + \beta Promoted_{hct} + \lambda_c + \sum_a \lambda_{ac}^i + \sum_a \lambda_{ac}^j + \theta \tilde{I}_{h,-4} + \varepsilon_{hct}$$
(4.1)

 \tilde{y}_{hct} is the within-household difference in the outcome of interest in household *h* in calendar year *c* observed *t* years before or after promotion, and *Promoted*_{hct} is a dummy variable equal to one for the household from the year of promotion onward. We estimate equation 4.1 using couples where one of the spouses was promoted during the period 2013 to 2015. We include calendar year fixed effects, λ_c , to control for general shocks and trends in the economy. $\lambda_{ac}^g, g \in i, j$ are age dummies for the promoted person *i* and the partner *j* to control for that the age trajectories in the outcome may differ by spouse.¹⁴ To avoid data sparseness, we bin the ages by five year bands. $\tilde{I}_{h,-4}$ is the within-household difference in income four years prior to promotion, included to capture how (un)equal the couple is. We test the robustness of the results in the appendix. When we consider TPL outcomes we also control for the number of children aged 0–3, 4–6 and 7–10. Standard errors ε_{hct} are clustered by municipality (equivalent to the school district for public schools).¹⁵

Couple differences in outcomes are defined as the promoted person minus the non-promoted spouse. Estimates of β therefore show the change in the within-couple difference in wages and temporary parental leave between the promoted teacher and their partner relative to the within-couple difference between not-yet promoted teachers and their partners. We also consider the effect of promotion on the gender wage gap and on the probability that the wife earns more than the husband. For TPL, we also consider the likelihood that the person takes out TPL.

The empirical strategy relies on the assumption that, in absence of promotion, the outcome variable would have evolved in parallel for promoted and not-yet promoted couples. Hence, the *timing* of when the teacher is promoted must be uncorrelated with changes in other determinants of the outcome that we do not control for. To assess whether the identifying assumption is likely to hold and to estimate the dynamics of being promoted, we adopt an event-study approach similar to Angelov et al. (2016), who study the effect of parenthood on the couple gender pay gap. Specifically, we consider the following expansion of equation 4.1:

¹⁴We also estimate the specification separately by gender of the promoted person. When doing so, the age dummies control for that age trajectories may differ by gender.

¹⁵In light of our long pre-period (we have data from 2007), we include a dummy in all estimations of equation 4.1 to capture observations more than three event years before promotion. The results are not sensitive to the inclusion of this dummy.

$$\tilde{y}_{hct} = \alpha + \sum_{t \neq -1} \gamma_t \mathbf{1}[t] + \lambda_c + \sum_a \lambda_{ac}^i + \sum_a \lambda_{ac}^j + \theta \tilde{I}_{h,-4} + \varepsilon_{hct}$$
(4.2)

where *t* denotes time relative to the year of promotion and promotion occurs when *t* equals $0.^{16}$ Observations three or more event years after promotion $(t \ge 3)$ are grouped.¹⁷ The year before promotion is omitted, such that the parameters of interest γ_t show the couples' gap in outcome relative to the year prior to promotion.

4.3.1 Data

The underlying population used for the analysis consists of all ever promoted teachers and their spouses. Using a family identifier that combines spouses, a family panel with annual information is created. Information is linked to the household from different registry data sources collected and compiled by Statistics Sweden. The panel includes detailed information about the couple such as their wages and earnings, place of work, use of social insurance benefits, number of children and demographic variables. We follow couples even if they separate, as long as they are a couple four years prior to the teacher's promotion. Data is drawn from three main data sources: a longitudinal individual-level database that covers all individuals in Sweden aged 16 to 74 (*LOUISE*), matched employee-employer data (*RAMS*), and structural earnings statistics. We have access to *LOUISE* between 2007 and 2017, and *RAMS* and the structural earnings statistics between 2007 and 2018.

Promoted teachers: We identify promoted teachers using occupational codes and full-time equivalent wages, which come from the structural earnings statistics. Structural earnings statistics are available for all individuals in the public sector and a sample of individuals in the private sector.¹⁸ Teachers are defined as individuals who have an occupational code as a teacher at their main place of work.¹⁹ While we do not have identifiers for actual promotions in the data, we exploit the institutional features of the reform: a teacher is

¹⁶Notice that this is not a standard event-specification, just like equation 4.1 is not a standard difference-in-difference specification, as we include the lagged difference in income instead of couple fixed effects. The results are robust to instead including couple fixed effects, reported in the appendix.

¹⁷As before, a dummy is also included to capture observations more than three event years before promotion (t < -3) in all estimations of 4.2. An alternative to binning endpoints is to fully saturate the model with event indicators. As Borusyak and Jaravel (2018) point out, in a standard event study design such a fully dynamic specification is underidentified and treatment effects can only be identified up to a linear trend. Thus, two pre-treatment indicators need to be omitted. We test this alternative specification and find that our results are robust.

¹⁸Approximately 50% of private sector employees are included. The sampling is stratified by industry and firm size, with an oversampling of larger firms.

¹⁹Precisely, we use occupation (*SSYK*) codes 2321, 2322, 2323, 2330, 2340, and 2351 until 2014, and codes 2320, 2330, 2341, 2351 from 2014 onward.

classified as promoted from year *s* until the last period of observation if the monthly full-time equivalent wage rises by 5,000 SEK or more between s - 1 and *s*, where *s* falls between 2013 and 2015.²⁰ The year of promotion is the year when the wage increase is observed. We identify 13,500 unique promoted teachers in our data.²¹

We have tested the robustness of this method in several ways. First, we estimate the wage increase associated with being promoted compared to teachers that are not promoted (see Table A.4). The results are very similar to those obtained in our companion paper Gröngvist et al. (2020), where we have access to data that includes promotion identifiers.²² Second, we have applied the method outlined above to the data used in Grönqvist et al. (ibid.), to compare how many actual promotions we capture using the promotion proxy. We have tested the sensitivity of the results to wage increase thresholds ranging from 3,000 SEK to 7,000 SEK, in 500 SEK intervals. Ideally, we want to capture as many promotions in the correct promotion year as possible, but not capture teachers that have not actually been promoted. We consider four measures to evaluate the proxy: how many actual promotions that the proxy captures, how many promotions the proxy captures in the correct year, how many promotions the proxy captures that it should not capture (termed "excess share"), and finally how many promotions in a particular year the proxy captures that it should not. The denominator in all comparisons is the number of actual promotions based on the career teacher register between 2013 and 2015 for whom the promoted person has an occupational code as a teacher (i.e. the sum of all possible career teacher promotions that we can capture). The results of this exercise are included in Figure A.1. This shows that the proxy does well at capturing promotions, albeit with some measurement error. When the wage increase threshold is set at SEK 5,000, we capture 81% of actual promotions in the right year (87% of actual promotions), and 21% excess promotions-byyear (15% excess promotions). The next-best alternative to using a threshold of SEK 5,000 is to use SEK 5,500. With this threshold we capture 80% of actual promotions in the right year (85% of actual promotions), and 16% excess promotions-by-year (11% excess promotions). We use a wage increase threshold of SEK 5.500 in robustness checks and find similar results.

²⁰We stop in 2015 because another teacher salary reform, the Teachers' Salary Boost, was implemented in 2016. This entailed a smaller unconditional wage increase to about half of all teachers.

²¹In Table A.1 in the appendix we include the number of promoted persons and couples per year. The number of promotions is slightly lower, but closely follows, official statistics included in Statskontoret (2017) who identify 3,076 promoted teachers in autumn 2013, 12,114 in autumn 2014 and 14,340 in autumn 2015, compared to 3,064, 10,623 and 13,489 in our data. That we have a slightly lower number could among others be due to data not allowing us to capture all teachers in voucher schools.

²²The reason why we cannot use promotion identifiers here is that these are not available in the same dataset as the TPL data.

Partners: The analysis data include information on the promoted person and their partner irrespective of how long the couple has been together, as long as they are a couple four years prior to the teacher's promotion. We identify the partner as the oldest non-promoted person with the same family identifier as the promoted person. To ensure that we do not capture single persons with children, we apply the restriction that the age difference between the promoted teacher and the partner cannot exceed 18 years. We exclude couples where both partners have been promoted as well as same-sex couples from our main sample.²³ After these restrictions we identify 9,908 unique partners, or equivalently couples.

Wages: The individual wage measure is monthly full-time equivalent wages at the main workplace. The main workplace is defined as the workplace where the person has the highest positive earnings in a given year, using information from the matched employee-employer data. Wages are taken from the structural earnings statistics and are therefore only available for a subset of individuals. It is available for everyone working in the public sector and a 50 percent stratified sample of workers in the private sector. Wages are measured between September and November each year.

Temporary parental leave (TPL): We observe the annual amount of TPL per person in Swedish crowns (SEK). The benefits are a function of the person's wage, up to a cap. As the temporary parental leave in SEK depends both on the parent's wage and the number of days taken, we divide the temporary parental benefits in SEK by the replacement rate (just under 80%) multiplied by the minimum of the person's wage and the cap each year.²⁴ We refer to this measure as normalized TPL. It can approximately be interpreted as the fraction of annual TPL out of the person's replaced monthly wage. We can convert the measure to number of TPL days per year by multiplying by the average number of days worked per month (21 days).²⁵ For ease of interpretation we report the approximated number of TPL days per year in our main results.²⁶ We also consider TPL on the extensive margin. This is a dummy variable equal to one if the person has received positive TPL benefits.

²³There are 164 couples where both partners have been promoted and 32 same-sex couples.

²⁴We consequently can only define this measure for the sub-sample of persons for whom we observe monthly wages. The replacement rate is $0.97 \times 0.8 = 77.6\%$.

²⁵To see this, note that we observe monthly full-time equivalent wages and annual *TPL*. Suppose the monthly wages are below the cap. Let D_m be days worked per month and w_d be the daily wage. The wage per month is thus $w_m = w_d \times D_m$. Let D_{TPL} be days with *TPL* per year and *TPL*_y be the annual amount of *TPL* in SEK. Then *TPL*_y = $D_{TPL} \times 0.776 \times w_d = D_{TPL} \times 0.776 \times \frac{w_m}{D_m}$. Rearranging we see that $D_{TPL} = \frac{TPL_y}{0.776 \times w_m} \times D_m$ where $\frac{TPL_y}{0.776 \times w_m}$ is the normalized measure of *TPL*. The same calculation follows if the wage is above the cap, but w_m is replaced with the cap.

²⁶We have ordered spell data to complement our current data. In future versions of the paper, we plan to update the TPL measure to TPL days based on spell data.

4.4 Descriptive statistics

4.4.1 Summary statistics: teachers and couples before promotion

Table 4.1 describes the couples in the year prior to promotion. Focusing on the promoted teachers, they are predominantly female (79%) and married (82%). The proportion of female promoted teachers is the same as the proportion of women teaching at the lower secondary level in total. Thus, there appears to be no strong gender bias in the selection of teachers for promotion.

Around half of promoted teachers have children under 11 years old (referred to as "young children"), and, conditional on having young children, 75% claim temporary parental leave in the year prior to promotion.²⁷ In line with the eligibility requirements for promotion, all promoted teachers have post-secondary education. On average, they are more educated than their partners. Despite this, they are more often secondary earners in the household and also the primary childcare takers (75% claim benefits compared to 58% of partners, conditional on having young children). Of these couples, where it is primarily but not only the woman that gets promoted, the share of wage earned by the wife prior to promotion is 46% on average, and 30% are in relationships where the woman has a higher wage already prior to promotion.²⁸

The summary statistics mask differences between promoted women and promoted men. In Appendix Table A.2 we split the couples by gender. Promoted men and women are similar in terms of their age (45 on average) and educational level (post-secondary or higher). 59% of promoted men have young children, compared to 49% of promoted women. Conditional on having children under the age of 11, a similar share of promoted women and men claim temporary parental leave (75% compared to 74%). Even so, within-couple differences show that promoted men generally claim less benefits than their partners prior to promotion, while promoted women and promoted women have similar wages, the women are typically in relationships with men that earn substantially higher wages (10,500 SEK more) prior to promotion, while the men are typically in relationships with women that earn around the same wage as they do (500 SEK more).

²⁷TPL can be claimed until the day the child turns 12. We use 11 rather than 12 because we observe the ages of children in intervals.

²⁸ In Table A.3 we include couples for whom we can define normalized TPL and who have young children ("TPL sample"). They are similar to the full sample along many dimensions, such as the pre-promotion differences in wages and the share wage earned by the wife. A somewhat lower share of promoted persons are female compared to the overall sample (70% compared to 79%) and they are younger on average (40 compared to 45).

	Promot	ed teacher	Pa	rtner
	Mean	sd	Mean	sd
Individual characteristics:				
Female	0.79	(0.41)	0.21	(0.41)
Married	0.82	(0.38)	0.82	(0.38)
Age	44.78	(7.85)	46.27	(8.64)
Compulsory	0.00	(0.02)	0.04	(0.19)
Upper secondary	0.00	(0.06)	0.31	(0.46)
Post-secondary or higher	1.00	(0.06)	0.65	(0.48)
Education missing	0.00	(0.01)	0.00	(0.04)
Young children	0.51	(0.50)	0.51	(0.50)
Claim TPL*	0.75	(0.43)	0.58	(0.49)
TPL (100 SEK)*	39.70	(58.95)	33.34	(57.51)
TPL (normalized)*	0.19	(0.28)	0.17	(0.26)
TPL days*	3.95	(5.87)	3.58	(5.55)
Non-missing wage	1.00	(0.00)	0.53	(0.50)
Monthly wage (SEK)	30,087	(3,165)	38,019	(17,581)
Annual earnings (SEK)	345,542	(71,761)	433,593	(272,555)
Couple characteristics:				
Difference monthly wages	-7.819	(17.539)		
Difference annual earnings	-88.227	(280.801)		
Difference TPL (100 SEK)*	6.63	(59.60)		
Difference TPL (normalized)*	0.02	(0.28)		
Difference TPL days*	0.37	(5.81)		
Share wage earned by wife	0.46	(0.08)		
Wife has higher wage	0.30	(0.46)		
Observations	9,908		9,908	

 Table 4.1. Summary statistics in year before promotion
 Properties
 Properies

Note: The table shows summary statistics for promoted teachers with partners (cols. 1 and 2), and partners (cols. 3 and 4) in the year before promotion. Couples where both partners have been promoted, as well as same sex couples, are excluded. Young children identifies couples with children under 11 years old. The difference variables are calculated as the promoted person minus the partner. Wife refers to females, irrespective of whether the couple is legally married. TPL variables (marked with a *) condition on having children below the age of 11. The difference in wages (annual earnings) are only calculated for those couples where wages (annual earnings) are non-missing for both spouses.

4.4.2 Raw within-couple differences by year

Before formally estimating how promotion affects household behavior, we plot the mean within-couple difference in wages and TPL days per year relative to promotion year in Figure 4.1. On average, the promoted person earns SEK 8,000 less than their partner prior to promotion, a difference that is substantially reduced upon promotion. Similarly, the promoted person tends to take more TPL than their partner prior to promotion. The figure suggests that the promotion eliminates or even reverses this gap.



Figure 4.1. Raw within-couple differences by year relative to year of promotion

Note: The figure plots the within-couple difference in wages (panel a) and normalized TPL days (panel b). The difference is defined as the promoted person minus the partner. Panel (b) is for the sub-sample of couples that have children below age 11.

4.4.3 Promotion impact on household relative wage

Table 4.2 shows the effect of promotion on household relative wages, i.e. the β -coefficients obtained when estimating equation 4.1. Overall, our results suggest that the reform had a sudden, large and persistent effect on the wage distribution within couples.

Column (1) confirms that the promotion translates into a wage difference between the promoted teacher and his/her non-promoted partner that is very close to the stipulated wage increase of 5,000 SEK. In columns (2) and (3) we consider impacts on the within-couple gender wage distribution. Promotions decrease the gender wage gap by 6.4 percentage points (32 percent).²⁹ Similarly, the fraction of female primary wage earners increases by 9.1 percentage points (30 percent).

²⁹We convert the point estimate to a percentage point change using the formula $100 \times (e^{\beta} - 1)$.

	(1)	(2)	(3)
Outcome:	Wage diff.	Couple gender	Woman earns
	(SEK)	wage gap	higher wages
Promoted	4,979***	-0.066***	0.091***
	(229)	(0.007)	(0.012)
R^2	0.693	0.367	0.076
Ν	52,414	52,414	52,414
Mean outcome $_{-1}$	-7,819	.18	.3
Year FE	Yes	Yes	Yes
Age dummies	Yes	Yes	Yes
Diff earnings ₋₄	Yes	Yes	Yes

 Table 4.2. Effect of promotion on household relative wage

Note: This includes results on the effect of promotion on the within-couple difference in wage, the couple gender wage gap (defined as $\ln(wage)_m - \ln(wage)_f$), and on the likelihood that the woman has a higher wage. It reports $\hat{\beta}$ from equation 4.1. Standard errors are clustered by municipality and included in parentheses. Mean outcome_1 shows the unconditional mean of the outcome in the year prior to promotion.

In Figure 4.2 we plot the event dummies from equation 4.2 to further investigate whether the household relative wage outcomes are parallel in promoted and not-yet-promoted households prior to the promotion event. Reassuringly, trends evolve very similar prior to promotion. In addition, the pattern suggests that the promotions generate a within-household wage differential that is persistent over time.³⁰

The reform impact on women's wages relative to their partners, documented in Table 4.2, is non-trivial and reflects that the reform had a disproportionate impact on female spouses. In Table 4.3 we consider the household responses separately depending on whether the male or the female spouse was promoted. In line with the intention of the reform, we find that there is a symmetric impact on the within-couple wage difference of around 5,000 SEK (col. (1) and (2)). Consequently, the couple gender wage gap decreases (by 14.4 percentage points) in "female promoted couples" and increases (by 16.3 percentage points) in "male promoted couples". When interpreting these estimates it is useful to note that the pre-promotion wage differential differs quite substantially depending on the gender of the promoted spouse. Promoted women earn substantially less (10,000 SEK/1,100 USD) than their male partner before promotion while wages are much more similar in couples with promoted males.

³⁰Appendix event figures A.2, A.3 and A.4 show the sensitivity of the results to different empirical models, for the full sample and separately by gender. Reassuringly, the wage effects are very robust to the choice of controls. Appendix table A.5 show that the results are similar if we instead define the sample of promoted teachers using a wage threshold of SEK 5,500.



(b) Couple gender wage gap

(c) Woman earns higher wages

Figure 4.2. Effect of promotion on within-couple difference in wage

Note: The figure plots γ_t with 95% confidence intervals from estimating equation 4.2 for all promoted couples. Calendar year fixed effects, promotion/partner-specific age dummies and the within-couple difference in annual earnings four years prior to promotion are included as controls.

	(1)	(2)	(3)	(4)
Outcome:	Wage diffe	erence (SEK)	Couple gen	der wage gap
Sample:	Women	Men	Women	Men
Promoted	5,006***	5,219***	-0.155***	0.151***
	(291)	(359)	(0.006)	(0.009)
R^2	0.699	0.493	0.610	0.491
Ν	38,420	13,994	38,420	13,994
Mean outcome $_{-1}$	-10,542	-534	.24	.01
Year FE	Yes	Yes	Yes	Yes
Age dummies	Yes	Yes	Yes	Yes
Diff earnings ₋₄	Yes	Yes	Yes	Yes

Table 4.3. Effect of promotion on household relative wage – by gender

Note: This includes results on the effect of promotion on the within-couple difference in wage and couple gender wage gap, separately for promoted women and promoted men. It reports $\hat{\beta}$ from equation 4.1. Standard errors are clustered by municipality and included in parentheses.

So far we have considered the within-couple difference in wages. In Panel A of Table 4.4, we show estimates from individual-level wage regressions. These confirm that the relative wage increase is driven by the wage effect of the promoted person, with no statistically significant effect on the spouse's wage. Another interesting dimension to consider is how hours worked is affected. While the promotion involves a substantial wage increase, the promotion also entails new tasks, even though the teacher continues mainly to teach. Moreover, the spouse may respond to the promotion by reducing the hours worked and shifting attention to the home, in light of their relatively worse bargaining position in the household, or by increasing their hours worked. In Panel B of Table 4.4, we show estimates from individual-level regressions in hours worked per month.³¹ The results suggest that promoted persons have increased their hours worked by 2.5 hours per month (around 1.5%) while the partners have not adjusted their hours worked. This also supports our interpretation that the career teacher promotion's main influence on the family was through the wage increase rather than an increase in hours worked.

	(1)	(2)	(3)	(4)	(5)	(6)
	All prop Promoted	motions 1 Partner	Female p Promoted	romotions 1 Partner	Male pro Promoted	motions 1 Partner
Panel A: Wages						
Promoted	5244.1**	**-83.2	5248.0**	*-82.0	5236.2**	*-32.5
	(49.1)	(112.9)	(45.3)	(153.4)	(123.8)	(136.2)
R^2	0.947	0.934	0.955	0.932	0.925	0.915
Ν	52,414	52,414	38,306	38,306	14,108	14,108
Mean outcome $_{-1}$	30,200	38,019	29,946	40,488	30,882	31,416
Panel B: Hours						
Promoted	2.5***	-0.4	2.3**	-0.3	3.1**	-0.8
	(0.9)	(0.7)	(0.9)	(0.9)	(1.4)	(1.4)
R^2	0.406	0.441	0.418	0.433	0.363	0.425
Ν	45,792	45,792	32,923	32,923	12,869	12,869
$Mean \ outcome_{-1}$	151.59	148.43	150.58	151.23	154.15	141.34
Year FE	Yes	Yes	Yes	Yes	Yes	Yes
Person FE	Yes	Yes	Yes	Yes	Yes	Yes

 Table 4.4. Effect of promotion on allocation of wages and hours within couple

Note: This includes results on the effect of promotion on the person's monthly wages (Panel A) and total hours worked per month in a month between September and November (Panel B). It is based on estimating the following model: $y_{ict} = \beta Promoted_{hct} + \lambda_i + \lambda_c + \varepsilon_{ict}$, where *i* refers to the individual. Estimations are done separately for promoted individuals (cols. (1), (3) and (5)), and their partners (cols. (2), (4) and (6)). Standard errors are clustered by municipality and included in parentheses.

³¹Hours are available for the same individuals for whom we observe wages, and are also measured between September and November. Note that the use of TPL tends to be the highest in February.

To sum up, we find that promotions had a sudden and clear impact on relative wages in the household. This response is driven by an increase in the wage for the promoted person, while the partner does not appear to adjust their labor supply. The promotions led to a reduction in couples' gender wage gap on average, reflecting the higher promotion rate of women. In the next section, we turn to our main results, namely the impact of promotions on TPL use. In this analysis, we focus on couples with children below age 11 only.³²

4.5 Promotion impact on the division of childcare

Table 4.5 shows the promotion effect on the household division of Temporary Parental Leave (TPL). In column (1), we show the impact of the promotion on the within-couple difference in TPL days per year among all promotions.³³ In line with the patterns in the raw data displayed in Figure 4.1, promoted persons decrease their use of TPL relative to their partners. The within-couple difference in TPL benefits as a share of monthly wages decline by three percentage points, which corresponds to approximately 0.6 days (s.e. 0.18). Column (2) shows the estimate separately for female promotions, indicating that the response of female promoted spouses is approximately as strong as the overall response. Since there are relatively few promoted men with young children, we lack statistical power for promoted men. We only present results separately for male promotions in the appendix (see Table A.6).

	(1)	(2)
	Total	Female
Promoted	-0.627***	-0.573***
	(0.180)	(0.201)
R^2	0.058	0.050
Ν	25,680	18,420
Mean outcome $_{-1}$.35	1.02
Child controls	Yes	Yes
Year FE	Yes	Yes
Age dummies	Yes	Yes
Diff earnings ₋₄	Yes	Yes

Table 4.5. Effect of promotion on within-couple difference in TPL days

Note: This reports $\hat{\beta}$ from equation 4.1 for the within-couple difference in normalized TPL days, defined as promoted person minus partner. Only couples with children under age 11 are included. Standard errors are clustered by municipality and included in parentheses. Mean outcome₋₁ shows the unconditional mean of the outcome in the year prior to promotion.

³²Similar wage results are found when we apply this sample restriction, i.e. in the TPL sample. ³³For details on how we compute this measure, see Section 4.3.1.

To put these estimates into context, promoted women generally take four days of TPL in the year prior to promotion while their partners take three days of TPL in the sample used for the estimation. Over the post-promotion period, the promotion therefore closes over half the gender gap in temporary parental leave for promoted women.

We have tested the robustness of these results in several ways. Appendix table A.6 shows the sensitivity of the results to different empirical models. The estimates are consistently negative and of similar magnitude across empirical specifications, which suggests that differences in observable couple characteristics have little impact on the estimates. We also assess the robustness of the estimates to including couple fixed effects instead of parametrically controlling for the within-couple difference in income four years prior to promotion, also finding negative effects. In our main analysis, we restrict the sample to couples that have children below age 11, but not all parents with young children take TPL. Panel A of appendix table A.7 shows that the point estimates are slightly larger if we further restrict the sample by removing couples where neither parent takes TPL. As above, the effects for promoted women indicate that over half of the pre-promotion gender difference in TPL is closed. We obtain similar estimates to our baseline results if we define the sample of promoted teachers using a wage threshold of SEK 5,500 instead of SEK 5,000, presented in Panel B of the same table. Finally, Panel C of the table includes results that show that the effect of promotion on the within-couple difference in TPL, measured in hundreds of crowns without normalizing, is also negative.

Figure 4.3 shows the corresponding event graphs from estimating equation 4.2. These confirm a clear negative response in total and women's use of TPL after promotion relative to their partners, which appears to grow slightly over time. Two years after the promotion, the promoted woman has reduced her use of normalized TPL just over 1 day compared to her partner. This corresponds to a full closing of the pre-promotion unconditional gap between the promoted woman and her partner. Appendix Figure A.5 shows that the estimated effect on TPL use is robust irrespective of the controls included in equation 4.2.



(a) All promoted couples

(b) Promoted women

Figure 4.3. Effect of promotion on within-couple difference in TPL days

Note: The figure plots γ_t with 95% confidence intervals from estimating equation 4.2. Only couples with children under age 11 are included. Calendar year fixed effects, promotion/partner-specific age dummies, the number of children aged 0–3, 4–6 and 7–10, and the within-couple difference in annual earnings four years prior to promotion are included as controls.

To better gauge whether the promoted person or the spouse adjusts their use of leave, we turn to individual-level regressions next. We separately estimate the use of TPL for the promoted person and their spouse. In Panel A of Table 4.6, we consider the extensive margin response, i.e. whether the promoted person or their spouse takes any temporary parental leave. The results suggest that the promoted person does not change their extensive margin response, while the promotion coincides with the partner increasing their likelihood of take-up by around 2.5 percentage points. In Panel B we consider normalized TPL days. These results suggest that that the promotion reflects a reallocation within the couple: the promoted person reduces the number of days taken, while the spouse increases the number of days taken. It should be noted that the results are imprecise, though the estimates typically reach statistical significance if we include fewer controls.

While our estimates capture the impact on temporary parental leave use, they can also be seen as a proxy for the allocation of household activities more generally. Using supplementary data, Ichino et al. (2019) show that one additional half-day of father's TPL is associated with a ten percentage point rise in the father's share of total home production. Based on their estimates, the female promotion response in the father's take-up of TPL by 0.3 days would translate into a 6 percentage point increase in the father's share of home production.

	(1)	(2)	(3)	(4)
	All pro	notions	Female p	omotions
	Promoted	Partner	Promoted	Partner
Panel A: Extensive margin				
Promoted	0.001	0.025**	0.001	0.024
	(0.011)	(0.013)	(0.013)	(0.015)
R^2	0.039	0.079	0.045	0.096
Ν	53,665	52,882	41,074	40,646
Mean outcome $_{-1}$.75	.58	.75	.53
Panel B: TPL days				
Promoted	-0.350**	0.245	-0.214	0.292*
	(0.154)	(0.202)	(0.190)	(0.174)
R^2	0.050	0.097	0.045	0.107
Ν	25,837	25,790	18,485	18,507
Mean outcome $_{-1}$	3.91	3.56	3.99	2.96
Child controls	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes
Person age dummies	Yes	Yes	Yes	Yes
Person earnings ₋₄	Yes	Yes	Yes	Yes

Table 4.6. Effect of promotion on allocation of TPL within couple

Note: This includes results on the effect of promotion on whether the person takes any leave (Panel A) and TPL days (Panel B). Estimations are done separately for promoted individuals (cols. (1) and (3)), and their partners (cols. (2) and (4)). Only couples with children under age 11 are included. Standard errors are clustered by municipality and included in parentheses.

4.5.1 Heterogeneity analysis

Next, we present results of a heterogeneity analysis. We modify our previous empirical model (4.1) by interacting the effect of promotion with Z as follows, where Z is the characteristic in question that we are interested in:

$$\tilde{y}_{hct} = \alpha + \beta Promoted_{hct} + \theta Promoted_{hct} \times Z_h + \psi Z_h$$

$$+ \lambda_c + \sum_a \lambda_{ac}^i + \sum_a \lambda_{ac}^j + \theta \tilde{I}_{h,-4} + \varepsilon_{hct}$$
(4.3)

As previously, \tilde{y}_{hct} is the within-couple difference in TPL days, defined as the promoted person minus the partner. As before we also include controls for calendar year fixed effects, the number of children, the promoted person *i* and the partner *j*'s age, and the within-couple difference in income four years prior to promotion.

First, we consider whether the promotion response differs by the age of the children. Parents tend to take out the most leave when children are young. For example, in the year prior to promotion the parents on average take a total of eleven days of TPL if they have a child that is between 0 and 3; nine days if the child is between 4 and 6; and a bit less than seven days if the child is between 7

and 10. At the same time, as the child ages the promoted person increasingly tends to take more leave than the partner. For example, for children below age 7 the promoted person tends to take 0.2 days more in the year prior to promotion, while for children aged 7 to 10 the promoted person tends to take 0.4 days more.

In columns (1) and (2) of Table 4.7 we present our coefficients of interest, $\hat{\beta}$ and $\hat{\theta}$. Here Z is an indicator variable equal to one if the couple has at least one child aged 7 to 10.³⁴ $\hat{\beta}$ shows the estimated effect of promotion in couples where the promoted person only has children aged up to 7, and $\hat{\beta} + \hat{\theta}$ shows the estimated effect in couples where they have at least one child aged 7 to 10. The results suggest that the relative reduction in TPL use is greater among couples with pre-school aged children, with an attenuated effect if the couple has at least one older child. Comparing the estimates to the unconditional means in the outcome in the year prior to promotion, promotions among teachers with children below age 7 reverses the gap in TPL such that the partner takes more leave than the promoted person following promotion, while promotions among couples with children aged 7 and above closes the gap in leave between the promoted person and their spouse. While the same general patterns hold for promoted women, there is no statistically significant differential effect for those with children aged 7 and above at the 10% level. Moreover, since women tend to take more leave than men, women with children below age 7 also take more leave than their spouse on average in the post-period, but the gap in leave is substantially reduced.

The results in Table 4.5 are consistent with traditional specialization models suggesting that when one spouse's contribution to family income increases then his or her time spent on childcare decreases. However, relative income shifts may affect the bargaining positions within couples. To further investigate the underlying mechanisms we consider if the response differs depending on whether the promoted person becomes the primary earner thanks to the promotion or not. In the estimations we compare primary earners and nonprimary earners.³⁵ A positive estimate of $\hat{\theta}$ would suggest that, in couples where the promoted person becomes the primary earner, relative TPL use is affected less than in non-primary earner couples, while a negative estimate of $\hat{\theta}$ would suggest that the same couples respond more strongly to the promotion.

Columns (3) and (4) of Table 4.7 present the results of this exercise. Column (3), which is for both female and male promotions, suggests that the

³⁴Because the indicator variable is highly correlated with the number of children in a particular age interval, we do not include child controls in this specification.

 $^{^{35}}$ We define primary earner couples as couples where the promoted person earned between SEK 0 and 5,000 less than their spouse in the year prior to promotion. We define non-primary earners as couples where the spouse was earning at least 5,000 more than the promoted person in the year prior to promotion. We do not include couples where the promoted person was already the primary earner prior to promotion in the analysis.

reduction in childcare time is considerably larger if the promoted teacher becomes the primary earner.³⁶ In column (4) we focus on promoted women only. We cannot reject that this primary earner effect is equal for women, although we suffer from lack of precision in the gender-specific regression. Even so, the results are consistent with the primary earner effect being particularly strong for promoted men. As a whole the results suggest that, in cases where the promotion leads the teacher to become the primary breadwinner, there is an even stronger effect on the division of household chores, in line with an increase in the bargaining position of the promoted party. We acknowledge that a household bargaining model gives no particular significance to the point where the primary earner shifts *per se*. Even so, the results are at odds with a normsbased interpretation where the woman tries to overcompensate by increasing time spent at home once she is the primary breadwinner (Bertrand et al. 2015).

Next, we consider two different spousal job characteristics. First, we can imagine that couples have different opportunities to reallocate childcare within the household. When the spouse has more flexibility to work from home, it may facilitate their ability to take leave to care for their children when they fall ill. To test for this, we match in information about the spouse's ability to work from home based on their 4-digit occupation. As our primary telework-measure, we use the measure derived in Hensvik et al. (2020), based on the American Time Use Survey. We define an indicator variable that is equal to one if the spouse has a more flexible occupation than the promoted teacher.³⁷ The results of this analysis are displayed in columns (5) and (6) of Table 4.7, where the variable *Spousal telework* indicates that the spouse's job is more flexible. The results support that there is no differential effect on the within-couple difference in TPL when the spouse is more able to work from home. Thus, the response does not appear to be driven by the types of households with high workplace flexibility.³⁸

For the second, and final, spousal characteristic, we consider whether the promoted teacher has a teacher spouse.³⁹ When the couple consists of two teachers, they will have the same workplace flexibility and their work will place similar demands on them, at least pre-promotion. When on the same labor market, the promotion may involve a larger shift in bargaining power

³⁶In our sample, the reform pushes the teacher across the primary earner threshold in around 20 percent of cases.

³⁷For data reasions, we can only include couples where we have information on the spouse's occupation (SSYK2012), included in the data from 2014. The telework measure is a constant within the couple, based on the first time we observe the SSYK2012 code for the spouse. The spouse has a more flexible occupation in 30% of cases.

³⁸To test the robustness of our results, we have complemented our teleworking-measure with the one derived in Dingle and Neiman (2020) and Mongey et al. (2020), instead based on O*NET. Using these measures, we again find no statistically significant differential effect at the 10% level if the spouse has a higher ability to work from home.

³⁹We define the teacher spouse in the year prior to promotion based on their occupation. In our sample, just under 20% of promoted teachers have a teacher spouse.

or comparative advantage when one person in the couple is promoted. At the same time, that the spouse does not get promoted could lead to jealousy and an unwillingness to share household chores that is strengthened by the previous similarity in their jobs. We present estimates of $\hat{\beta}$ and $\hat{\theta}$ in columns (7) and (8) of Table 4.7. The results indicate that promoted teachers who are not married to a teacher reduce their take-out of TPL by around 0.5 days relative to their partner following promotion. The relative reduction is over twice as large when the spouse is a teacher, at least for female promotions.⁴⁰ For women with teacher spouses, the woman on average takes 0.3 more days of leave than their spouse in the year prior to promotion, while women without a teacher spouse on average take 1.1 days more. Teacher couples therefore share TPL more evenly pre-promotion. The results support that the promotion reverses this relatively small gap such that the teacher spouse takes more leave than the promoted person post-promotion.

⁴⁰Part of this effect could be driven by the primary earner effect. Around one third of teacher spouses are in couples where the promoted teacher becomes the primary earner.

	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
	Total	Female	Total	Female	Total	Female	Total	Female
Promoted	-1.143***	-0.841^{***}	-0.698***	-0.555**	-0.563***	-0.523**	-0.509***	-0.441^{**}
	(0.298)	(0.317)	(0.227)	(0.268)	(0.198)	(0.219)	(0.192)	(0.214)
Promoted \times Child 7–10	0.702^{***}	0.388						
	(0.268)	(0.306)						
Promoted \times Primary earner			-0.380*	-0.272				
			(0.199)	(0.227)				
Promoted \times Spousal telework					0.048	0.130		
					(0.202)	(0.225)		
Promoted $ imes$ Teacher spouse							-0.304	-0.620**
							(0.190)	(0.251)
R^2	0.057	0.045	0.047	0.045	0.070	0.062	0.058	0.050
Z	25,680	18,420	14,065	11,562	23,342	16,484	21,630	15,464
Child controls			Yes	Yes	Yes	Yes	Yes	Yes
Year FE	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Age dummies	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Diff earnings t-4	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
<i>Note:</i> This includes results on how the $\hat{\beta}$ and $\hat{\theta}$ from eqn. 4.3. Columns (1)	he effect of pron and (2) conside	notion on the w r whether the co	ithin-couple dif ouple has a chil	fference in nor d aged 7–10; c	malized TPL day olumns (3) and (ys differs by di (4) show how r	fferent character esults differ for J	istics. It reports primary earners;
columns (5) and (6) consider wheth	er the spouse ha	an occupation	n with a higher	telework abilit	y than the prom	oted person; a	nd columns (7) a	und (8) consider

day	
TPL	
in	
le difference	
-coup	
within	
ио	
fpromotion	
(O)	
Effec	
Heterogeneity:	
F :	
4	
Table	

by municipality and included in parentheses.

whether the promoted person has a teacher spouse. Only couples with children under the age of 11 are included in the regressions. Standard errors are clustered

4.6 Conclusion

Despite women's progress in the labor market and the universal and genderneutral family policies implemented in the Nordic countries, women continue to have the primary responsibility for child rearing. In this paper, we study the household childcare time allocation response to a policy-generated shift in household relative income. To this end, we exploit matched employeremployee data and a promotion reform in the Swedish labor market for teachers. The reform targeted skilled teachers and its main component was a pay increase of 20 percent of pre-reform mean wages. A particularly interesting feature of the reform is that it primarily affected women and secondary earners of the household.

We focus on dual-earner couples where one of the spouses was promoted due to the reform. Hence, our empirical strategy relies on the assumption that the timing of promotion is exogenous. The assumption is supported by the institutional feature that the number of promotions were rolled-out over time, constraining school districts' ability to promote skilled teachers. We also show that couple outcomes evolve in parallel in promoted vs. not-yet promoted couples prior to promotion, supporting our empirical strategy.

We find that the promotions have a pronounced and persistent impact on the wage distribution within couples: promotions of female spouses reduce a pre-existing couple gender wage gap by over 50 percent while promotions of male spouses generate a wage gap that was non-existent prior to the reform. However, since more women were promoted, the reform give rise to a withinhousehold gender pay gap reduction by 6.4 percentage points (32 percent) and an increase in the fraction of women who are the primary earner by 9.1 percentage points (30 percent).

Our results suggest that couples respond to promotions by reallocating childcare time away from the promoted spouse relative to their partner. The conclusion also holds if we only consider couples where the female spouse is promoted, suggesting that female promotions lead to a more equal division of childcare responsibilities. The impact is stronger when couples have small children and hence when childcare demands are greater. In addition, there is evidence of a "primary earner effect", suggesting a stronger response when the promoted partner becomes the primary wage earner thanks to the reform. A possible explanation for this could be that partners' bargaining power jumps at the point when they out-earn their partner.

In light of a recent literature highlighting the role of norms in explaining the persistence of traditional gender roles in the family, our results suggest that more traditional economic explanations manifested in the labor market should not be overlooked. Labor market policies directly targeting gender pay inequalities can thus be a powerful tool in promoting a more equal division of child responsibilities between mothers and fathers.

References

- Akerlof, G. A. and R. E Kranton (2000). "Economics and Identity". *The Quarterly Journal of Economics* 115.3, pp. 715–753.
- Angelov, N., Johansson, and E. Lindahl (2016). "Parenthood and the Gender Gap in Pay". *Journal of Labor Economics* 34.3, pp. 545–579.
- Avdic, D. and A. Karimi (2018). "Modern Family? Paternity Leave and Marital Stability". *American Economic Journal: Applied Economics* 10.4, pp. 283– 307.
- Azmat, G., L. Hensvik, and O. Rosenqvist (2020). *Workplace Presenteeism, Job Substitutability and Gender Inequality*. Tech. rep. 2020:9. IFAU.
- Becker, G. S., E. M. Landes, and R. T. Michael (1977). "An Economic Analysis of Marital Instability". *Journal of Political Economy* 85.6, pp. 1141– 1187.
- Bertrand, M., E. Kamenica, and J. Pan (2015). "Gender Identity and Relative Income Within Households". *The Quarterly Journal of Economics* 130.2, pp. 571–614.
- Borusyak, K. and X. Jaravel (2018). "Revisting Event Study Designs". Unpublished manuscript.
- Cortés, P. and J. Pan (2019). "When Time Binds: Substitutes for Household Production, Returns to Working Long Hours, and the Skilled Gender Wage Gap". *Journal of Labor Economics* 37.2, pp. 351–398.
- (2020). *Children and the Remaining Gender Gaps in the Labor Market*. Tech. rep. 27980. National Bureau of Economic Research.
- Denning, J. T., B. Jacob, L. Lefgren, and C. vom Lehn (2019). The Return to Hours Worked Within and Across Occupations: Implications for the Gender Wage Gap. Tech. rep. 25739. National Bureau of Economic Research.
- Dingle, J.I. and B. Neiman (2020). *How Many Jobs Can Be Done at Home?* Tech. rep. 26948. National Bureau of Economic Research.
- Duchini, E. and C. Van Effenterre (2020). *School Schedule and the Gender Pay Gap.* Tech. rep. 13791. IZA.
- Ekberg, J., R. Eriksson, and G. Friebel (2013). "Parental leave A Policy Evaluation of the Swedish "Daddy-Month" Reform". *Journal of Public Economics* 97, pp. 131–143.
- Eriksson, R. and M. Nermo (2010). "Care for Sick Children as a Proxy for Gender Equality in the Family". *Social Indicators Research* 97.3, pp. 341–356.
- Farré, L. and L. González (2019). "Does Paternity Leave Reduce Fertility?" *Journal of Public Economics* 172, pp. 52–66.

- Folke, O. and J. Rickne (2020). "All the Single Ladies: Job Promotions and the Durability of Marriage". *American Economic Journal: Applied Economics* 12.1, pp. 260–87.
- Försäkringskassan (2020). Socialförsäkringen i siffror 2020 (Social Insurance in Numbers 2020).
- Gelber, A. M. and J. W. Mitchell (2012). "Taxes and Time Allocation: Evidence from Single Women and Men". *The Review of Economic Studies* 79.3, pp. 863–897.
- Goldin, C. (2014). "A Grand Gender Convergence: Its Last Chapter". American Economic Review 104.4, pp. 1091–1119.
- Grönqvist, E., L. Hensvik, and A. Thoresson (2020). *Teacher Career Opportunities and School Quality*. Tech. rep. 2020:2. IFAU.
- Hensvik, L., T. Le Barbanchon, and R. Rathelot (2020). *Which Jobs Are Done from Home? Evidence from the American Time Use Survey*. Tech. rep. 13138. IZA.
- Hotz, V. J., P. Johansson, and A. Karimi (2017). *Parenthood, Family Friendly Workplaces, and the Gender Gaps in Early Work Careers*. Tech. rep. 24173. National Bureau of Economic Research.
- Ichino, A., M. Olsson, B. Petrongolo, and P. Skogman Thoursie (2019). Economic Incentives, Home Production and Gender Identity Norms. Tech. rep. 1283. IFN.
- Kleven, H., C. Landais, J. Posch, A. Steinhauer, and J. Zweimuller (2020a). Do Family Policies Reduce Gender Inequality? Evidence from 60 Years of Policy Experimentation. Tech. rep. 28082. National Bureau of Economic Research.
- Kleven, H., C. Landais, and J. E. Søgaard (2020b). "Does Biology Drive Child Penalties? Evidence from Biological and Adoptive Families". *AER Insights* forthcoming.
- Lundberg, S. and R. A. Pollak (1993). "Separate Spheres Bargaining and the Marriage Market". *Journal of Political Economy* 101.6, pp. 988–1010.
- Mongey, S., L. Pilossoph, and A. Weinberg (2020). *Which Workers Bear the Burden of Social Distancing Policies?* Tech. rep. 27085. National Bureau of Economic Research.
- OECD (2016). Walking the Tightrope: Background Brief on Parents' Work-Life Balance Across the Stages of Childhood. URL: http://www.oecd. org/social/family/Background-brief-parents-work-lifebalance-stages-childhood.pdf (visited on 02/11/2020).
- Patnaik, A. (2019). "Reserving Time for Daddy: The Consequences of Fathers' Quotas". *Journal of Labor Economics* 37.4, pp. 1009–1059.
- Pertold-Gebicka, Barbara, Filip Pertold, and Nabanita Datta Gupta (2016). *Employment Adjustments Around Childbirth*. Tech. rep. 9685. IZA.
- Regeringen (2013a). Förordning (2013:70) om statsbidrag till skolhuvudmän som inrättar karriärsteg för lärare (Ordinance (2013: 70) on state grants for school districts who establish career steps for teachers).

- (2013b). Karriärvägar för lärare i skolväsendet m.m. (Career paths for teachers in the school system, etc.) Regeringens proposition 2012/13:136.
- Schober, P. S. (2014). "Parental Leave and Domestic Work of Mothers and Fathers: A Longitudinal Study of Two Reforms in West Germany". *Journal of Social Policy* 43.2, p. 351.
- Stanfors, M. (2018). "Vad gör folk hela dagarna? Tidsanvändning och jämställdhet bland svenska kvinnor och män. (What Do People Do All Day? Time Use and Equality Among Swedish Women and Men". *Ekonomisk Debatt* 46.4, pp. 6–18.
- Statskontoret (2015). Uppföljning av karriärstegsreformen. Delrapport 1 (Followup of the career step reform. Interim report 1).
- (2017). Uppföljning av karriärstegsreformen. Delrapport 3 (Follow-up of the career step reform. Interim report 3).

Appendix A: Additional tables and figures



Figure A.1. Robustness: Validating proxy

Note: Using data from the career teacher promotion registers in Grönqvist et al. (2020), we investigate the robustness of the method outlined in Section 4.3 to calculate the number of promoted teachers. The figure plots how many actual promotions that the promotion proxy captures (solid line), how many promotions the proxy captures in the correct year (dashed line), how many promotions the proxy captures that it should not capture (termed "excess share") (dotted line), and finally how many promotions in a particular year the proxy captures that it should not (dash-dot line). The denominator in all comparisons is the number of actual promotions based on the career teacher register between 2013 and 2015 for whom the promoted person has an occupational code as a teacher (i.e. the sum of all possible career teacher promotions that we can capture with our method).



(b) Promoted women

(c) Promoted men

Figure A.2. Effect of promotion on within-couple difference in wage – by gender

Note: The figure plots γ_t with 95% confidence intervals from estimating equation 4.2 for promoted couples, using various specifications.



(b) Promoted women

(c) Promoted men

Figure A.3. Effect of promotion on couple gender wage gap – by gender

Note: The figure plots γ_t with 95% confidence intervals from estimating equation 4.2 for promoted couples, using various specifications.



Figure A.4. Effect of promotion on likelihood that woman has higher wage – by gender

Note: The figure plots γ_t with 95% confidence intervals from estimating equation 4.2 for promoted couples, using various specifications.



(a) All promoted couples

(b) Promoted women

Figure A.5. Robustness: Effect of promotion on within-couple difference in TPL days

Note: The figure plots γ_t with 95% confidence intervals from estimating equation 4.2 on the within-couple difference in normalized TPL days. Only couples with children under the age of 11 are included. The number of children aged 0–3, 4–6 and 7–10 are included as controls in all specifications that have controls.
	2007	2008	2009	2010	2011	2012	2013	2014	2015	2016	2017
Ever promoted	13,368	13,391	13,423	13,444	13,466	13,475	13,488	13,494	13,489	13,473	13,455
Couples ever promoted	9,861	9,869	9,891	9,902	9,907	9,903	9,904	9,905	9,903	9,899	9,885
Singles ever promoted	3,148	3,163	3,173	3,183	3,200	3,213	3,225	3,230	3,227	3,215	3,211
Promoted in c	0	0	0	0	0	0	3,064	10,623	13,489	13,473	13,455
Couples promoted in c	0	0	0	0	0	0	2,201	7,887	9,903	9,899	9,885

\sim
~
3
~
3
~
<u> </u>
0
5
-
5
-
Ś
2
_
5
5
-
é
2
-
5
0
<u> </u>
~
-
3
~
-
9
\sim
~
Ø
3
~
5
20
acl
eacl
teacl
l teacl
d teacl
ed teach
ed teach
ted teach
oted teacl
noted teacl
noted teacl
omoted teach
omoted teach
romoted teach
^o romoted teach
Promoted teach
. Promoted teach
L. Promoted teach
1. <i>Promoted teacl</i>
.1. Promoted teach
A.1. Promoted teach
A.1. Promoted teach
e A.1. Promoted teach
le A.1. Promoted teach
ole A.1. Promoted teacl
ble A.1. Promoted teacl

	Woman w. partner		Man w	v. partner
	Mean	sd	Mean	sd
Individual characteristics:				
Female	1.00	(0.00)	0.00	(0.00)
Married	0.82	(0.38)	0.81	(0.39)
Age	44.78	(7.73)	44.77	(8.27)
Compulsory	0.00	(0.01)	0.00	(0.04)
Upper secondary	0.00	(0.05)	0.01	(0.09)
Post-secondary or higher	1.00	(0.05)	0.99	(0.10)
Education missing	0.00	(0.00)	0.00	(0.02)
Young children	0.49	(0.50)	0.59	(0.49)
Claim TPL*	0.75	(0.43)	0.74	(0.44)
TPL (100 SEK)*	39.16	(52.18)	41.36	(76.30)
TPL (normalized)*	0.19	(0.25)	0.20	(0.36)
TPL days*	3.90	(5.19)	4.11	(7.61)
Non-missing wage	1.00	(0.00)	1.00	(0.00)
Monthly wage (SEK)	29,937	(3,091)	30,660	(3,371)
Annual earnings (SEK)	340,638	(71,253)	3641,65	(70,637)
Couple characteristics:				
Difference wage	-10.542	(19.148)	-534	(8,766)
Difference annual earnings	-122.761	(295.755)	41.805	(158.127)
Difference TPL (100 SEK)*	10 51	(5549)	-5 45	(69 54)
Difference TPL (normal)*	0.05	(0.27)	-0.06	(0.27)
Difference TPL days*	1.05	(5.77)	-1.27	(5.57)
Share wage earned by wife	0.44	(0.08)	0.50	(0.06)
Wife has higher wage	0.26	(0.44)	0.39	(0.49)
Observations	7,843		2,065	

Table A.2. Summary statistics in year before promotion – promoted women and men

Note: The table shows summary statistics for promoted teachers with partners, separately by whether the promoted person is a woman or a man. Couples where both partners have been promoted, as well as same sex couples, are excluded. Young children identifies couples with children up to 10 years old. The difference variables are calculated as the promoted person minus the partner. Wife refers to females, irrespective of whether the couple is legally married. TPL variables (marked with a *) condition on having children below the age of 11.

	Promoted with partner		Pa	rtner
	Mean	sd	Mean	sd
Individual characteristics:				
Female	0.70	(0.46)	0.30	(0.46)
Married	0.81	(0.39)	0.81	(0.39)
Age	39.65	(4.28)	40.61	(4.90)
Compulsory	0.00	(0.00)	0.01	(0.11)
Upper secondary	0.00	(0.05)	0.20	(0.40)
Post-secondary or higher	1.00	(0.05)	0.78	(0.41)
Education missing	0.00	(0.02)	0.00	(0.04)
Young children	1.00	(0.00)	1.00	(0.00)
Claim TPL	0.76	(0.43)	0.62	(0.49)
TPL (100 SEK)	39.34	(52.08)	35.20	(53.51)
TPL (normalized)	0.19	(0.25)	0.17	(0.26)
TPL days	3.91	(5.16)	3.56	(5.43)
Non-missing wage	1.00	(0.00)	1.00	(0.00)
Monthly wage	29,242	(2,807)	36,725	(16,338)
Annual earnings (SEK)	326,611	(67,251)	422,778	(242,620)
Couple characteristics				
Difference wages	-7.483	(16.272)		
Difference annual earnings	-96.167	(254.968)		
Difference TPL (100 SEK)	4.13	(57.10)		
Difference TPL (normal.)	0.02	(0.28)		
Difference TPL days	0.35	(5.79)		
Share wage earned by wife	0.46	(0.08)		
Wife has higher wage	0.31	(0.46)		
Observations	2,730		2,730	

 Table A.3. Summary statistics in year before promotion – TPL sample

Note: The table shows summary statistics for couples that have children below age 11, for whom a normalized measure of temporary parental leave (TPL) can be computed. Couples where both partners have been promoted, as well as same sex couples, are excluded. The difference variables are calculated as the promoted person minus the partner. Wife refers to female, irrespective of whether the couple is legally married.

	(1)	(2)	(3)	(4)
Panel A: ln(wage)				
Promoted	0.211***	0.191***	0.184***	0.145***
	(0.002)	(0.002)	(0.002)	(0.001)
R^2	0.382	0.617	0.725	0.940
Ν	1,281,935	1,281,935	1,279,366	1,225,662
Panel B: Monthly wage (SEK)				
Promoted	7188.7***	6678.4***	6478.8***	5321.1***
	(93.2)	(89.0)	(58.8)	(62.9)
R^2	0.415	0.619	0.723	0.935
Ν	1,281,935	1,281,935	1,279,366	1,225,662
Year FE	Yes	Yes	Yes	Yes
Workplace FE			Yes	
Teacher FE				Yes
Controls		Yes	Yes	

Table A.4. Teacher wage effects of promotion

Note: This table provides the results of estimating $y_{ic} = \theta Promoted_{ic} + \beta X_{ic} + \lambda_c + \varepsilon_{ic}$ among all teachers *i* in calendar years $c = 2007, 2008, \dots 2017$. Controls are dummies for female, age (in five age bands) and level of education (in six categories). Standard errors are clustered by municipality and included in parentheses.

	(1)	(2)	(3)
Outcome:	Wage diff.	Couple gender	Woman earns
	(SEK)	wage gap	higher wages
Promoted	5,107***	-0.062***	0.086***
	(245)	(0.007)	(0.012)
R^2	0.698	0.365	0.075
Ν	49,951	49,951	49,951
Mean outcome $_{-1}$	-7,780	.17	.3
Year FE	Yes	Yes	Yes
Age dummies	Yes	Yes	Yes
Diff earnings ₋₄	Yes	Yes	Yes

Table A.5. Robustness: Effect of promotion on household relative wage

Note: This includes results on the effect of promotion on the within-couple difference in wage, the couple gender wage gap, and on the likelihood that the woman has a higher wage. It reports $\hat{\beta}$ from equation 4.1. Promotions are identified using the method outlined in Section 4.3.1, but a wage increase threshold of SEK 5,500 is used instead of 5,000. Standard errors are clustered by municipality and included in parentheses. Mean outcome_1 shows the unconditional mean of the outcome in the year prior to promotion.

	(1)	(2)	(3)	(4)	(5)
All couples					
Promoted	-0.535***	-0.720***	-0.741***	-0.627***	-0.405**
	(0.082)	(0.187)	(0.185)	(0.180)	(0.161)
R^2	0.002	0.006	0.013	0.058	0.468
Ν	25,958	25,958	25,958	25,680	25,392
Mean outc. $_1$.35				
Promoted					
women					
Promoted	-0.494***	-0.588***	-0.611***	-0.573***	-0.372*
	(0.090)	(0.200)	(0.200)	(0.201)	(0.190)
R^2	0.001	0.010	0.012	0.050	0.463
Ν	18,581	18,581	18,581	18,420	18,134
Mean outc. $_1$	1.02				
Promoted					
men					
Promoted	-0.394**	-0.708**	-0.724**	-0.593*	-0.344
	(0.176)	(0.350)	(0.349)	(0.353)	(0.335)
R^2	0.003	0.009	0.014	0.032	0.449
Ν	7,377	7,377	7,377	7,260	7,258
Mean outc. $_1$	-1.26				
Child controls		Yes	Yes	Yes	Yes
Year FE		Yes	Yes	Yes	Yes
Age dummies			Yes	Yes	
$\Delta \text{ earnings}_{-4}$				Yes	
Couple FE					Yes

Table A.6. Effect of promotion on within-couple difference in TPL days

Note: The table includes results on the effect of promotion on the within-couple difference in normalized TPL days (defined as promoted person minus partner). It reports $\hat{\beta}$ from equation 4.1. Only couples with children under the age of 11 are included in the regressions. Child controls are number of children aged 0–3, 4–6 and 7–10. Age dummies are promotion/partner-specific age dummies in five-year bins. The difference in earnings shows the within-couple difference in annual earnings four years prior to promotion. A dummy is included to capture data more than three years prior to promotion in all models. Standard errors are clustered by municipality and included in parentheses. Mean outcome_1 shows the unconditional mean of the outcome in the year prior to promotion.

	(1)	(2)
	Total	Female
Panel A: Exclude no TPL		
Promoted	-0.711***	-0.696***
	(0.207)	(0.239)
R^2	0.075	0.067
Ν	21,852	15,504
Mean outcome $_{-1}$.41	1.22
Panel B: SEK 5,500 wage threshold		
Promoted	-0.606***	-0.548***
	(0.186)	(0.210)
R^2	0.058	0.049
Ν	24,613	17,569
Mean outcome $_{-1}$.33	.99
Panel C: TPL (100 SEK)		
Promoted	-4.183***	-3.342**
	(1.390)	(1.396)
R^2	0.026	0.065
Ν	52,521	40,340
Mean outcome $_{-1}$	6.3	10.06
Child controls	Yes	Yes
Year FE	Yes	Yes
Age dummies	Yes	Yes
Diff earnings ₋₄	Yes	Yes

Table A.7. Robustness: Effect of promotion on within-couple difference in TPL

Note: This includes results on the effect of promotion on the within-couple difference in TPL, defined as promoted person minus partner. It reports $\hat{\beta}$ from equation 4.1. Only couples with children under age 11 are included. In Panel A and B, the outcome is the difference in normalized TPL days, but the sample differs from the main analysis. In Panel A, we further restrict the sample by excluding couples where neither parent takes TPL. In Panel B, promotions are identified using the method outlined in Section 4.3.1, but a wage increase threshold of SEK 5,500 is used instead of 5,000. In Panel C, couples with children under age 11 are included. The outcome is the difference in TPL, measured in hundreds of SEK. Standard errors are clustered by municipality and included in parentheses.

Economic Studies

- 1987:1 Haraldson, Marty. To Care and To Cure. A linear programming approach to national health planning in developing countries. 98 pp.
- 1989:1 Chryssanthou, Nikos. The Portfolio Demand for the ECU. A Transaction Cost Approach. 42 pp.
- 1989:2 Hansson, Bengt. Construction of Swedish Capital Stocks, 1963-87. An Application of the Hulten-Wykoff Studies. 37 pp.
- 1989:3 Choe, Byung-Tae. Some Notes on Utility Functions Demand and Aggregation. 39 pp.
- 1989:4 Skedinger, Per. Studies of Wage and Employment Determination in the Swedish Wood Industry. 89 pp.
- 1990:1 Gustafson, Claes-Håkan. Inventory Investment in Manufacturing Firms. Theory and Evidence. 98 pp.
- 1990:2 Bantekas, Apostolos. The Demand for Male and Female Workers in Swedish Manufacturing. 56 pp.
- 1991:1 Lundholm, Michael. Compulsory Social Insurance. A Critical Review. 109 pp.
- 1992:1 Sundberg, Gun. The Demand for Health and Medical Care in Sweden. 58 pp.
- 1992:2 Gustavsson, Thomas. No Arbitrage Pricing and the Term Structure of Interest Rates. 47 pp.
- 1992:3 Elvander, Nils. Labour Market Relations in Sweden and Great Britain. A Comparative Study of Local Wage Formation in the Private Sector during the 1980s. 43 pp.
- 12 Dillén, Mats. Studies in Optimal Taxation, Stabilization, and Imperfect Competition. 1993. 143 pp.
- 13 Banks, Ferdinand E., A Modern Introduction to International Money, Banking and Finance. 1993. 303 pp.
- 14 Mellander, Erik. Measuring Productivity and Inefficiency Without Quantitative Output Data. 1993. 140 pp.
- 15 Ackum Agell. Susanne. Essays on Work and Pay. 1993. 116 pp.
- 16 Eriksson, Claes. Essays on Growth and Distribution. 1994. 129 pp.
- 17 Banks, Ferdinand E., A Modern Introduction to International Money, Banking and Finance. 2nd version, 1994. 313 pp.

- 18 Apel, Mikael. Essays on Taxation and Economic Behavior. 1994. 144 pp.
- 19 Dillén, Hans. Asset Prices in Open Monetary Economies. A Contingent Claims Approach. 1994. 100 pp.
- 20 Jansson, Per. Essays on Empirical Macroeconomics. 1994. 146 pp.
- 21 Banks, Ferdinand E., A Modern Introduction to International Money, Banking, and Finance. 3rd version, 1995. 313 pp.
- 22 Dufwenberg, Martin. On Rationality and Belief Formation in Games. 1995. 93 pp.
- 23 Lindén, Johan. Job Search and Wage Bargaining. 1995. 127 pp.
- 24 Shahnazarian, Hovick. Three Essays on Corporate Taxation. 1996. 112 pp.
- Svensson, Roger. Foreign Activities of Swedish Multinational Corporations. 1996.
 166 pp.
- 26 Sundberg, Gun. Essays on Health Economics. 1996. 174 pp.
- 27 Sacklén, Hans. Essays on Empirical Models of Labor Supply. 1996. 168 pp.
- 28 Fredriksson, Peter. Education, Migration and Active Labor Market Policy. 1997. 106 pp.
- 29 Ekman, Erik. Household and Corporate Behaviour under Uncertainty. 1997. 160 pp.
- 30 Stoltz, Bo. Essays on Portfolio Behavior and Asset Pricing. 1997. 122 pp.
- 31 Dahlberg, Matz. Essays on Estimation Methods and Local Public Economics. 1997. 179 pp.
- 32 Kolm, Ann-Sofie. Taxation, Wage Formation, Unemployment and Welfare. 1997. 162 pp.
- Boije, Robert. Capitalisation, Efficiency and the Demand for Local Public Services. 1997.148 pp.
- 34 Hort, Katinka. On Price Formation and Quantity Adjustment in Swedish Housing Markets. 1997. 185 pp.
- 35 Lindström, Thomas. Studies in Empirical Macroeconomics. 1998. 113 pp.
- 36 Hemström, Maria. Salary Determination in Professional Labour Markets. 1998. 127 pp.
- 37 Forsling, Gunnar. Utilization of Tax Allowances and Corporate Borrowing. 1998. 96 pp.
- 38 Nydahl, Stefan. Essays on Stock Prices and Exchange Rates. 1998. 133 pp.
- 39 Bergström, Pål. Essays on Labour Economics and Econometrics. 1998. 163 pp.

- 40 Heiborn, Marie. Essays on Demographic Factors and Housing Markets. 1998. 138 pp.
- 41 Åsberg, Per. Four Essays in Housing Economics. 1998. 166 pp.
- 42 Hokkanen, Jyry. Interpreting Budget Deficits and Productivity Fluctuations. 1998. 146 pp.
- 43 Lunander, Anders. Bids and Values. 1999. 127 pp.
- 44 Eklöf, Matias. Studies in Empirical Microeconomics. 1999. 213 pp.
- Johansson, Eva. Essays on Local Public Finance and Intergovernmental Grants. 1999.
 156 pp.
- 46 Lundin, Douglas. Studies in Empirical Public Economics. 1999. 97 pp.
- 47 Hansen, Sten. Essays on Finance, Taxation and Corporate Investment. 1999. 140 pp.
- 48 Widmalm, Frida. Studies in Growth and Household Allocation. 2000. 100 pp.
- 49 Arslanogullari, Sebastian. Household Adjustment to Unemployment. 2000. 153 pp.
- 50 Lindberg, Sara. Studies in Credit Constraints and Economic Behavior. 2000. 135 pp.
- 51 Nordblom, Katarina. Essays on Fiscal Policy, Growth, and the Importance of Family Altruism. 2000. 105 pp.
- 52 Andersson, Björn. Growth, Saving, and Demography. 2000. 99 pp.
- 53 Åslund, Olof. Health, Immigration, and Settlement Policies. 2000. 224 pp.
- 54 Bali Swain, Ranjula. Demand, Segmentation and Rationing in the Rural Credit Markets of Puri. 2001. 160 pp.
- 55 Löfqvist, Richard. Tax Avoidance, Dividend Signaling and Shareholder Taxation in an Open Economy. 2001. 145 pp.
- 56 Vejsiu, Altin. Essays on Labor Market Dynamics. 2001. 209 pp.
- 57 Zetterström, Erik. Residential Mobility and Tenure Choice in the Swedish Housing Market. 2001. 125 pp.
- 58 Grahn, Sofia. Topics in Cooperative Game Theory. 2001. 106 pp.
- 59 Laséen, Stefan. Macroeconomic Fluctuations and Microeconomic Adjustments. Wages, Capital, and Labor Market Policy. 2001. 142 pp.
- 60 Arnek, Magnus. Empirical Essays on Procurement and Regulation. 2002. 155 pp.
- Jordahl, Henrik. Essays on Voting Behavior, Labor Market Policy, and Taxation. 2002.
 172 pp.

- 62 Lindhe, Tobias. Corporate Tax Integration and the Cost of Capital. 2002. 102 pp.
- 63 Hallberg, Daniel. Essays on Household Behavior and Time-Use. 2002. 170 pp.
- 64 Larsson, Laura. Evaluating Social Programs: Active Labor Market Policies and Social Insurance. 2002. 126 pp.
- 65 Bergvall, Anders. Essays on Exchange Rates and Macroeconomic Stability. 2002. 122 pp.
- 66 Nordström Skans, Oskar. Labour Market Effects of Working Time Reductions and Demographic Changes. 2002. 118 pp.
- 67 Jansson, Joakim. Empirical Studies in Corporate Finance, Taxation and Investment. 2002. 132 pp.
- 68 Carlsson, Mikael. Macroeconomic Fluctuations and Firm Dynamics: Technology, Production and Capital Formation. 2002. 149 pp.
- 69 Eriksson, Stefan. The Persistence of Unemployment: Does Competition between Employed and Unemployed Job Applicants Matter? 2002. 154 pp.
- 70 Huitfeldt, Henrik. Labour Market Behaviour in a Transition Economy: The Czech Experience. 2003. 110 pp.
- 71 Johnsson, Richard. Transport Tax Policy Simulations and Satellite Accounting within a CGE Framework. 2003. 84 pp.
- 72 Öberg, Ann. Essays on Capital Income Taxation in the Corporate and Housing Sectors. 2003. 183 pp.
- 73 Andersson, Fredrik. Causes and Labor Market Consequences of Producer Heterogeneity. 2003. 197 pp.
- 74 Engström, Per. Optimal Taxation in Search Equilibrium. 2003. 127 pp.
- 75 Lundin, Magnus. The Dynamic Behavior of Prices and Investment: Financial Constraints and Customer Markets. 2003. 125 pp.
- 76 Ekström, Erika. Essays on Inequality and Education. 2003. 166 pp.
- 77 Barot, Bharat. Empirical Studies in Consumption, House Prices and the Accuracy of European Growth and Inflation Forecasts. 2003. 137 pp.
- 78 Österholm, Pär. Time Series and Macroeconomics: Studies in Demography and Monetary Policy. 2004. 116 pp.
- 79 Bruér, Mattias. Empirical Studies in Demography and Macroeconomics. 2004. 113 pp.
- 80 Gustavsson, Magnus. Empirical Essays on Earnings Inequality. 2004. 154 pp.

- 81 Toll, Stefan. Studies in Mortgage Pricing and Finance Theory. 2004. 100 pp.
- 82 Hesselius, Patrik. Sickness Absence and Labour Market Outcomes. 2004. 109 pp.
- 83 Häkkinen, Iida. Essays on School Resources, Academic Achievement and Student Employment. 2004. 123 pp.
- 84 Armelius, Hanna. Distributional Side Effects of Tax Policies: An Analysis of Tax Avoidance and Congestion Tolls. 2004. 96 pp.
- 85 Ahlin, Åsa. Compulsory Schooling in a Decentralized Setting: Studies of the Swedish Case. 2004. 148 pp.
- Heldt, Tobias. Sustainable Nature Tourism and the Nature of Tourists' Cooperative Behavior: Recreation Conflicts, Conditional Cooperation and the Public Good Problem.
 2005. 148 pp.
- 87 Holmberg, Pär. Modelling Bidding Behaviour in Electricity Auctions: Supply Function Equilibria with Uncertain Demand and Capacity Constraints. 2005. 43 pp.
- 88 Welz, Peter. Quantitative new Keynesian macroeconomics and monetary policy 2005. 128 pp.
- 89 Ågren, Hanna. Essays on Political Representation, Electoral Accountability and Strategic Interactions. 2005. 147 pp.
- 90 Budh, Erika. Essays on environmental economics. 2005. 115 pp.
- 91 Chen, Jie. Empirical Essays on Housing Allowances, Housing Wealth and Aggregate Consumption. 2005. 192 pp.
- 92 Angelov, Nikolay. Essays on Unit-Root Testing and on Discrete-Response Modelling of Firm Mergers. 2006. 127 pp.
- 93 Savvidou, Eleni. Technology, Human Capital and Labor Demand. 2006. 151 pp.
- 94 Lindvall, Lars. Public Expenditures and Youth Crime. 2006. 112 pp.
- 95 Söderström, Martin. Evaluating Institutional Changes in Education and Wage Policy. 2006. 131 pp.
- 96 Lagerström, Jonas. Discrimination, Sickness Absence, and Labor Market Policy. 2006. 105 pp.
- 97 Johansson, Kerstin. Empirical essays on labor-force participation, matching, and trade. 2006. 168 pp.
- 98 Ågren, Martin. Essays on Prospect Theory and the Statistical Modeling of Financial Returns. 2006. 105 pp.

- 99 Nahum, Ruth-Aïda. Studies on the Determinants and Effects of Health, Inequality and Labour Supply: Micro and Macro Evidence. 2006. 153 pp.
- 100 Žamac, Jovan. Education, Pensions, and Demography. 2007. 105 pp.
- 101 Post, Erik. Macroeconomic Uncertainty and Exchange Rate Policy. 2007. 129 pp.
- 102 Nordberg, Mikael. Allies Yet Rivals: Input Joint Ventures and Their Competitive Effects. 2007. 122 pp.
- 103 Johansson, Fredrik. Essays on Measurement Error and Nonresponse. 2007. 130 pp.
- 104 Haraldsson, Mattias. Essays on Transport Economics. 2007. 104 pp.
- 105 Edmark, Karin. Strategic Interactions among Swedish Local Governments. 2007. 141 pp.
- 106 Oreland, Carl. Family Control in Swedish Public Companies. Implications for Firm Performance, Dividends and CEO Cash Compensation. 2007. 121 pp.
- 107 Andersson, Christian. Teachers and Student Outcomes: Evidence using Swedish Data. 2007. 154 pp.
- 108 Kjellberg, David. Expectations, Uncertainty, and Monetary Policy. 2007. 132 pp.
- 109 Nykvist, Jenny. Self-employment Entry and Survival Evidence from Sweden. 2008. 94 pp.
- 110 Selin, Håkan. Four Empirical Essays on Responses to Income Taxation. 2008. 133 pp.
- 111 Lindahl, Erica. Empirical studies of public policies within the primary school and the sickness insurance. 2008. 143 pp.
- 112 Liang, Che-Yuan. Essays in Political Economics and Public Finance. 2008. 125 pp.
- 113 Elinder, Mikael. Essays on Economic Voting, Cognitive Dissonance, and Trust. 2008. 120 pp.
- 114 Grönqvist, Hans. Essays in Labor and Demographic Economics. 2009. 120 pp.
- 115 Bengtsson, Niklas. Essays in Development and Labor Economics. 2009. 93 pp.
- 116 Vikström, Johan. Incentives and Norms in Social Insurance: Applications, Identification and Inference. 2009. 205 pp.
- 117 Liu, Qian. Essays on Labor Economics: Education, Employment, and Gender. 2009. 133 pp.
- 118 Glans, Erik. Pension reforms and retirement behaviour. 2009. 126 pp.
- 119 Douhan, Robin. Development, Education and Entrepreneurship. 2009.

- 120 Nilsson, Peter. Essays on Social Interactions and the Long-term Effects of Early-life Conditions. 2009. 180 pp.
- 121 Johansson, Elly-Ann. Essays on schooling, gender, and parental leave. 2010. 131 pp.
- 122 Hall, Caroline. Empirical Essays on Education and Social Insurance Policies. 2010. 147 pp.
- 123 Enström-Öst, Cecilia. Housing policy and family formation. 2010. 98 pp.
- 124 Winstrand, Jakob. Essays on Valuation of Environmental Attributes. 2010. 96 pp.
- 125 Söderberg, Johan. Price Setting, Inflation Dynamics, and Monetary Policy. 2010. 102 pp.
- 126 Rickne, Johanna. Essays in Development, Institutions and Gender. 2011. 138 pp.
- Hensvik, Lena. The effects of markets, managers and peers on worker outcomes. 2011.179 pp.
- 128 Lundqvist, Heléne. Empirical Essays in Political and Public. 2011. 157 pp.
- 129 Bastani, Spencer. Essays on the Economics of Income Taxation. 2012. 257 pp.
- 130 Corbo, Vesna. Monetary Policy, Trade Dynamics, and Labor Markets in Open Economies. 2012. 262 pp.
- 131 Nordin, Mattias. Information, Voting Behavior and Electoral Accountability. 2012. 187 pp.
- 132 Vikman, Ulrika. Benefits or Work? Social Programs and Labor Supply. 2013. 161 pp.
- 133 Ek, Susanne. Essays on unemployment insurance design. 2013. 136 pp.
- 134 Österholm, Göran. Essays on Managerial Compensation. 2013. 143 pp.
- 135 Adermon, Adrian. Essays on the transmission of human capital and the impact of technological change. 2013. 138 pp.
- 136 Kolsrud, Jonas. Insuring Against Unemployment 2013. 140 pp.
- 137 Hanspers, Kajsa. Essays on Welfare Dependency and the Privatization of Welfare Services. 2013. 208 pp.
- 138 Persson, Anna. Activation Programs, Benefit Take-Up, and Labor Market Attachment. 2013. 164 pp.
- 139 Engdahl, Mattias. International Mobility and the Labor Market. 2013. 216 pp.
- 140 Krzysztof Karbownik. Essays in education and family economics. 2013. 182 pp.

- 141 Oscar Erixson. Economic Decisions and Social Norms in Life and Death Situations. 2013.183 pp.
- 142 Pia Fromlet. Essays on Inflation Targeting and Export Price Dynamics. 2013. 145 pp.
- 143 Daniel Avdic. Microeconometric Analyses of Individual Behavior in Public Welfare Systems. Applications in Health and Education Economics. 2014. 176 pp.
- 144 Arizo Karimi. Impacts of Policies, Peers and Parenthood on Labor Market Outcomes. 2014. 221 pp.
- 145 Karolina Stadin. Employment Dynamics. 2014. 134 pp.
- 146 Haishan Yu. Essays on Environmental and Energy Economics. 132 pp.
- 147 Martin Nilsson. Essays on Health Shocks and Social Insurance. 139 pp.
- 148 Tove Eliasson. Empirical Essays on Wage Setting and Immigrant Labor Market Opportunities. 2014. 144 pp.
- 149 Erik Spector. Financial Frictions and Firm Dynamics. 2014. 129 pp.
- 150 Michihito Ando. Essays on the Evaluation of Public Policies. 2015. 193 pp.
- 151 Selva Bahar Baziki. Firms, International Competition, and the Labor Market. 2015. 183 pp.
- 152 Fredrik Sävje. What would have happened? Four essays investigating causality. 2015. 229 pp.
- 153 Ina Blind. Essays on Urban Economics. 2015. 197 pp.
- 154 Jonas Poulsen. Essays on Development and Politics in Sub-Saharan Africa. 2015. 240 pp.
- 155 Lovisa Persson. Essays on Politics, Fiscal Institutions, and Public Finance. 2015. 137 pp.
- 156 Gabriella Chirico Willstedt. Demand, Competition and Redistribution in Swedish Dental Care. 2015. 119 pp.
- 157 Yuwei Zhao de Gosson de Varennes. Benefit Design, Retirement Decisions and Welfare Within and Across Generations in Defined Contribution Pension Schemes. 2016. 148 pp.
- 158 Johannes Hagen. Essays on Pensions, Retirement and Tax Evasion. 2016. 195 pp.
- 159 Rachatar Nilavongse. Housing, Banking and the Macro Economy. 2016. 156 pp.
- 160 Linna Martén. Essays on Politics, Law, and Economics. 2016. 150 pp.
- 161 Olof Rosenqvist. Essays on Determinants of Individual Performance and Labor Market Outcomes. 2016. 151 pp.
- 162 Linuz Aggeborn. Essays on Politics and Health Economics. 2016. 203 pp.

- 163 Glenn Mickelsson. DSGE Model Estimation and Labor Market Dynamics. 2016. 166 pp.
- 164 Sebastian Axbard. Crime, Corruption and Development. 2016. 150 pp.
- 165 Mattias Öhman. Essays on Cognitive Development and Medical Care. 2016. 181 pp.
- 166 Jon Frank. Essays on Corporate Finance and Asset Pricing. 2017. 160 pp.
- 167 Ylva Moberg. Gender, Incentives, and the Division of Labor. 2017. 220 pp.
- 168 Sebastian Escobar. Essays on inheritance, small businesses and energy consumption. 2017. 194 pp.
- 169 Evelina Björkegren. Family, Neighborhoods, and Health. 2017. 226 pp.
- 170 Jenny Jans. Causes and Consequences of Early-life Conditions. Alcohol, Pollution and Parental Leave Policies. 2017. 209 pp.
- 171 Josefine Andersson. Insurances against job loss and disability. Private and public interventions and their effects on job search and labor supply. 2017. 175 pp.
- 172 Jacob Lundberg. Essays on Income Taxation and Wealth Inequality. 2017. 173 pp.
- 173 Anna Norén. Caring, Sharing, and Childbearing. Essays on Labor Supply, Infant Health, and Family Policies. 2017. 206 pp.
- 174 Irina Andone. Exchange Rates, Exports, Inflation, and International Monetary Cooperation. 2018. 174 pp.
- 175 Henrik Andersson. Immigration and the Neighborhood. Essays on the Causes and Consequences of International Migration. 2018. 181 pp.
- 176 Aino-Maija Aalto. Incentives and Inequalities in Family and Working Life. 2018. 131 pp.
- 177 Gunnar Brandén. Understanding Intergenerational Mobility. Inequality, Student Aid and Nature-Nurture Interactions. 2018. 125 pp.
- 178 Mohammad H. Sepahvand. Essays on Risk Attitudes in Sub-Saharan Africa. 2019. 215 pp.
- 179 Mathias von Buxhoeveden. Partial and General Equilibrium Effects of Unemployment Insurance. Identification, Estimation and Inference. 2019. 89 pp.
- 180 Stefano Lombardi. Essays on Event History Analysis and the Effects of Social Programs on Individuals and Firms. 2019. 150 pp.
- 181 Arnaldur Stefansson. Essays in Public Finance and Behavioral Economics. 2019. 191 pp.
- 182 Cristina Bratu. Immigration: Policies, Mobility and Integration. 2019. 173 pp.
- 183 Tamás Vasi. Banks, Shocks and Monetary Policy. 2020. 148 pp.

- 184 Jonas Cederlöf. Job Loss: Consequences and Labor Market Policy. 2020. 213 pp.
- 185 Dmytro Stoyko. Expectations, Financial Markets and Monetary Policy. 2020. 153 pp.
- 186 Paula Roth. Essays on Inequality, Insolvency and Innovation. 2020. 191 pp.
- 187 Fredrik Hansson. Consequences of Poor Housing, Essays on Urban and Health Economics. 2020. 143 pp.
- 188 Maria Olsson. Essays on Macroeconomics: Wage Rigidity and Aggregate Fluctuations. 2020. 130 pp.
- 189 Dagmar Müller. Social Networks and the School-to-Work Transition. 2020. 146 pp.
- 190 Maria Sandström. Essays on Savings and Intangible Capital. 2020. 129 pp.
- Anna Thoresson. Wages and Their Impact on Individuals, Households and Firms. 2020.
 220 pp.